Essays in Behavioral and Experimental Economics

Inauguraldissertation zur Erlangung des akademischen Grades eines Doktors der Wirtschaftswissenschaften der Universität Mannheim

vorgelegt von

Raphael Epperson

im Frühjahrs-/Sommersemester 2021

Abteilungssprecher:	Prof. Volker Nocke, Ph.D.
Referent:	Prof. Ulrich J. Wagner, Ph.D.
Korreferent:	Prof. Dr. Henrik Orzen
Tag der mündlichen Prüfung:	18. Juni 2021

Acknowledgements

First and foremost, I would like to thank my advisors Ulrich Wagner and Henrik Orzen for their continuous support and guidance throughout my Ph.D. studies. I am very grateful for their thorough comments on my work, their valuable advice, and the fact that they have always been available to discuss any research-related issue.

Throughout my Ph.D. studies, I have also benefited from extensive discussions with numerous fellow researchers at the University of Mannheim and other institutions. Their comments and suggestions helped me to become a better researcher and greatly improved the chapters in this dissertation (see also the detailed acknowledgments at the beginning of each chapter). I would like to especially thank my co-authors Johannes Diederich, Andreas Gerster, and Timo Goeschl for the fruitful collaborations and their support aside from our joint research projects. Special thanks also go to Matthias Stelter for the stimulating conversations at the time when we were sharing an office or, later on, at our lunch meetings.

For Chapter 1 and Chapter 4, I gratefully acknowledge funding by the German Research Foundation (DFG) through CRC TR 224 (Project B07) and SFB 884 (Project B10), respectively. More generally, I thank the Graduate School of Economic and Social Sciences at the University of Mannheim for providing financial support and a productive environment to develop my research skills.

Lastly, and most importantly, I would like to express my deepest gratitude to my family and closest friends. They have always been an incredible support and a source of inspiration. The path that lies behind me would have been impossible without them.

Contents

\mathbf{Li}	st of	Figure	es	vii
\mathbf{Li}	st of	Tables	3	ix
Pr	reface	e		1
1	Info	rmatic	on Avoidance and Moral Behavior	5
	1.1	Introd	uction	5
	1.2	Experi	ment	11
		1.2.1	Information Signal About Intensive Farming	11
		1.2.2	Design \ldots	13
		1.2.3	Procedure	16
		1.2.4	Data	19
	1.3	Empiri	ical Strategy	23
	1.4	Result	s	26
		1.4.1	Demand for Information	26
		1.4.2	Information Effect in the Laboratory	27
		1.4.3	Information Effect in the Field	27
		1.4.4	Heterogeneity by Information Demand	31
		1.4.5	Exploratory Heterogeneity Analyses	32
	1.5	Discus	sion \ldots	34
	1.6	Conclu	usion	37
2	Hov	v to De	esign the Ask?	41
	2.1	Introd	uction	41
	2.2	Experi	mental Design	44
		2.2.1	Donation Appeal	44
		2.2.2	Experimental Protocol	45

	2.3	Results	7
		2.3.1 Donations $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 44$	9
		2.3.2 Propensity to Give	1
	2.4	Discussion and Conclusion	3
3	Sub	osidizing Unit Donations 55	5
	3.1	Introduction	5
	3.2	Background	3
		3.2.1 Unit vs. Money Donations	3
		3.2.2 Subsidizing Unit Donations	9
		3.2.3 Related Literature	1
	3.3	Experimental Design	3
		3.3.1 Donation Appeal	3
		3.3.2 Experimental Protocol	4
	3.4	Results	7
		3.4.1 Rebates versus Matches)
		3.4.2 Discount Subsidies	3
	3.5	Discussion	7
	3.6	Conclusion	9
	Ð		
4	Do	Beliefs About Lobbying Affect Pro-environmental Behavior? 81	L
	4.1	Introduction	L
	4.2	Experiment	1
		4.2.1 The German Internet Panel	1
		4.2.2 Experimental Design	5
		4.2.3 Follow-Up	7
		$4.2.4 \text{Sample} \dots \dots$	3
	4.3	Results	1
		$4.3.1 \text{Beliefs} \dots \dots \dots \dots \dots \dots \dots \dots \dots $	1
		4.3.2 Pro-environmental Behavior	3
	4.4	Concluding Remarks	7
\mathbf{A}	ppen	dices 99)
	А	Appendix to Chapter 1	9
		A.1 Mapping of Paper and Pre-analysis Plan 99	9
		A.2 Theoretical Predictions about Heterogeneity 102	1

iv

	A.3	Estimation Details	. 106
	A.4	Beliefs	. 108
	A.5	Robustness Checks	. 111
	A.6	Instructions for WTP Elicitation	. 119
	A.7	Video Content	. 124
	A.8	Additional Tables	. 126
	A.9	Additional Figures	. 139
В	Appen	dix to Chapter 2 \ldots \ldots \ldots \ldots \ldots \ldots	. 145
	B.1	Wording of Donation Appeal	. 145
	B.2	Additional Tables	. 148
\mathbf{C}	Appen	dix to Chapter $3 \ldots \ldots$. 149
	C.1	Ordered Probit Model	. 149
	C.2	Additional Figures and Tables	. 153
	C.3	Instructions	. 157
	C.4	Within-Subjects Design	. 158
D	Appen	dix to Chapter 4	. 163
	D.1	Additional Tables	. 163
	D.2	Additional Figures	. 170
Bibliog	graphy		173
Eidesst	attlich	ne Erklärung	189
Curric	ulum V	Vitae	191

v

List of Figures

1.1	Illustration of Laboratory Procedure
1.2	Demand for Information
1.3	Effect of Information over Time
1.4	Effect of Information Provision on Meat Consumption in the Lab-
	oratory
2.1	Cumulative Distribution of Donations in the Pure Scheme Treat-
	ments
3.1	Charity Receipts (Divided into Net Donations and Subsidy Pay-
	ments) by Treatment
4.1	Treatment Assignment in Wave 44 (November 2019)
4.2	Distribution of Beliefs in 2019
4.3	Beliefs over Time
4.4	Contributions to CO_2 Reduction via atmosfair
4.5	Standardized Treatment Effects
A.1	Responses Related to Intensive Farming
A.2	Responses Related to Meat Consumption
A.3	Demand for Information by Location
A.4	Screenshot Multiple Price List
A.5	Distribution of Prices
A.6	Treatment Status by WTP for Information
A.7	Observations in Moving Time Window
A.8	Effect of Information over Time
A.9	Moving Average of Meat Consumption
A.10	Likelihood to Purchase Food over Time

A.11	Information Effects by Selection in the Laboratory 142
A.12	Effect of Information Provision on Meat Consumption in the Field 143
A.13	Information Effects by Selection in the Field
C.1	Example Donation Appeal, 50% Discount Treatment. The Final
	Paragraph Differed between Treatments
C.2	Donation Appeal in the WS Design
D.1	Distribution of Beliefs in 2019
D.2	Distribution of Beliefs in 2020
D.3	Beliefs over Time by Baseline Belief

List of Tables

1.1	Descriptive Statistics	21
1.2	Information Effect in the Laboratory	28
1.3	Information Effect in the Field	29
1.4	Information Effect in the Laboratory – Avoider and Seeker	32
1.5	Information Effect by Meat-Eating Habit	33
2.1	Treatments	46
2.2	Summary Statistics	47
2.3	Donations	48
2.4	Effect on Donations (OLS Regression)	50
2.5	Effect on the Propensity to Give (Linear Probability Model)	52
3.1	Treatment Conditions	64
3.2	Final Part of Donation Appeal Wording by Treatment \ldots .	65
3.3	Descriptive Results	69
3.4	Estimation Results	73
4.1	Summary Statistics	90
4.2	Estimation Results – Pro-environmental Behavior	95
A.1	Mapping of Paper and Pre-analysis Plan	100
A.2	Information Effect in the Laboratory Based on True Propensity	
	Scores	112
A.3	Information Effect in the Field Based on True Propensity Scores .	112
A.4	Information Effect in the Laboratory Based on True Propensity	
	Scores – Avoider and Seeker	113
A.5	Information Effect in the Field Based on True Propensity Scores	
	– Avoider and Seeker	114

A.6	Information Effect Based on OLS
A.7	Information Effect Based on OLS – Avoider and Seeker 115
A.8	Information Effect in the Laboratory by Location \hdots
A.9	Information Effect in the Laboratory by Location – Avoider and
	Seeker
A.10	Information Effect in the Field by Location $\ . \ . \ . \ . \ . \ . \ . \ . \ . \ $
A.11	Information Effect in the Field by Location – Avoider and Seeker $$ 118 $$
A.12	Video Content
A.13	Legal Situation $\ldots \ldots 125$
A.14	Comparison by Whether Subjects Purchased Meal after ${\rm Experiment}126$
A.15	Comparison of Treatment Groups
A.16	Correlation of WTP and Observable Characteristics
A.17	Likelihood to Purchase Meal in the Field
A.18	Information Effect in the Field Controlling for Meat Eating Habit 129
A.19	Information Effect in the Laboratory with Restricted Sample 130 $$
A.20	Information Effect in the Field – Avoider and Seeker $\ . \ . \ . \ . \ . \ . \ . \ . \ . \ $
A.21	Information Effect in the Laboratory with Restricted Sample –
	Avoider and Seeker
A.22	Information Effect in the Field – Meat from Terrestrial and Aquatic
	Animals
A.23	Information Effect in the Field – Pork and Meat from Other Ter-
	restrial Animals
A.24	Information Effect in the Field over Time
A.25	Likelihood to Purchase Food over Time
A.26	Information Effect in the Field over Time for Constant Sample $\ . \ 136$
A.27	Information Effect in the Laboratory – Low and High WTP $$ 136
A.28	Information Effect in the Field – Low and High WTP 137
A.29	Information Effect in the Field for Meat from Terrestrial Animals
	– Avoider and Seeker
B.1	Summary Statistics by Treatment
C.1	Latent Variable and Individual Choice
C.2	Summary Statistics
C.3	Choice Set by Treatment
C.4	Robustness Check for Charity Receipts Censored at 4 Packages . 156

C.5	Summary Statistics for BS Sample, WS Sample, and Combined			
	Sample			
C.6	Descriptive Results, Within-Subjects Design			
D.1	Statements by Treatment Group (Translated)			
D.2	Contribution Decision (Translated)			
D.3	Variables $\ldots \ldots 165$			
D.4	Estimation Results – Beliefs			
D.5	Summary Statistics – Outcome Variables			

Preface

This dissertation consists of four self-contained chapters in Behavioral and Experimental Economics. The chapters address distinct research questions about individual behavior in various domains, with a particular focus on prosocial behavior and the role of information in economic decision-making. Chapter 1 studies the demand for, and avoidance of, moral information and its impact on behavior in the context of meat consumption. Chapters 2 and 3 focus on charitable giving. In particular, Chapter 2 explores how asking individuals to fund charitable goods (instead of simply providing money) influences donations, while Chapter 3 complements this line of research by analyzing the effectiveness of subsidies in such a decision environment. Chapter 4 investigates whether beliefs about lobbying in the context of climate protection affect individuals' pro-environmental behavior.

Besides the common focus on individual behavior, a unifying element is that all chapters rely on experimental methods. Over the last decades, experiments have emerged as a valuable tool in economic research, offering major advantages such as clean and simple inference on causal effects. Chapter 1 employs a laboratory experiment (and complementary field data), which allows us to control information acquisition and its costs. Chapters 2 and 3 are based on online experiments where only specific characteristics of the decision environment are changed in order to identify their relevance. Chapter 4 makes use of an experiment in a survey panel to exogenously shift beliefs and study the consequences for individual behavior. In all of these cases, the experimental approach is crucial in establishing causality and addressing the considered research questions. The studied populations are diverse, including students (Chapter 1), crowd workers (chapters 2 and 3), and a large heterogeneous sample from the German population (Chapter 4). In the following, I provide a brief summary of each chapter.

Chapter 1 is joint work with Andreas Gerster. In this chapter, we investigate the presence and consequences of information avoidance in a moral context, i.e., a situation where an individual's action might negatively affect others. Information about the consequences of consumption behavior can help consumers to reduce actions that are morally questionable or cause social costs. However, consumers might also avoid costless information in order to circumvent moral concerns from behaving selfishly. We explore this issue in the context of food choices, where the consumption of meat from intensive farming negatively affects animal welfare. We conduct a laboratory experiment and use complementary field data from university canteens to address three main research objectives. First, we elicit individuals' willingness-to-pay for information on the living conditions of animals in intensive farming and test whether information avoidance is present in this setting. Second, we estimate the effect of receiving such information on individuals' meat consumption in the laboratory and in the field. Third, we analyze how the effect of information differs between information seekers and avoiders, i.e., individuals with a non-negative and negative willingness-to-pay for information, respectively.

We find that about 30 percent of subjects avoid information on animals' living conditions in intensive farming. When receiving information, subjects significantly reduce their propensity to consume meat on average by about 12 percentage points in the laboratory and 6 to 9 percentage points in university canteens. The effect in the field is mainly driven by the first days after the experiment and is statistically indistinguishable from zero about one week after the experiment. In the laboratory, we also find suggestive evidence that individuals who select out of information are particularly responsive to it. This selection pattern impedes the effectiveness of information provision, even when information is provided for free.

Chapter 2 focuses on charitable giving and is joint work with Johannes Diederich and Timo Goeschl. Charities frequently deviate from the standard donation scheme in which potential donors are asked how much money they are willing to give. Instead, they ask donors to choose how many units of a charitable good (e.g., meals, bed nets, or trees) to fund at a given unit price. In an online donation experiment, we compare the performance of such a "unit donation" scheme with that of the standard "money donation" and investigate the factors that could explain differences. We find that despite the additional demands that it imposes on the charity, the unit donation does not outperform the money donation scheme in terms of overall donations. It significantly differs, however, with respect to the propensity to give. The sign of the difference depends on the granularity of the scheme. When one unit of the charitable good is cheap, unit donation schemes increase the propensity to give and hence can serve as an effective tool for recruiting donors.

Chapter 3 is joint work with Johannes Diederich, Catherine C. Eckel, Timo Goeschl, and Philip J. Grossman. In this chapter, we complement the research on unit donation schemes presented in Chapter 2 by investigating the effectiveness of subsidies in such a decision context. An influential result in the literature on charitable giving is that matching subsidies – where a third party matches an individual's donation at a given rate – dominate rebate subsidies – where a third party refunds a fraction of the donation back to the donor – in raising funds. We investigate whether this result extends to unit donation schemes. In particular, we compare matches and rebates as well as simple discounts on the unit price in an online experiment in which donors' choices are over the number of units of a charitable good to fund (rather than the amount of money to give). In contrast to the standard result in the money donation literature, we find no evidence of dominance: The three subsidy types are equally effective overall. At a more disaggregated level, rebates lead to a higher likelihood of giving, while matching and discount subsidies lead to larger donations by donors. This suggests that charities using a unit donation scheme enjoy additional degrees of freedom in choosing a subsidy type. Rebates merit additional consideration if the primary goal is to attract donors.

Chapter 4 presents results from a survey experiment designed to identify the causal impact of beliefs about lobbying on individuals' pro-environmental behavior. In the context of environmental policy, lobbying is often expected to lower or prevent stricter regulations. A crucial question is whether such beliefs affect individuals' willingness to engage in pro-environmental behavior. I use a survey experiment with a heterogeneous sample from Germany that induces random variation in individuals' beliefs about the impact of lobbying. In the experiment, participants read three statements which either all provide reasons that lobbying increases or decreases climate protection. Afterward, participants have the opportunity to contribute to the reduction of carbon emissions by donating to a climate protection organization. The donation decision serves as a measure of pro-environmental behavior. A follow-up survey provides information on stated pro-environmental behaviors and the persistence of beliefs. While the exogenous

belief manipulation is successful, I find mixed evidence on whether expecting a more negative impact of lobbying on climate protection affects pro-environmental behavior. Being confronted with positive instead of negative statements significantly decreases stated pro-environmental behavior. Although the estimated treatment effect on the observed contribution to the reduction of carbon emissions points in the same direction, it is not statistically different from zero at any conventional level.

Chapter 1

Information Avoidance and Moral Behavior: Experimental Evidence from Food Choices^{*}

with Andreas Gerster

1.1 Introduction

Information technologies, such as the Internet, give individuals access to comprehensive information about the consequences of their actions. Such information may help consumers to reduce behaviors that are morally questionable or cause social costs. Whether the availability of information induces behavioral change depends on consumers' willingness to acquire it, however. Evidence from laboratory experiments shows that individuals willfully ignore information about the

^{*}We are grateful to Ulrich Wagner, Henrik Orzen, Timo Goeschl, Efi Adamopoulou, Marius Alt, Antoine Camous, Carlo Gallier, Wolfgang Habla, Franziska Heinicke, Lorenz Goette, Wladislaw Mill, Davide Pace, Yoshiyasu Rai, Christoph Rothe, Matthias Stelter, Nora Szech, Dimitri Szerman, Joël van der Weele, Florian Zimmermann, and audiences at the 5th Workshop on Experimental Economics for the Environment in Heidelberg, the ESA 2020 Global Online Around-the-Clock Conference, ESA's Job-Market Candidates Seminar Series, the Innsbruck Winter Summit 2021, Heidelberg University, the University of Bonn, the University of Hamburg, the University of Mannheim, and ZEW – Leibniz Centre for European Economic Research for helpful comments and suggestions. We thank Max Dreher and Marina Hoch for excellent research assistance. Funding by the German Research Foundation (DFG) through CRC TR 224 (Project B07) is gratefully acknowledged. This study was pre-registered on the AEA RCT Registry (AEARCTR-0005015). IRB approval was obtained from the Ethics Committee at the University of Mannheim (IRB approval number: EK Mannheim 30/2019).

consequences of their actions for others in order to behave more selfishly (e.g., Dana et al., 2007; Larson and Capra, 2009; Feiler, 2014). A crucial, yet little explored, question is whether individuals also avoid moral information that is relevant outside of a laboratory game. If they do, better access to moral information may only have a negligible effect on everyday behaviors.

In this paper, we investigate the presence and consequences of information avoidance in the context of food consumption. Expenditures for food account for a substantial share of GDP,¹ and farming practices have far reaching implications for land use, greenhouse gas emissions, and animal welfare. We focus on consumers' decision to eat meat from intensive farming. The production of such meat typically harms animals, for example, by substantially restricting their living space. Since many individuals consider inflicting harm on animals as morally questionable behavior (see Lin-Schilstra and Fischer, 2020, for a review), consuming meat from intensive farming involves a tradeoff between enjoying cheap meat and not harming or killing animals.² As a result, individuals might avoid information about animals' living conditions to circumvent moral concerns from meat consumption (Hestermann et al., 2020). Information avoidance could thereby contribute to elevated meat consumption levels that compromise animal welfare and exacerbate societal problems such as climate change (Gerber et al., 2013; Girod et al., 2014; Poore and Nemecek, 2018), fresh water scarcity (Mekonnen and Hoekstra, 2012; Jalava et al., 2014), and excessive land use (Machovina et al., 2015; Poore and Nemecek, 2018).

Our paper has three main objectives. First, we elicit individuals' willingnessto-pay for information on the living conditions of animals in intensive farming and test whether information avoidance is present in this setting. Second, we estimate the effect of receiving such information on individuals' meat consumption. Third, we analyze how the effect of information differs between information seekers and avoiders, i.e., individuals with a non-negative and negative willingness-to-pay for information, respectively. Based on these heterogeneous treatment effect estimates, we discuss how the effect of providing moral information depends on

¹For example, food expenditures in the United States amounted to 1.77 trillion dollars in 2019 (https://www.ers.usda.gov/data-products/chart-gallery/gallery/chart-detail/?chartId =58364, accessed August 21, 2020), which corresponds to about 8 percent of U.S. GDP.

²It has also been documented that individuals are willing to make sacrifices when facing a tradeoff between their own payoff and animals' well-being (Falk and Szech, 2013; Albrecht et al., 2017; Falk et al., 2020).

1.1 Introduction

the implied costs of information acquisition.

Our analysis builds on a laboratory experiment and complementary data on food consumption from university canteens. In the laboratory experiment, we give subjects the option to watch a 360° video about the living conditions of pigs in intensive farming via a virtual reality (VR) headset. The outside option is to watch a 360° video of the same duration on the German central bank, which all subjects have already watched at an earlier stage of the experiment. We elicit the willingness-to-pay (WTP) for information about intensive farming through an incentivized multiple price list approach and randomly implement one choice from that list. To measure meat consumption in the laboratory, we let subjects choose between a voucher for a meal with or without meat at the university canteen. Based on consumption data from the university canteens before and after the experiment, we additionally measure subjects' meat consumption in the field.³ In our main analysis, we focus on the meat of any animal, including fish. To identify the effect of information on meat consumption, we exploit the fact that the likelihood to receive information depends merely on participants' WTP, which we elicit. Hence, we can use standard program evaluation methods, such as inverse probability weighting, to estimate average treatment effects (Imbens and Wooldridge, 2009).

We find that about 30 percent of subjects avoid information on intensive farming when it is costless. Furthermore, our results show that receiving such information decreases the likelihood to eat meat. In the laboratory, information reduces the likelihood to choose a voucher for a meal with meat by about 12 percentage points. In the field, the reduction of meat consumption amounts to 6 to 9 percentage points. The information effect is particularly pronounced immediately after the experiment and becomes indistinguishable from zero after about one week.

Furthermore, the laboratory data provides suggestive evidence that information avoiders, i.e., subjects with a negative WTP for information about intensive farming, are more responsive to information than subjects with a WTP of at least zero. In other words, particularly responsive individuals select out of information. By contrast, our field data does not show a clear pattern of selection based on treatment effects.⁴ We also explore how the costs of acquiring moral

 $^{^{3}}$ The meat products offered at the university canteens are generally not declared to come from any special husbandry conditions that improve animal welfare beyond legal regulations.

⁴The outbreak of the coronavirus prevented us from reaching the pre-registered sample size

information influence the effectiveness of interventions that aim at disseminating such information. Our results suggest that providing free access to information without compensating individuals for their opportunity cost of time, for example, is unlikely to generate large effects, as many responsive individuals do not acquire information in that case.

Information avoidance has received considerable attention in the recent literature (see Hertwig and Engel, 2016; Golman et al., 2017; Handel and Schwartzstein, 2018, for reviews). The focus of our paper is on information avoidance in a moral context, i.e., a situation in which an individual's action might have consequences for others. Theoretically, a variety of mechanisms are able to explain why individuals avoid information in such a context. For example, individuals with self-image concerns may decide to avoid information about adverse consequences of their actions in order to sustain a positive self-image despite behaving selfishly (Grossman and van der Weele, 2017). Another explanation is that information avoidance facilitates motivated reasoning without costly self-deception, which may allow individuals to circumvent moral concerns from animal suffering caused by meat consumption (Hestermann et al., 2020). More generally, information avoidance could also occur because the information might trigger negative feelings (Golman et al., 2017), increase the attention paid to an unpleasant belief (Golman et al., 2021), or challenge the individual's current understanding of the world (Chater and Loewenstein, 2016).⁵ Each of these mechanisms might contribute to the information avoidance observed in our experiment.

Empirically, a series of laboratory and online experiments have shown that a substantial percentage of subjects prefer to stay uninformed about the consequences of their action for another player (Dana et al., 2007; Larson and Capra, 2009; Matthey and Regner, 2011; Cain and Dana, 2012; Feiler, 2014; Grossman, 2014; Grossman and van der Weele, 2017; Exley and Kessler, 2021). The standard setting used is a binary choice dictator game in which a subject is initially unsure how a certain action affects the payoff of another player but can reveal all payoffs at no cost. Information avoidance has also been documented for laboratory games in which a subject's action might affect the donation to a charity (Kajackaite, 2015; Felgendreher, 2018; Lind et al., 2019; Momsen and Ohndorf,

of 500 subjects, which reduces statistical power to detect heterogeneity in the laboratory and the field.

 $^{^5 {\}rm For}$ a comprehensive overview of models to rationalize information avoidance, see Golman et al. (2017).

1.1 Introduction

2020; Serra-Garcia and Szech, 2020). We contribute to this literature by testing whether individuals avoid information on the living conditions of animals in intensive farming. In contrast to most previous experiments in a moral context, this information is not only relevant to subjects' behavior within the laboratory but concerns consumption decisions in their everyday life. In that sense, our paper relates to Serra-Garcia and Szech (2020), who show that preferences for moral information in a laboratory game are predictive for the acquisition of information on the living conditions of cows in the dairy industry.⁶

While the presence of information avoidance in moral settings has been studied widely, its consequences on individual decision-making are only partly understood. Some studies find that the opportunity to avoid information leads to more selfish choices on average (Dana et al., 2007; Larson and Capra, 2009; Feiler, 2014; Exley and Kessler, 2021), but others do not detect a significant impact (Felgendreher, 2018; Lind et al., 2019; Momsen and Ohndorf, 2020). We contribute to that literature by exploring the consequences of information avoidance for an important everyday activity, food consumption. The use of complementary data from university canteens enables us to go beyond previous experimental studies that have analyzed subjects' behavior in the laboratory. In particular, we test whether behavioral changes in response to moral information extend from the laboratory to the field.

We implement an experimental design that allows us to estimate both the average response to information and the heterogeneity of information effects by subjects' information demand. Whether and how the selection into treatment interacts with the magnitude of the treatment effect is important for assessing policy implications of interventions (see, e.g., Heckman and Vytlacil, 2001, for a review and Maestas et al., 2013; Cornelissen et al., 2018; Carneiro et al., 2011; Kamhöfer et al., 2018, for applications). More specifically, if information avoiders do not respond to information, their avoidance decision does not reduce the effectiveness of providing information. The converse holds true, however, if avoiders react similarly or even more strongly than information seekers, i.e., individuals who acquire costless information.

How the selection into information relates to the heterogeneity in the effects

⁶In other contexts, d'Adda et al. (2018) show that a majority of subjects in an online survey prefer not to receive information about the environmental impacts of air conditioning usage and Freddi (forthcoming) finds that residents of regions with high refugee inflow access fewer newspaper articles about refugees.

of information is unclear, both empirically and theoretically. So far, empirical studies have found that selfish individuals tend to select out of information (Feiler, 2014; Kajackaite, 2015; Grossman and van der Weele, 2017; Serra-Garcia and Szech, 2020), but heterogeneity in the impact of information has remained unexplored. Theoretically, most information avoidance models that focus on a moral context do not unambiguously predict whether the response to information is stronger for information seekers or avoiders (see Appendix A.2 for a detailed discussion).⁷ Our experimental design allows us to test whether information avoiders react to information on the living conditions of pigs in intensive farming and whether their reaction differs from that of information seekers. In addition, we use our elicitation of subjects' information demand and our estimates on treatment effect heterogeneity to explore how changes in the cost of information determine the impact of information provision. This analysis extends earlier studies that have highlighted the importance of information costs for ignorance (Cain and Dana, 2012; Grossman and van der Weele, 2017; Momsen and Ohndorf, 2020; Serra-Garcia and Szech, 2020) and selfish behavior (Cain and Dana, 2012).

We also contribute to a growing literature on meat consumption. Several studies investigate how people might try to mitigate moral concerns from eating meat (Loughnan et al., 2010; Kunst and Hohle, 2016; Bastian and Loughnan, 2017; Hestermann et al., 2020). In line with our results, survey studies document that people confess to avoiding information on animal welfare (Onwezen and van der Weele, 2016; Bell et al., 2017). The survey of Bell et al. (2017) also includes an information choice task which results in a similar share of information avoiders as in our experiment (31 to 33 percent). Furthermore, Tonsor and Olynk (2011) show that increased media coverage of animal welfare led to lower levels of meat consumption in the U.S. We extend this literature by using an experimental study to elicit the demand for information on animal welfare and to test whether

⁷For example, in the model by Hestermann et al. (2020), information avoiders might reduce their meat consumption in response to news about bad living conditions more or less than information seekers, depending on the cost of self-deception. If information avoiders face a low cost of self-deception, they fully suppress any arriving information. Hence, receiving information neither affects their beliefs about the state of the world nor their meat consumption. If instead, self-deception is rather costly, avoiders only partially suppress information and receiving information affects their beliefs as well as meat consumption. Depending on the shape of the utility function this can lead to a larger adjustment in the level of meat consumption than for information seekers (see Section A.2.1 for a concrete example).

information affects meat-eating behaviors. More broadly, our paper also relates to research on the effect of providing individuals with information about the consequences of their consumption behavior. Examples include labels informing about the environmental impact of products (see Yokessa and Marette, 2019, for a review) or the provision of information on carbon emissions from energy consumption (Newell and Siikamäki, 2014; Andor et al., 2020).

The remainder of the paper is structured as follows. In the next section, we describe the experiment. In Section 1.3, we explain the empirical strategy to identify the average effect of information on meat consumption. We present our results in Section 1.4 and discuss the relevance of information costs for the effectiveness of moral information provision in Section 1.5. Section 1.6 concludes.

1.2 Experiment

To investigate our research questions, we combine a laboratory experiment with field data from university canteens. The experiment provided subjects with the opportunity to receive information about the living conditions of pigs in intensive farming. It took place at the University of Bonn (BonnEconLab) and the University of Mannheim (Mannheim Laboratory for Experimental Economics). In the following, we introduce the information on intensive farming that subjects in our experiment chose to receive or to avoid (Section 1.2.1). We then explain our experimental design (Section 1.2.2) and procedure (Section 1.2.3). Afterwards, we provide details about the data (Section 1.2.4). We pre-registered our experiment at the AEA RCT Registry (Epperson and Gerster, 2020). A detailed mapping between our pre-analysis plan and the analyses in this paper can be found in Appendix A.1.

1.2.1 Information Signal About Intensive Farming

Following Arrow (1996), we refer to information as any signal that is correlated with the true state of the world and thus allows individuals to update their beliefs. In our setting, uncertainty about the state of the world – the living conditions of pigs in intensive farming – is multifaceted. For example, acquiring information about the living conditions requires not only to know about animal regulations in intensive farming but also to understand the implications of these regulations for animal welfare.⁸

As the information signal in our experiment, we used a 360° video about the living conditions of pigs in intensive farming. The video allowed us to provide subjects with a comprehensive depiction of intensive farming practices. It is based on recordings from Germany, Great Britain, Italy, Mexico, and Spain. The original video is 7:27 minutes long and was published by the animal rights organization Animal Equality.⁹ To keep the video within a 5-minute time frame, we presented a selection of scenes, which cover the life of a pig in intensive farming from birth until slaughter. In Appendix A.7, we provide a detailed description of the scenes (Table A.12) and show that they accord with German animal welfare regulations (Table A.13).

Subjects watched the video by using a virtual reality (VR) headset, which allowed them to adjust their line of vision within the video by turning their heads. A key advantage of this technology is that it creates a realistic three-dimensional picture of the scenes and thereby improves subjects' perception of spatial dimensions (Paes et al., 2017; Horvat et al., 2019). In our context, spatial perceptions are important for judging the implications of minimum floor surface requirements for animal welfare, for example. VR headsets are considered a powerful tool to convey information and have been widely used for educational training or data visualization (see Slater and Sanchez-Vives, 2016, for a review). They have also been applied to familiarize individuals with new perspectives (Banakou et al., 2013; Peck et al., 2013; Banakou et al., 2016; Seinfeld et al., 2018).¹⁰ In the video shown to participants, the change of perspectives is implemented by two means. First, the recorded scenes create the impression of standing within a pig farm and seeing the scenes with one's own eyes. Second, the video is accompanied by a narrator who describes the conditions shown in the respective scenes from the "perspective of a pig" (see Table A.12 in Appendix A.7 for the translated transcript).

The information signal in our experiment closely resembles information sig-

⁸By contrast, uncertainty in previous laboratory experiments on information avoidance in a moral context has usually been limited to the financial payoffs for another player. This uncertainty can be resolved by simply revealing the payoffs.

⁹The original video was awarded with the German Web Video award for the best 360° video in 2016 and was also presented at an exhibition at the European Parliament. It is available online: https://www.youtube.com/watch?v=_pC0_mqmp6w (accessed March 19, 2020).

¹⁰Many of these applications create the illusion of inhabiting the body of someone else. This does not apply to the video in our experiment.

nals about intensive farming that are provided by animal protection organizations. Such signals often present information in a frame that is likely to trigger emphatic feelings towards animals, e.g., by describing scenes from the perspective of a pig. The fact that the sender of information provides a particular frame to an information signal is common in real-world settings. Importantly, in our case, the information signal is based on actual recordings from widespread intensive farming practices. Hence, in line with the broad definition by Arrow (1996), we refer to the video about intensive farming as providing *information*. This does not imply that different information signals about intensive farming will lead to the same response or avoidance behavior as the information signal in our experiment.

1.2.2 Design

Our experimental design has three core elements. First, we elicited subjects' willingness-to-pay (WTP) for information on the living conditions of pigs in intensive farming. Second, we introduced random variation in whether subjects receive the information or not. Third, we elicited two outcome variables: subjects' meat consumption in the laboratory and in the university canteens.

To quantify subjects' WTP for information on intensive farming, we followed a multiple price list approach. The key feature of this approach is that each subject chose whether to receive the information at several different prices. One of these prices was randomly drawn and the corresponding choice was implemented. This approach allowed us to elicit the demand curve for information at the individual level. Since each choice was implemented with a positive probability, answering truthfully was incentive-compatible.

To detect active information avoidance, our experimental design ensures that acquiring information did not involve effort, time, or unintended financial costs. In particular, subjects chose between the information about the living conditions of pigs in intensive farming and an (uninformative) outside option. The outside option was to watch another 360° video that subjects had already watched at an earlier stage of the experiment. This video gives a tour of the German central bank building.¹¹ We used this outside option for the following reasons. First, it

¹¹The original video was published by the German central bank and is available online: https://www.youtube.com/watch?v=5S9nArmo_x4 (accessed March 19, 2020). We used a selection of scenes from the video that take exactly 5 minutes to watch.

eliminates the impact of opportunity costs of time that could otherwise prevent subjects from watching the video on intensive farming. Both videos are 5 minutes long, and hence subjects cannot save time by selecting the outside option. Second, by using a video that subjects have watched only minutes earlier, we ensure that the outside option does not convey novel information. In fact, when we test this conjecture based on our data, we find that most subjects considered watching the video again neither as informative nor as entertaining.¹² Third, the outside option also precludes that a desire to experiment with the virtual reality headset biases our WTP measure upwards: Subjects used the VR headset for the same time irrespective of their choice.

We let subjects choose between the two video options at 11 different relative prices of information about intensive farming, $p \in \{-8, -5, -3, -1, -0.5, 0, 0.5, 1, 3, 5, 8\}$ in euro. These prices were implemented by offering subjects varying payments from 0 to 8 euro for watching one of the two videos (see Appendix A.6 for the detailed instructions and a screenshot of the decision screen). The videos were labeled as "Option A" and "Option B" (randomized assignment), with Option B always being presented to the right of Option A. To limit the cognitive burden for subjects, prices for Option A decreased monotonically from the top to the bottom of the multiple price list (Andersen et al., 2006). We tested our price range in two pilot sessions with 22 participants and found that their WTP was always within that range.

Prior to the WTP elicitation task, subjects obtained detailed instructions and read a short description of both videos – the video about intensive farming and the video about the German central bank, which subjects had already watched before. To check whether they understood the WTP elicitation task, they answered a comprehension question and were only forwarded to the decision screen once they had answered the comprehension question correctly. We did not enforce consistency across decisions in the multiple price list. Similar to the procedure in Allcott and Taubinsky (2015), subjects with inconsistent choices

 $^{^{12}}$ After having watched the video about the German central bank for the first time, we asked subjects how informative and how entertaining it would be for them to watch the same video again. Answers were given on a five-point likert scale ranging from "1 – Very informative" to "5 – Not at all informative" and from "1 – Very entertaining" to "5 – Not at all entertaining." The percentage of subjects who selected at least 3 on the likert scale amounts to 90 and 85 percent, respectively. If for some reason watching the video about the German national bank again even creates disutility (e.g., through boredom), our WTP measure explained below provides an upper bound for the WTP for information.

received a hint and could revise their choices once.¹³ When subjects preferred one option over the other at all prices, we asked them to state their hypothetical WTP using an open-ended elicitation format on a subsequent screen.

For each subject, we randomly selected one of the 11 prices from the multiple price list and implemented the subject's choice at this price. The random selection of one price introduced exogenous variation in whether subjects received the information. For example, a subject with a WTP of 6 euro watched the video about intensive farming if and only if the randomly selected price did not exceed 6 euro. We exploit this variation for identifying the average treatment effect of obtaining information (see Section 1.3). The probability of drawing the largest price (8 euro) and the probability of drawing the smallest price (-8 euro) was 27.5 percent. Each of the other prices was drawn with a probability of 5 percent. This design achieves a more balanced treatment assignment and hence improves the statistical power to identify treatment effects compared to drawing each price with the same probability.¹⁴

As outcome variables, we obtained revealed-preference measures of meat consumption from the laboratory and the field. In the laboratory, subjects had a 50 percent chance of winning a voucher for the university canteens. At the end of the experiment, each subject decided whether the voucher should be issued for a meal with or without meat. Both categories were offered daily at the university canteens and we used the standard terminology from the canteens when describing both options.¹⁵ We also provided subjects with a description of all

¹³For example, if a subject selected Option A when the payment for Option A was 0 euro and the payment for Option B was 5 euro but switched to Option B when the payment for Option B decreased to 3 euro, the subject saw the following hint after submitting his choices: "Please consider the following hint. At decision E2 you have chosen Option A and at decision E3 you have chosen Option B, although the payment for Option B is higher at decision E2 than at decision E3. Please reexamine your choices. If you want to, you can revise your choices." If there were several inconsistent switching points, each was addressed in the hint. 2 out of 7 subjects who submitted inconsistent choices in the multiple price list, made consistent choices after receiving a hint.

¹⁴For example, someone whose WTP for information was -6 euro only received the information if a price of -8 euro was implemented. Given our selection probabilities, this happened with a probability of 27.5 percent. If instead, we had used a uniform distribution, the probability would have been about 9.1 percent. We announced the implementation probabilities to the subjects.

¹⁵In Mannheim, the vegetarian meal is called "Menü Vegetarisch" and the non-vegetarian meal is called "Menü 1". Both have a value of 3 euro and often contain a side dish, a soup, and a salad in addition to the main dish. The voucher was valid only in the main canteen. In Bonn, the meal categories are called "Hauptkomponente Vegetarisch/Vegan" and "Hauptkomponente Fleisch/Fisch", respectively. The prices of the two categories vary, usually between 1.35 and

meals offered under the two categories in a pre-defined sample week. To ensure that peer pressure does not affect voucher decisions, we informed subjects that the vouchers would be handed out in an envelope.

In the field, we observed subjects' food purchases at the university canteens before and after the experiment. These purchases were made with electronic payment cards, which are routinely used by students to purchase a variety of university-related services, including food at the university canteens.¹⁶ We were able to track subjects' purchases because subjects provided the number of their electronic payment card during the experiment and agreed to the scientific use of the related data. We describe the detailed procedure in the next section. Details about the field data as well as the construction of our outcome variable(s) for the field are presented in Section 1.2.4.

1.2.3 Procedure

Subjects were invited via email to sign up for an experimental session. The emails included a link to a consent form which informed subjects about the use of VR glasses, data protection rules, and further participation requirements. In particular, we required subjects to be in good physical and psychological health. We collected the signed consent forms before the start of the experiment.

Subjects received a show-up fee of 5 euro. They had a 50 percent chance of winning a voucher for the university canteen and were able to earn up to 8 euro extra, conditional on the outcome of the WTP elicitation task. The voucher was handed out after the experiment and the payment was transferred to subjects' electronic payment card account. We informed subjects about the payment mode and the requirement to possess a valid electronic card in our invitation email.

At the beginning of the experiment, the experimenter instructed subjects in the use of the VR headsets. We used the "standalone" VR headset OculusGo, which does not require a connection to a computer. In addition, every subject obtained headphones which were connected to the VR headset. After

^{2.05} euro and tend to be slightly cheaper for the vegetarian meal. In Bonn, the voucher only paid for the main dish and was valid in two different canteens. The vouchers were always valid for at least two weeks after the experimental session to give subjects enough time to use them.

¹⁶At the University of Mannheim, every enrolled student receives an ecUM (electronic card Universität Mannheim) which also serves as student ID. The MensaCard at the University of Bonn is optional and can be obtained against a deposit of 10 euro.

the instructions, subjects watched a 360° test video for five minutes.¹⁷ The purpose of this video was to provide subjects with a first VR experience and to identify any problems that they may have when using the VR headset.

Throughout the experiment, the experimenter ensured that subjects started and finished videos at the same time, by giving instructions when to put on the headsets. VR headsets were handed out to subjects before they watched a video and collected as soon as the video was over. When subjects put on the VR headset, the video started automatically. Subjects were not able to continue with the experiment until their VR headset was collected.

The main experiment started after the VR instructions and consisted of three parts, which are illustrated in Figure 1.1. The first part was computer-based. Subjects read additional instructions and answered some questions about their familiarity with VR headsets. They were also prompted to enter the number of their electronic payment card on a separate sheet of paper with their participant label, which allowed us to pay out subjects and to link the laboratory and field data. Subsequently, subjects were informed that they were going to watch a 360° video about a virtual tour of the German central bank building and answered questions on that institution. As soon as all subjects had answered the questionnaire, they watched the video and, afterwards, answered questions about the video on the computer. We included "filler questions" about the German central bank and the corresponding video to obfuscate our research question, as for example proposed by Zizzo (2010).

In the second part of the experiment, we informed subjects about the opportunity to watch a video about intensive farming of pigs and asked questions about this topic. For example, we asked how well-informed subjects feel about the living conditions of pigs in intensive farming, how they assess these living conditions, and how the capacity of pigs to feel pain compares to that of humans. The questions were similar in structure to those about the German central bank. In a subsequent step, we elicited subjects' WTP for watching the video about intensive farming of pigs based on our multiple price list approach. After all subjects completed the WTP elicitation task, the computer randomly selected one price from the multiple price list for each subject and subjects watched the

¹⁷The video was a documentary about the tropical rain forest, published by the Federal Ministry for Economic Cooperation and Development, and is available online: https://www.youtube.com/watch?v=5S9nArmo_x4 (accessed March 19, 2020).

Intro	 Introduction to virtual reality (VR) headsets Test video (5min) 		
	\downarrow \downarrow		
Part I	 Questions related to German central bank Video 1 (German central bank) Questions related to Video 1 		
	\downarrow \downarrow		
Part II	 Questions related to intensive farming Elicitation of WTP for information Video 2 (information or Video 1) Questions related to Video 2 		
	\downarrow \downarrow		
Part III	 Questions on demographics and VR headsets Voucher choice 		

Figure 1.1: Illustration of Laboratory Procedure

Notes: The figure illustrates the different parts of the laboratory experiment.

video chosen at this price. Afterwards, subjects answered questions about the video on the computer.

The third part of the experiment consisted of questions on demographics and the VR headset as well as the voucher lottery. After all subjects chose among the vegetarian and the non-vegetarian voucher, the lottery outcome was randomly drawn and presented to each subject on a final computer screen. This screen asked students to come to the desk of the experimenter when their seat number was called. At the desk, the experimenter handed over an envelope with information about the subject's payment and a voucher, if applicable. To activate their payment, subjects had to put their card on an electronic card reader at the main university canteen.¹⁸

The experimental sessions were conducted between November 2019 and March 2020.¹⁹ Each experimental session took about 1 hour and average payouts including the voucher were worth 11.3 euro. We used oTree (Chen et al., 2016) to

 $^{^{18}}$ In Mannheim, subjects could activate their payment on Tuesday in the week after the experiment. In Bonn, activation was possible on Monday two weeks after the experiment.

 $^{^{19}}$ Days with sessions in Bonn: January 15–18, 2020. Days with sessions in Mannheim: November 11–15, 2019; November 18–19, 2019; November 21–22, 2019; November 25, 2019; November 27, 2019; December 5–6, 2019; December 10, 2019; February 24–27, 2020; March 3, 2020.

program the experiment. Our final sample consists of 330 subjects: 126 students from the University of Bonn recruited via hroot (Bock et al., 2014) and 204 students from the University of Mannheim recruited via ORSEE (Greiner, 2015). Since we are interested in subjects' WTP for information and use this measure for our identification strategy of the information effect, our final sample excludes 8 subjects who made inconsistent choices in the WTP elicitation task.²⁰ Due to the coronavirus pandemic, we had to cancel planned sessions, which leaves us with fewer than the 500 participants that we pre-registered as our sample size. We intend to reach that number through additional sessions as soon as experimental laboratories reopen.

1.2.4 Data

As shown in Table 1.1, the average age of subjects in our experimental sample is 22 years. 46 percent of the subjects are female and 29 percent have obtained at least a Bachelor's degree. Furthermore, 62 percent of subjects took part in our experiment in Mannheim, while the remaining 38 percent took part in Bonn. The stated meat-eating habit varies across subjects: 13 percent state to never eat meat and can thus be classified as vegetarians. 51 percent eat meat at least several times per week, while the remaining 36 percent state to eat meat only occasionally, i.e., several times per month or several times per year.

Although vegetarians might be less likely to adjust their meat consumption in response to receiving information, we deliberately include them in our main analysis for two reasons. First, receiving the information could affect the likelihood that a vegetarian eats meat in the near future. Second, the share of vegetarians that (do not) avoid information might influence whether information avoiders are more or less responsive to information than information seekers. We explore how the response to information depends on the stated meat eating habit in Section 1.4.5.

About half of the subjects (173 out of 330) received the information on the

²⁰In addition, 2 individuals accidentally participated without having student status at the respective university, which was a requirement for participation. They were compensated for taking part, but are excluded from our analysis. In our main analysis, we additionally exclude 4 senior students at age 61 or above. They are arguably very different from the rest of the sample. Including these subjects instead does not affect our main results. Another subject indicated an age of 223 years. We expect this to be a typo and impute an age of 23. The results are not sensitive to this observation.

living conditions of pigs in intensive farming. In Appendix A.4, we present the detailed responses to questions related to intensive farming and meat consumption, which were asked before the elicitation of subjects' WTP. On average, the living conditions of pigs were expected to be bad and subjects expected significantly worse living conditions after receiving the information.

We complement our laboratory experiment with data on subjects' food purchases at the university canteens. The observation period runs from August 1, 2019, until March 16, 2020, in Mannheim and from October 1, 2019, until March 19, 2020, in Bonn. The end of the observation periods was determined by the shut down of the university canteens to fight the spread of the coronavirus.²¹ The data includes any food item that was purchased by a subject with his or her electronic payment card at a university canteen. The cards are routinely used by students to make such purchases.²² For the purpose of our study, we are interested in food items that typically constitute a meal, such as a main and side dish, as well as bakery products like sandwiches. We thus exclude desserts, candy, and small snacks like crisps or fruits.²³ Food items purchased with a voucher from our experiment are generally not included in the data set as they were booked separately. In some rare cases, they nevertheless show up as a food item with a price of zero. We drop purchases with such a food item from our field data (19 observations).

As a meal, we consider all food items that are purchased by the same subject within a 20-minute time frame. We use the menus of the canteens to determine whether a meal includes at least one food item with meat. In some cases, we

²¹We planned to include two lecture periods at each university (August 1, 2019, until May 31, 2020, in Mannheim and October 1, 2019, until July 19, 2020, in Bonn). At the University of Mannheim, all canteens were closed as of March 17, 2020. At the University of Bonn, all canteens were closed as of March 20, 2020. Both university canteens established a takeaway service, but only 6 and 4 of our laboratory subjects made use of this service in Mannheim and Bonn, respectively. We exclude the period after canteens were closed, which affects less than 0.6 percent of all purchased food items in the data set.

 $^{^{22}}$ At the main canteen of the University of Mannheim, 97 percent of all transactions during the observation period were paid with an electronic card. At the two main canteens of the University of Bonn, 100 and 54 percent of all transactions during the observation period were paid with an electronic card. In Mannheim, paying with the electronic card is a prerequisite for receiving a student discount.

²³These food categories typically do not contain meat and thus do not confront subjects with the choice of whether to eat meat. In Section 1.4.3, we show that receiving the information signal did not affect the likelihood to purchase food in the considered food categories, which suggests that there was no systematic substitution between included and excluded food categories.

Variable	Mean	S.d.	N
Data from the laboratory			
WTP for information on intensive farming $(euro)^a$	0.32	1.86	312
Age (years)	21.99	3.27	330
Female (1:yes, 0:no)	0.46	0.50	330
Bachelor's degree or higher (1:yes, 0:no)	0.29	0.45	330
Location: Mannheim (1:yes, 0:no)	0.62	0.49	330
Meat-eating habit (stated frequency)			
Never	0.13	0.33	330
Several times per year	0.11	0.31	330
Several times per month	0.25	0.44	330
Several times per week	0.38	0.49	330
Daily	0.13	0.33	330
Received information (1:yes, 0:no)	0.52	0.50	330
Data from the university canteens			
Number of purchases before experiment	19.15	18.07	330
Number of purchases after experiment	10.04	12.22	330
Observation period before experiment (days)	129.22	41.50	330
Observation period after experiment (days)	76.25	39.51	330
At least one purchase after experiment (1:yes, 0:no)	0.84	0.37	330
Share of meals with meat before experiment	0.42	0.31	292
Share of meals without meat before experiment	0.43	0.32	292
Share of ambiguous meals before experiment	0.15	0.22	292
Outcome variables			
Data from the laboratory			
Chose voucher for meal with meat (1:yes, 0:no)	0.37	0.48	330
Data from university canteens			
Share of meals with meat after experiment	0.42	0.35	276
Share of meals without meat after experiment	0.46	0.35	276

 Table 1.1:
 Descriptive Statistics

Notes: ^aThe WTP for information is based on the midpoint of the corresponding WTP interval and only subjects with a WTP that is bounded by the prices in the multiple price list are considered.

are unable to determine with certainty whether a food item contains meat. This occurs, for example, when a food item exists in two varieties that are labeled identically in the data set. Therefore, we construct two different measures of meat consumption for each subject: (i) the share of meals with at least one food item that contains meat for sure and (ii) the share of meals without any food item that may contain meat. The residual category captures the share of meals with no food item that contains meat for sure and at least one food item that may contain meat.

In total, we observe 9634 meal purchases. This is equivalent to 29 meals per subjects, with 19 meals before and 10 meals after the experiment (see Table 1.1).²⁴ The average purchase price amounts to 3.11 euro and closely corresponds to typical price levels for meals at the university canteens (see Figure A.5 in Appendix A.9 for the full price distribution). Only less than 2 percent of purchase prices exceed 7 euro, which is the maximum price for a standard meal. Such rare high prices can occur when students purchase a meal that is priced by its weight (e.g., salads) or when students purchase multiple meals. Our results are robust to excluding purchases that are worth more than 7 euro from the analysis.

After our experiment, we observe at least one meal purchase for 276 subjects (84 percent).²⁵ Table A.14 in Appendix A.8 compares these subjects to those that did not purchase a meal in the period after the experiment. The latter are more likely to be female and to be located in Bonn. As expected, they also purchased fewer meals before the experiment and have a shorter (longer) observation period after (before) the experiment. All other pairwise comparisons, including the likelihood of having received the information about intensive farming, are not statistically significant. In Section 1.4, we conduct robustness checks to show that differences in sample composition across our laboratory and field data do not affect our findings.

 $^{^{24}}$ The larger number of meals before the experiment is partly driven by the longer time period. If we restrict our attention to the period from 13 days before to 13 days after the experiment (the shortest post-experimental period), the number of meals before and after the experiment amount to 2.9 and 3.1. Hence, we do not find evidence that the experiment reduced subjects' propensity to purchase meals in the university canteens (see also Figure A.7 in Appendix A.9).

²⁵Five subjects did not provide a valid electronic card number. Additionally, there are two instances where the same electronic card number was used by two students. Since it is not possible to participate in the experiment twice, a subject might have entered the electronic card number of his or her friend. We link the data from the electronic card number to the subject who participated in an earlier session.
1.3 Empirical Strategy

To introduce our empirical strategy, we use the potential outcomes framework by Rubin (1974). Let Y_i^1 and Y_i^0 denote the potential outcome of individual *i* if she receives treatment ($T_i = 1$) or not ($T_i = 0$), respectively. In our case, treatment corresponds to watching the video about intensive farming of animals and the outcome to a measure of a subject's meat consumption.

We are interested in the average treatment effect, defined as $ATE = E(Y_i^1 - E_i^1)$ Y_i^0). Following the program evaluation literature (Imbens and Wooldridge, 2009), we can identify the ATE if two main assumptions are fulfilled: *unconfoundedness* and overlap. In our application, both assumptions are met as a consequence of our experimental design. Unconfoundedness requires that the potential outcomes Y_i^1 and Y_i^0 are as good as random, i.e., independent of treatment status T_i conditional on a covariate X_i : $(Y_i^1, Y_i^0) \perp T_i | X_i$. In our experiment, a subject receives the treatment only if her willingness to pay for information exceeds the randomly drawn price from the multiple price list. Hence, treatment is as good as random after conditioning on willingness-to-pay: $(Y_i^1, Y_i^0) \perp T_i | WTP_i$. Furthermore, overlap requires that, conditional on WTP_i , the treatment probabilities for individuals in the treatment and control group are larger than zero and smaller than one. This assumption holds for all subjects whose WTP is bounded by the prices in the multiple price list (95 percent of our sample). We know that their theoretical propensity of being treated lies between 27.5 and 72.5 percent. We exclude the remaining 5 percent of subjects with a WTP above 8 or below -8 from our estimation sample, as they are always or never treated, respectively. In these rare cases, treatment effects cannot be estimated without further restrictive identifying assumptions.

For our estimation sample of 312 subjects, we use an inverse probability weighting (IPW) estimator to obtain a consistent estimate of the average treatment effect of information. The idea of this approach is to eliminate differences between the treatment and control group by weighting the observations based on their propensity of being treated. To improve the statistical efficiency of the IPW estimator, we estimate the propensity scores by calculating the empirical treatment probability for each value of the WTP rather than using the true propensity scores that directly follow from the experimental design (Hirano et al., 2003). The IPW estimator can be expressed as

$$\hat{\tau}_{IPW} = \frac{1}{N} \sum_{i=1}^{N} \frac{T_i Y_i}{\hat{p}(WTP_i)} - \frac{1}{N} \sum_{i=1}^{N} \frac{(1-T_i)Y_i}{1-\hat{p}(WTP_i)}$$

$$= \hat{\mu}_{T=1} - \hat{\mu}_{T=0},$$
(1.1)

where WTP_i is the WTP of individual *i*, Y_i is the individual's propensity to eat meat, and $\hat{p}(.)$ is the estimated propensity of being treated, which only depends on the WTP in our experiment. The estimator can be understood as the difference of two terms. The first term estimates the average propensity to eat meat if everyone in the underlying population received the information $(\hat{\mu}_{T=1})$. For this purpose, it weights the outcomes of all treated subjects by their inverse probability of being treated. In particular, subjects who are treated despite having a low propensity of treatment receive a larger weight since they are underrepresented in the treatment group. The second term estimates the average propensity to eat meat if no one of the underlying population would receive the information $(\hat{\mu}_{T=0})$. Here, the weights are the inverse of the probability of not being treated, such that individuals who are underrepresented in the control group receive a larger weight. The propensity score is estimated based on the fraction of subjects with a particular WTP that received the information:

$$\hat{p}(WTP) = \frac{\sum_{j=1}^{N} \mathbb{1}\{WTP_j = WTP\}T_j}{\sum_{j=1}^{N} \mathbb{1}\{WTP_j = WTP\}}.$$
(1.2)

Given our empirical estimator of the propensity score, the IPW estimator presented in equation 1.1 is equivalent to averaging over the estimated conditional average treatment effects at each WTP level.

In total, 161 of the 312 subjects with a bounded WTP received the information in our experiment (see Figure A.6 of Appendix A.9 for the distribution conditional on WTP). Without weighting, the WTP for information is larger in the group of individuals that received information, compared to the group that did not receive information (for details, see Table A.15 in the Appendix). If individuals' WTP for information was correlated with their meat-eating behaviors, this imbalance could confound the comparison of the treatment and control group average. Weighting eliminates this imbalance in the WTP. As our design ensures that treatment probabilities merely depend on subjects' WTP, weighting allows us to rule out the impact of other confounding factors. Nevertheless, small differences in group characteristics can arise from sampling variation. In fact, even after weighting, pairwise comparisons suggest that treated subjects tend to be less likely to hold a degree and tend to eat less meat, as measured by their stated meat consumption as well as their observed meat consumption in the canteens before the experiment.

To account for such differences, we also estimate the ATE based on weighted least squares (WLS) and augmented inverse probability weighting (AIPW). Both estimators allow us to control for individual characteristics, which serves two purposes. First, it improves the precision of our estimates and, second, controls for differences between the treatment and control group that are present even after weighting.²⁶ The WLS estimator regresses the outcome variable on a set of control variables, weighting observations by the inverse probability weights. It coincides with the IPW estimator if only a constant and a treatment dummy are included. The AIPW estimator "augments" the IPW estimator by using the control variables to predict outcome levels for each treatment group based on separate regressions (see, e.g., Lunceford and Davidian, 2004). Since our outcome variables of meat consumption are either binary or fractional, we use logistic regression models. More details of the AIPW estimator are presented in Appendix A.3.1.

The WLS and the AIPW estimators (as well as the IPW estimator) are consistent if the propensity score model is correctly specified (see, e.g., Lunceford and Davidian, 2004; Kang and Schafer, 2007). We can rule out any misspecification of the propensity score model as the treatment assignment is determined by our experimental design. To estimate heterogeneous treatment effects by different groups, we apply the IPW estimator to each group separately or include group-specific treatment and group dummies (WLS). Since allowing for heterogeneous effects is demanding in terms of sample size, we do not apply the more flexible AIPW estimator for that purpose.

For all estimators, we calculate asymptotic standard errors that account for the estimation of the propensity scores based on M-estimation methods (Stefanski and Boos, 2002). In Appendix A.5, we show that our results are robust to using

²⁶The WLS estimator and the AIPW estimator can improve efficiency compared to the IPW estimator particularly in cases where both the propensity score model as well as the included outcome model are correctly specified (Lunceford and Davidian, 2004; Kang and Schafer, 2007).



Figure 1.2: Demand for Information

the true propensity scores instead. Details about the IPW estimator with the true propensity scores are presented in Appendix A.3.2. We also show that results are similar if we use simple OLS regressions to condition on the WTP, instead of weighting the observations based on the propensity scores.

1.4 Results

1.4.1 Demand for Information

Figure 1.2 presents the demand for information in our experiment. The share of subjects who choose to obtain information on intensive farming decreases from about 98 percent at a price of -8 euro to about 4 percent at a price of 8 euro. As hypothesized, we observe information avoidance: when information is costless, about 30 percent of subjects do not want to receive it.

While the demand curve for information is flat at high absolute prices, it is very steep, i.e., price elastic, around zero. When the price of information increases from 0 to 0.5 euro, the share of subjects who choose to obtain information decreases from 70 to 35 percent. At a price of -0.5 euro, i.e., a subsidy for information acquisition, the share of subjects not acquiring the information is cut in half, relative to providing information for free. These results are in line with Cain and Dana (2012), Momsen and Ohndorf (2020), and Serra-Garcia and

Notes: The figure shows the percentage of subjects choosing to receive information about intensive farming at a given price of information from the multiple price list.

Szech (2020), who find that introducing a small monetary incentive or cost can have substantial effects on the level of information acquisition in experimental games. The large price elasticity around zero also emphasizes the importance of holding opportunity costs constant when eliciting the WTP for information.

The median WTP for information is in the interval [0, 0.5) euro. When we use the midpoints of the WTP intervals as estimates for the WTP and focus on individuals with a WTP between -8 and 8 euro, the average WTP for information is 0.32 euro. In Table A.16 of Appendix A.8, we also present correlations between the WTP and observable characteristics. For example, a higher level of meat consumption is associated with a lower WTP for information about intensive farming.

1.4.2 Information Effect in the Laboratory

Table 1.2 presents the estimation results for the average effect of information on the likelihood to choose the voucher that contains meat. Based on the IPW estimator, we find that information on pigs' living conditions in intensive farming reduces the likelihood to choose meat by about 16 percentage points (column 1). Accounting for the meat-eating habit and further control variables reduces the effect size slightly, to 11 to 12 percentage points, but the effect remains economically and statistically significant (columns 2 to 5). Hence, obtaining information on pigs' living conditions significantly reduced meat consumption in the laboratory. The effect size is substantial given that the estimated average likelihood to choose a voucher for a meal with meat in the absence of information amounts to 42 to 45 percent (see Table 1.2). The finding is robust to focusing on the sample that is used to estimate the information effect in the field and controlling for the baseline meat consumption levels from the university canteens instead of the stated meat-eating habit (see Table A.19 in Appendix A.8).

1.4.3 Information Effect in the Field

To estimate the average effect of information on meat consumption in the field, we use (i) the share of purchased meals that certainly contain meat and (ii) the share of purchased meals that certainly do not contain meat as the outcome variable for each subject. These outcome variables measure the intensive margin of adjusting food choices when eating at the university canteens. We focus on the intensive

	Dependent variable: Choosing voucher for meal with meat						
	IPW	W	LS	AIPW			
	(1)	(2)	(2) (3)		(5)		
Info effect	-0.156^{***} (0.055)	-0.116^{**} (0.054)	-0.119^{**} (0.050)	-0.111^{**} (0.046)	-0.107^{**} (0.045)		
Mean~(w/o~info)	0.450	0.430	0.431	0.421	0.424		
Meat-eating habit Additional controls Observations	No No 312	Yes No 312	Yes Yes 312	Yes No 312	Yes Yes 312		

 Table 1.2:
 Information Effect in the Laboratory

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2 to 5, we use indicator variables. Since never eating meat is a perfect predictor of choosing the vegetarian voucher, the expected likelihood for individuals with this meat-eating habit is not predicted via a logit model but directly set to zero for the models in columns 4 and 5. Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

margin as we do not detect any effects on the likelihood of purchasing food at the university canteens (see Table A.17 in Appendix A.8). In the main specifications of the WLS and AIPW estimators, we directly control for the baseline level of our outcome variable instead of subjects' stated meat-eating habits.²⁷

Table 1.3 presents the estimation results for the average information effect in the field. Based on our IPW estimator, we find that receiving information reduces the propensity that a purchased meal contains meat (Panel A) by 12 percentage points on average. The estimated reduction amounts to 6 to 7 percentage points based our WLS and AIPW estimators that additionally control for covariates, such as baseline meat consumption, and is statistically significant at the 10 percent level throughout. Controlling for baseline levels allows us to account for the fact that, due to sampling variation, treated subjects were already slightly less likely to eat meat before the experiment (as discussed in Section 1.3).

Panel B of Table 1.3 shows our results for the alternative outcome variable, i.e., the propensity that a purchased meal contains no meat. The estimates mirror the results from Panel A: Receiving information significantly increases the

²⁷Results are robust to using the stated meat-eating habit as control instead (see Table A.18 in Appendix A.8). For the few subjects who purchased a meal at the canteens after but not before the experiment (N = 12), we simply impute the average baseline level of subjects with the same stated meat-eating habit.

	IPW WLS		VLS	AII	PW
	(1)	(2)	(3)	(4)	(5)
A. Share of meals with meat					
Info effect	-0.124***	-0.063*	-0.066**	-0.064**	-0.066**
	(0.042)	(0.033)	(0.032)	(0.032)	(0.031)
Mean (w/o info)	0.480	0.454	0.455	0.455	0.454
B. Share of meals without meat					
Info effect	0.147^{***}	0.089^{**}	0.089^{**}	0.087^{***}	0.087^{***}
	(0.041)	(0.044)	(0.041)	(0.033)	(0.033)
Mean (w/o info)	0.374	0.399	0.399	0.398	0.401
Baseline level	No	Yes	Yes	Yes	Yes
Additional controls	No	No	Yes	No	Yes
Observations	261	261	261	261	261

 Table 1.3:
 Information Effect in the Field

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

propensity to purchase a meal that does not contain meat by 15 percentage points for our IPW estimator and about 9 percentage points for our WLS and AIPW estimators. Taken together, these estimates imply that the effect of information extends to the field.

Our data also allows us to explore the dynamics of the information effect in the field (not pre-specified). To account for the fact that most subjects did not purchase food in the canteens daily, we define our outcome variable as the percentage of purchased meals that certainly contain meat (or not) in a moving time window of seven days. We focus on the first 19 days after the experiment as the number of observations in the moving time window is stable during this period (see Figure A.7 in Appendix A.9). The decline of observations thereafter can be explained by the semester breaks and the fact that university canteens were closed in response to the outbreak of the coronavirus. We also estimate treatment effects in a moving 7-day window during the 14-day period before the experiment as a placebo test.

Figure 1.3 plots the treatment effects against the day that represents the midpoint of the 7-day window. The results are based on the WLS estimator and thus account for covariates and meat consumption before day -14. As expected, we do not observe any significant treatment effect before the experiment. In the first days after the experiment, the treatment effect is statistically significant



Figure 1.3: Effect of Information over Time

Notes: The figure shows the average effect of receiving information on the propensity to eat a meal with meat (Panel A) and a meal without meat (Panel B) over time. Day 0 represents the day of the experimental session. The effect is estimated based on a moving time window of seven days with the day on the x-axis as the midpoint (± 3 days). Around day 0, the time window captures only those days before or after the experiment. We use the WLS specification and control for covariates and the meat consumption before day -14 (missing values are imputed as described in Section 1.4.3). To estimate propensity scores, we merge WTP groups only consisting of treated or untreated subjects with neighboring WTP groups.

and comparable to the effect observed in the laboratory. However, the effect diminishes rapidly and about one week after the experiment, the point estimates are close to zero and no longer statistically significant at any conventional level. As the number of observations within the moving 7-day window is stable during the considered period, the decline in effect sizes is unlikely to reflect changes in sample composition.

The dynamics of the information effect look similar if we extend the considered post-experimental period (see Figure A.8 in Appendix A.9), directly compare the evolution of the average meat consumption by information status (see Figure A.9 in Appendix A.9), or estimate the effect of receiving information for the first three days after the experiment, the first seven days after the experiment, and the period more than one week after the experiment (see Table A.24 in Appendix A.8).²⁸ Hence, our results suggest that the effect of receiving information is particularly strong in the first days after the experiment but does

²⁸We do not detect any statistically or economically significant treatment effect on the extensive margin (see Figure A.10 in Appendix A.9 and Table A.25 in Appendix A.8). Furthermore, Table A.26 in Appendix A.8 shows that the pattern of a diminishing effect over time becomes stronger if we restrict our attention to subjects who are observed to purchase food during as well as after the first week of the post-experimental period.

not persist over time. A potential explanation for the lack of persistence is that individuals gradually suppress information, as found by Zimmermann (2020) for negative feedback on relative IQ-test performance.

1.4.4 Heterogeneity by Information Demand

The results presented in the previous sections show that (i) a substantial percentage of subjects avoid information on the living conditions of pigs in intensive farming and that (ii) receiving information significantly reduces the likelihood to consume meat, in the laboratory as well as in the field. To assess the consequences of information avoidance, we now investigate whether the information effect varies with the demand for information. We allow the information effect to differ between individuals who avoid information when it is costless (WTP < 0) and those who acquire costless information ($WTP \ge 0$).

Table 1.4 presents the estimation results for the laboratory. We find that information has a large impact on the meat consumption of avoiders. Receiving the information reduces their likelihood to choose the meat voucher by about 23 percentage points (see columns 2 and 3). In contrast, our point estimates for information seekers are considerably smaller in magnitude. Whether the difference between the information effects is statistically significant depends on the model specification. While we can reject the null hypothesis of homogeneous effects for the IPW estimator, the *p*-value for the comparison based on the WLS specification is just above 10 percent ($p \approx 0.12$).²⁹ A stronger effect for avoiders corresponds to a selection pattern "out of responsiveness." It implies that interventions that merely provide free access to moral information will only have modest consequences: Those individuals most susceptible to changing their behavior upon receiving this information will choose to avoid it.

Results from the field are less conclusive, which may be explained by the fact that the field sample is smaller than the one from the laboratory. We only find a significant information effect for avoiders in the case of the IPW estimator (see column 1 of Table A.20 in Appendix A.8). When we control for baseline levels the significance vanishes and differences in the information effect between avoiders and seekers are not statistically significant. The sensitivity to the inclusion of control variables is mainly driven by the fact that even after weighting,

²⁹The lack of significance might be driven by the fact that the outbreak of the coronavirus prevented us from reaching the pre-registered sample size of 500 participants.

	Dependent variable: Choosing voucher for meal with meat			
	IPW	WLS		
	(1)	(2)	(3)	
Avoider				
Info effect	-0.341^{***} (0.100)	-0.231^{***} (0.086)	-0.233^{***} (0.082)	
Mean $(w/o info)$	0.581	0.532	0.528	
Seeker				
Info effect	-0.077 (0.065)	-0.065 (0.064)	-0.071 (0.062)	
Mean (w/o info)	0.394	0.386	0.389	
Comparison of effects $(p-value)$	0.03	0.12	0.12	
Meat-eating habit Additional controls	No No 212	Yes No	Yes Yes	
Observations	312	312	312	

Table 1.4: Information Effect in the Laboratory – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Indicator variables are used to control for the meat-eating habit in columns 2 to 5. Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

information avoiders who receive information have a lower baseline tendency to eat meat than information avoiders who do not receive information.³⁰

Taken together, the patterns observed in the field and the laboratory suggest that avoiders are not less affected by information on intensive farming than information seekers.

1.4.5 Exploratory Heterogeneity Analyses

Beyond the analyses that we have pre-specified, our comprehensive data set allows us to conduct further exploratory analyses along two dimensions. First, we explore whether the information effect depends on the meat-eating habit of the subjects. Similar to Hestermann et al. (2020), we divide subjects into three different categories based on their reported frequency to eat meat. *Vegetarians*

³⁰We also check whether the results from the laboratory change if we only consider subjects who are considered for the estimation of the information effect in the field and use the baseline levels from the field instead of the stated meat-eating habit as a control variable. The difference between information avoiders and seekers becomes less pronounced but the point estimate for information avoiders remains larger in absolute terms (see Table A.21 in Appendix A.8).

1.4 Results

	Laboratory Meal with meat		Field				
			Meal w	Meal with meat		v/o meat	
	(1)	(2)	(3)	(4)	(5)	(6)	
Omnivores							
Info effect	-0.207^{**} (0.086)	-0.201^{**} (0.082)	-0.087^{*} (0.049)	-0.106^{**} (0.048)	0.104^{**} (0.044)	0.119^{***} (0.044)	
Mean (w/o info)	0.726	0.724	0.658	0.666	0.224	0.219	
Flexitarians							
Info effect	-0.027 (0.074)	-0.041 (0.070)	-0.023 (0.063)	-0.026 (0.062)	$0.062 \\ (0.094)$	$0.060 \\ (0.086)$	
Mean (w/o info)	0.158	0.162	0.278	0.281	0.548	0.545	
Vegetarians							
Info effect	$0.000 \\ (0.000)$	-0.016 (0.028)	-0.051 (0.039)	-0.032 (0.040)	$0.120 \\ (0.105)$	$0.114 \\ (0.109)$	
Mean (w/o info)	0.000	0.008	0.082	0.070	0.698	0.701	
Meat-eating habit Additional controls Observations	Yes No 312	Yes Yes 312	Yes No 261	Yes Yes 261	Yes No 261	Yes Yes 261	

Table 1.5: Information Effect by Meat-Eating Habit

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meateating habit, we use indicator variables. Additional controls are age, gender, degree, location, and WTP. All estimates are based on the WLS estimator. In columns 1 and 2, the dependent variable is whether the individual chose the voucher for the meal with meat. In column 3 and 4, the dependent variable is the share of meals with meat. In columns 5 and 6, the dependent variable is the share of meals without meat.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

never eat meat (13 percent of subjects), *flexitarians* eat meat several times per month or year (37 percent of subjects), and *omnivores* eat meat several times per week or on a daily basis (51 percent of subjects). Table 1.5 presents the estimation results from WLS regressions for the laboratory and the field. The results suggest that the information effect is primarily driven by omnivores. The absolute information effect for this group ranges from about 21 percent in the laboratory to about 9 percent in the field and is always statistically significant at the 10 percent level or lower. The information effects for the other two groups are usually much smaller and never statistically significant at any conventional level. Furthermore, we can reject the null hypothesis that omnivores and vegetarians have the same information effect in the laboratory (p < 0.05).³¹

Second, we investigate the responses to information on the living conditions

 $^{^{31}}$ We cannot reject the null hypothesis that the information effect for omnivores is different from that for flexitarians at any conventional significance level.

of pigs for two separate meat categories: meat from terrestrial animals, such as pork and chicken, and meat from aquatic animals, such as fish and calamari. Our results show that the point estimates for both categories are almost indistinguishable (see Table A.22 in Appendix A.8), which suggests that subjects adjust their consumption of meat from terrestrial as well as aquatic animals. Hence, subjects may update their beliefs about animal welfare in meat production more generally rather than only for terrestrial animals or pigs. To further disentangle such effects, we divide the meat from terrestrial animals in pork and meat from other terrestrial animals (see Table A.23 in Appendix A.8). The point estimates support the impression that the change in meat consumption is not limited to pork.³² Hence, our findings offer suggestive evidence that subjects do not only change the consumption of the meat on which information was provided.

1.5 Discussion

In this section, we discuss the relevance of information avoidance for decisionmaking and explore the consequences of information provision policies that change the cost of information. In practice, policy makers can change the cost of information by providing free access to information or by subsidizing information acquisition, for example. The two crucial factors for this exercise are the percentage of subjects who select into information at a given cost and the response of these subjects to information.

Our experimental design allows us to identify both. We first estimate separate effects of receiving information for subjects that do and those that do not select into information at a given cost. For this purpose, we use the WLS estimator and control for the stated meat-eating habit (laboratory data) or baseline consumption (field data). The estimation results are shown in Figure A.11 and Figure A.13 in Appendix A.9. We then multiply the information effect for subjects who select into information with their share in our study population to obtain the effect of providing information at a given cost on the average level of meat consumption.

We discuss the average treatment effect of information on meat consumption

³²One caveat to these heterogeneity analyses is that analyzing the effect for each meat category reduces the statistical power to detect significant effects if the aggregate behavioral adjustment actually comes from several categories (as it seems to be the case).

1.5 Discussion

in our study population for various scenarios that differ in the cost of information. The reference scenario for the effects is always that the information is not available or available only at a prohibitive cost. As before, we focus on subjects whose WTP for information is bounded in the range of -8 to 8 euro. For meal purchases at university canteens, results are based on subjects who purchased a meal at the university canteens after the experiment. We focus on the laboratory results in the main discussion and emphasize differences compared to the field results at the end of the section.

The first scenario that we consider is characterized by no information acquisition cost, except for the opportunity cost that subjects face from acquiring information. This scenario allows us to assess the consequences of information technologies, such as the Internet, that make information available at no direct cost. Using the hourly average payout in our experiment (11.3 euro) as a measure of subjects' value of time, the opportunity cost of watching the 5-minute video is about 1 euro. At a cost of 1 euro, only about one-fifth of all individuals watch the video (bottom panel of Figure 1.4). Furthermore, we find that the average reduction in our study population's meat consumption is negligible and amounts to about 2 percentage points (top panel of Figure 1.4). These results are a consequence of the steep information demand curve that prevents about 80 percent of the study population from demanding information at this cost.

Our next scenario explores the implications of compensating individuals for their opportunity cost of time, e.g., through a 1 euro subsidy for watching the video. In this case, the cost of information is zero and the percentage of subjects who choose to receive information increases strongly to 70 percent while the average reduction in meat consumption reaches 5 percentage points. This finding demonstrates that opportunity costs are a major obstacle for individuals to access moral information.

Next, we consider the consequences of reducing the cost of information slightly to -0.5 euro. Such reductions might result from a small financial reward for acquiring information or from requiring individuals to incur some effort to avoid information. In the context of a real-world information campaign, such effort could consist of taking another (more distant) entry to a supermarket or avoiding contact to information campaign staff more generally. Compared to the previous scenario, a decrease in the cost of information to -0.5 euro increases the percentage of subjects obtaining information to about 86 percent. More im-





Panel B. Information acquisition



Figure 1.4: Effect of Information Provision on Meat Consumption in the Laboratory

Notes: Panel A shows the average effect of making information available at a given cost. The outcome variable is whether a subject chooses the voucher for a meal with meat. The effect at a given cost c is calculated as $ATE(WTP \ge c) \frac{1}{N} \sum_{i=1}^{N} \mathbbm{1}\{WTP_i \ge c\}$, where WTP_i denotes the willingness-to-pay of subject i, $ATE(WTP \ge c)$ denotes the average treatment effect of receiving information for subjects with a willingness-to-pay for information at least as large as the cost, $\mathbbm{1}\{\cdot\}$ denotes the indicator function, and N denotes the number of subjects. Estimates of $ATE(WTP \ge c)$ are based on the WLS estimator, allowing for heterogeneous information effects based on the selection into information at a given cost and controlling for the stated meat-eating habit. Panel B shows the fraction of subjects obtaining information at a given cost, $\frac{1}{N} \sum_{i=1}^{N} \mathbbmmmatula \{WTP_i \ge c\}$. Only individuals whose WTP for information is bounded by -8 and 8 euro are considered.

portantly, we find that the average treatment effect from access to information at that cost rises sharply in magnitude and amounts to -10 percentage points. This increase reflects our finding that information avoiders tend to have larger treatment effects than information seekers. Hence, subtle encouragements, such as small financial incentives, can be highly effective in changing behaviors as they push highly responsive individuals into receiving information.

To gauge the impact of strong encouragements, we consider a scenario where the cost of information reduces to -8 euro. At this cost, all subjects choose to obtain information and the average information effect amounts to -12 percentage points. Very low costs of information could be achieved by large subsidies, but also by interventions that substantially increase the cost for individuals to not obtain information, e.g., through salient labels on food products or the inclusion of the respective information into school curricula. Despite the large change in the cost of information, the reduction in meat consumption is only slightly more pronounced than in the previous scenario with a much smaller encouragement (-12 instead of -10 percentage points). This finding suggests that strong interventions that mandate information acquisition may not be substantially more effective than subtle encouragements.

Our results remain similar when we analyze the scenarios based on our field data. A minor difference is that reducing the cost of information below zero has only little or no additional impact on the average treatment effect (see Figure A.12 in Appendix A.9).

1.6 Conclusion

Our results shed light on the role of moral information for everyday choices. We consider the context of food consumption, where the decision to eat meat from intensive farming entails negative consequences for animal welfare. In our experiment, about 30 percent of subjects avoid information on pigs' living conditions in intensive farming when it is costless. This level of information avoidance is substantial but lies within the range that has been observed in laboratory and online experiments where the information is about the payoff to another subject or a charity (e.g., Dana et al., 2007; Larson and Capra, 2009; Feiler, 2014; Kajackaite, 2015; Grossman and van der Weele, 2017; Serra-Garcia and Szech, 2020; Exley and Kessler, 2021). In particular, these studies have found avoidance rates

from 9 to 72 percent.

To analyze the impact of information provision on consumption choices, we combine our laboratory experiment with data from university canteens. We find that receiving information on the living conditions of pigs reduces the level of meat consumption in the laboratory as well as in the field. The reduction amounts to about 12 percentage points in the laboratory and 6 to 9 percentage points in university canteens. These magnitudes are large, given that the treatment only consists of a 5-minute video about intensive farming practices. The effect of information reduces considerably over the course of the first week after the experiment, which could reflect that individuals suppress information over time (Zimmermann, 2020). Hence, a single informational intervention may not lead to the formation of a new habit and additional interventions may be necessary to achieve persistent behavioral change (see, e.g., Ito et al. 2018 for a discussion of the persistence of moral suasion).

Furthermore, our research design enables us to show that information avoiders, i.e., subjects who avoid information when it is costless, are at least as responsive to information as information seekers, who demand costless information. In a subsequent step, we use our information demand measure and heterogeneous treatment effect estimates to assess the consequences of information provision under varying information acquisition costs. We find that merely providing free access to moral information is likely to only have a small effect on behaviors. The reason for this finding is that a substantial percentage of individuals do not acquire costless information even though they would change their behavior if they received it. However, our results also show that small encouragements to receive information can be sufficient to motivate many responsive individuals to acquire information and thereby strongly increase the effectiveness of information provision.

Our findings have immediate policy implications. On the one hand, they demonstrate that information provision about the consequences of one's actions for others can effectively alter everyday behaviors but that providing information once might be insufficient to create persistent change. On the other hand, they show that an important policy challenge is to target those individuals who choose to remain uninformed when information is costless. Hence, the mere deployment of information technologies that reduce information acquisition costs, such as the Internet, is not a panacea for fostering prosocial behaviors. Instead,

1.6 Conclusion

additional incentives to acquire information might be crucial. In practice, such encouragements may take the form of small financial rewards for obtaining information. For example, many health insurance companies reward their customers for taking up pre-emptive medical screening and getting informed about their health status. Rewarding individuals for acquiring information in the context of prosocial behavior might be more difficult, however. An alternative strategy for policy makers would be to increase the cost of information avoidance, e.g., by displaying information prominently on product packages.

More broadly, our findings demonstrate the importance of assessing how the responsiveness to information relates to individuals' information preferences. If not accounted for, the selection of responsive individuals out of information could undermine the effectiveness of a wide range of policies that inform about the consequences of own behavior in order to foster prosocial actions, including policies to fight climate change, poverty, or a pandemic.

Chapter 2

How to Design the Ask? Funding Units vs. Giving Money^{*}

with Johannes Diederich and Timo Goeschl

2.1 Introduction

Donation calls in which the potential donors are asked how many units of a charitable good they wish to fund are a frequently used solicitation scheme among fundraising practitioners. A prominent example that has attracted about one million donors from all over the world is ShareTheMeal, a smartphone app and initiative of the United Nations World Food Programme which is used to provide food to children in need. Donors for ShareTheMeal are informed that feeding one child for a day costs $\in 0.70$ and are then asked to indicate the number of feeding days ("meals") that they would like to fund ("share"). Over 88 million meals have been provided through the organization's app so far. Unit donation schemes are not only implemented in food programs. Development aid agencies, for example, promote child sponsorships by fixing the monthly donation for the sponsorship – usually around \$35 – and prospective donors choose the number of

^{*}This chapter has been published as a working paper (Diederich et al., 2021). We are grateful to Mark Ottoni-Wilhelm and Jan Schmitz for helpful comments, and the people at Sign of Hope e.V. for cooperation. We also thank participants at the 6th Science of Philanthropy Initiative Conference, Chicago, the 4th Workshop Experimental Economics for the Environment, Muenster, the 25th EAERE Annual Conference, the EEA Virtual 2020, the VfS Annual Conference, and the ESA 2020 Global Online Around-the-Clock Conference for their helpful remarks.

child-months to sponsor rather than the amount of money to donate. UNICEF Canada goes beyond this and provides a whole online shop for specific charitable goods, such as a set of measles vaccines for \$16, a teacher training for \$114, or a water pump for \$492. Similarly, fundraising drives for biodiversity conservation or reforestation programs let donors indicate the number of acres or trees to fund. For instance, in the Monarch Butterfly Habitat Exchange program of the Environmental Defense Fund, donors sponsored acres of milkweed habitat for \$35 per acre. In the Plant A Tree program of the Jewish National Fund, donors were asked to choose the number of trees to be planted at \$18 a tree.

The prevalence of such schemes in fundraising must reflect a belief among practitioners that they can outperform alternative schemes in particular circumstances. This difference in expected performance needs to outweigh the need for the charity to make more information available to the donor and the restriction on how the raised funds can be used. The belief in better performance may well be justified: Research has shown that seemingly small changes in the choice architecture of a decision problem can lead to substantial changes in donor behavior. Examples are the announcement of seed money (List and Lucking-Reiley, 2002; Huck et al., 2015), specifying default amounts (Goswami and Urminsky, 2016; Altmann et al., 2019), suggesting donation levels (Edwards and List, 2014; Reiley and Samek, 2019), and providing information about the efficacy of the donation (Latour and Manrai, 1989; Cryder et al., 2013). However, to the best of our knowledge, no previous study has systematically compared a solicitation scheme in which the charity asks potential donors to fund units – which we refer to as a "unit donation" scheme – to the traditional scheme of "simply asking for money" (Landry et al., 2010) – which we refer to as a "money donation" scheme.

This paper seeks to close this gap in the literature by comparing both schemes under controlled conditions in an online donation experiment with real money at stake. In their purest forms, the two schemes differ along three dimensions: Unit donation schemes (i) frame the choice in terms of physical units of the charitable good instead of money, (ii) restrict the choice to complete units (i.e. the donation to multiples of the price per unit), and (iii) provide information about the effectiveness of a donation (i.e. the price per unit). The restriction to complete units reflects that some charitable goods are indivisible as a matter of nature (planting half a tree or donating half a coat) or of choice (offering half a meal). Stating the price per unit is necessary for the potential donor to calculate her expenses. Our experimental variations reflect these three dimensions: Subjects in the money donation treatment are simply asked how much money they would like to donate for the provision of food to malnourished children. Subjects in the unit donation treatments are asked how many nutritional rations they would like to fund at a given price, without the possibility to fund fractions. We compare two different unit sizes: A smaller sized unit of the ration that feeds one child for one day (price of \$0.50) and a larger sized unit that feeds one child for one week (price of \$3.50). In addition to these pure implementations of the money and unit donation schemes, we implement three intermediate forms that selectively activate one or two of the three dimensions that distinguish the two schemes.

Our experiment delivers three main findings. The first is that the money and unit donation schemes were, on average, equally effective in raising donations: There is no statistically significant difference in average donations between the two schemes. This is surprising in light of the higher demands that the unit donation scheme places on the charity. If our results hold more generally, this means that practitioners of unit donation schemes either hold erroneous beliefs about their fundraising effectiveness or use it to pursue other objectives than maximizing the size of the average donation.

Our second finding highlights a plausible alternative objective implicit in unit donation schemes. In our experiment, the unit donation scheme increased the propensity to become a donor when the unit size was small: For the one-day ration at a price of \$0.5, the share of donors was about 13 percentage points higher than for the baseline money donation scheme. This is a statistically and quantitatively significant increase in the propensity to give. An appropriately designed unit donation scheme has therefore the potential to recruit more donors than a money donation scheme. Such a recruitment is likely to be valuable to fundraisers in its own right: Previous research has shown that it is easier to reactivate prior or lapsed donors compared to "cold calling" an unselected sample (Eckel and Grossman, 2008a; Landry et al., 2010). Unit donation schemes therefore have a plausible role in growing a charity's donor base.

Our third finding is that unit size matters: In our experiment, the larger-sized unit (one-week ration at \$3.50) reduced the propensity to become a donor by 22 percentage points compared to the pure money donation scheme, a statistically and quantitatively significant amount. Larger unit-sizes therefore deter donors at the extensive margin. Our intermediate treatments indicate that this decrease can be traced back to discretizing the donation choice rather than to framing donations in units or to informing about the effectiveness of a donation. In lights of these results, we conclude with the hypothesis that unit donation schemes with small unit sizes decrease a possible stigma associated with "penny donations" or similarly small donation sizes.

2.2 Experimental Design

2.2.1 Donation Appeal

The experiment consisted of a real donation ask administered to subjects during an unrelated online survey. Designing the ask as a pure money donation scheme, a pure unit donation scheme, and various intermediate schemes, requires a charitable good or service readily divisible into discrete and meaningful units. We partenered with a relief organization, Sign of Hope e.V., which frequently uses intermediate schemes of unit donation calls in their own fundraising campaigns. Among their activities, we chose the treatment of malnourished children with a special nutritional paste and high energy cookies in a bush clinic in South Sudan. This service offered practicable units and prices for our experiment. The associated expenses amounted to \$0.50 per day or \$3.50 per week. These benchmarks provided the two different unit sizes for the experiment: (i) a *one-week* nutritional ration per child at a price of \$3.50 and (ii) a *one-day* nutritional ration per child at a price of \$0.50.

The first part of the donation appeal in the experiment was uniform across all treatments. It introduced the charity, the charitable good (treatment of malnourished children with special nutrition in a hospital in South Sudan), and the charitable cause (a high need due to decades of civil war in the country and hence, a high incidence of malnourishment).¹

The second part of the donation appeal was treatment specific. The six experimental treatments (see Table 2.1) were designed to (i) compare contributions under the unit and money donation scheme, (ii) investigate whether the size of a unit matters for this comparison and (iii) disentangle the channels through which differences may arise. The three treatments of type A ("pure schemes") address

¹We also provided a link to the charity's web page and informed about a transparency certificate the charity holds to increase trust in the charity (Adena et al., 2019).

aspects (i) and (ii). They consist of a treatment with a pure money donation scheme, in which subjects are simply asked how much money they would like to give, and two treatments with a pure unit donation scheme. In the latter, subjects were asked how many rations they would like to fund (one-week and one-day rations, respectively), had their choice restricted to whole units, and learned the price of a unit (\$3.50 and \$0.50, respectively). Subjects entered their desired amount of money or number of nutritional rations in an input field at the bottom of the solicitation screen. We provide the exact wording of the donation appeal for each treatment in the Appendix.

The three treatments of type B ("intermediate schemes") address aspect (iii). They include a strict subset of the three characteristics that jointly make up a unit donation scheme (unit framing, restricted choice set, price information). This allows us to identify the channels through which potential differences in giving behavior between the pure schemes arise. Specifically, we conduct an unrestricted money donation scheme that provides unit price information (Info), a money donation scheme restricted to multiples of a disclosed unit price (Info + Restricted), and a unit donation scheme with perfectly divisible units (Info +Frame). The three intermediate schemes are implemented only for the case of the one-week ration because we expect larger treatment effects for the larger-sized The three intermediate schemes deliberately leave out three additional unit. possible combinations of the three characteristics because they do not have a meaningful real-world counterpart. There are, for obvious reasons, no schemes that ask for units to fund without also informing the donor about the unit price; and schemes do not arbitrarily restrict the choices of monetary amounts to multiples of some unit price unless information on the unit price is given.²

2.2.2 Experimental Protocol

We conducted the experiment online recruiting U.S. residents from the online labor market Amazon Mechanical Turk (AMT).³ In the posted task, we informed

²To illustrate, imagine a donor who decides how much money to give. If the donor is informed that for 3.50 the charity can provide a nutritional ration that feeds one child for one week, a restriction of donations to multiples of 3.50 will most likely seem reasonable to the donor. However, if this information is not given such a restriction will probably appear arbitrary to the donor.

³AMT is known to provide several benefits to researchers, among them fast and easy access to subjects, a diverse subject pool, and low costs (Paolacci et al., 2010; Mason and Suri,

Treatment	Framing	Choice set	Price info	Unit size
A. Pure schemes Money Donation Unit Donation – Large Unit Donation – Small	Money Units Units	Unrestricted Restricted Restricted	No Yes Yes	Large Small
B. Intermediate schemes Info Info + Frame Info + Restricted	Money Units Money	Unrestricted Unrestricted Restricted	Yes Yes Yes	Large Large Large

Table 2.1: Treatments

Notes: The table provides an overview of the different treatments in our experiment. Framing: whether the ask is framed in terms of money or physical units. Choice set: whether the choice set is restricted to complete units of the charitable good (multiples of the unit price when the donation is framed in money). Price info: whether information about the unit price of the charitable good is provided. Unit size: Whether one physical unit is a one-day ration of food (small) or a one-week ration of food (large).

workers that they would earn \$7 for answering a 20-minute academic survey on several topics, including demographics, occupational background, religion, and opinions about some political and societal challenges. Interested workers followed a link to the survey on LimeSurvey. Before the start of the survey, workers read and confirmed a consent form about the research study.

The experimental survey consisted of 22 questions on sociodemographics, employment, religious beliefs, and political attitude before subjects encountered the donation ask, and 12 unrelated questions after the call. One of the treatments was drawn at random and presented to the subject (between-subjects design). The survey ended with five manipulation check questions. After completing the survey, subjects received a unique code that had to be entered into the survey task window on AMT for payment.

In total, 900 subjects completed the survey experiment. We chose the sample size to be able to pick up significant differences in mean donations starting at about 5% of the endowment, according to power calculations based on data from a pilot experiment (80% power and 5% significance level). The concern that some

^{2012).} Regarding data quality, several papers highlight a high internal consistency of self-reported demographics, an incentive-compatibility of earnings, and a "spammer"-free workforce from the built-in reputation system (Ross et al., 2199; Mason and Suri, 2012). They also provide evidence that results from standard experimental games successfully replicate on AMT (e.g. Paolacci et al., 2010; Rand, 2012). In implementing our experiment, we followed the suggestions for researchers in that literature and the Guidelines for Academic Requesters on AMT (WeAreDynamo, 2014).

Variable	Mean	SD	N
Female	0.54	0.50	842
Age (years)	37.11	10.59	847
Has children	0.46	0.50	844
College graduate	0.49	0.50	845
Crowdworker	0.25	0.43	834

 Table 2.2: Summary Statistics

Notes: The total sample consists of 848 observations. We did not force subjects to answer the sociodemographic questions in the survey. The variable *crowdworker* indicates whether online crowdworking is the subject's primary source of income.

subjects may fraudulently use multiple accounts to participate more than once is generally seen as a minor problem in online experiments (Horton et al., 2011; Paolacci et al., 2010).⁴ We nevertheless follow the common approach to exclude subjects with duplicate Internet Protocol addresses from the analysis. Including them does not change the results. This leaves us with a sample of 848 subjects. Average payouts were \$5.87 excluding donations. Subjects took on average 10.1 minutes to complete the experiment.

2.3 Results

Table 2.2 shows summary statistics for the sample that participated in our experiment. Our average subject is slightly more likely to be female than male, has an average age of 37.1 years, and has children and a college degree with a probability of 46% and 49%, respectively. About 25% of subjects state that online crowdworking is their primary source of income. To check the balance of these sociodemgraphic variables across the experimental groups, we regress each variable on treatment dummies and conduct F-tests for the joint significance of the coefficients. The corresponding p-values are reported together with the group means in Appendix Table B.1. The test results suggest that the randomization led to experimental groups that are balanced on all sociodemgraphic variables (p-values between 0.25 and 0.93).

Table 2.3 reports the share of donors and the mean donation (including non-

⁴In the case of AMT, having multiple accounts is forbidden by Amazon's Terms of Service (Mason and Suri, 2012) and creating an account requires a unique credit card number (Paolacci et al., 2010).

Treatment	Share of donors	Mean donation	N
A. Pure schemes			
Money Donation	0.47 (0.50)	0.95(1.53)	152
Unit Donation - Large $(p = \$3.50)$	0.26(0.44)	1.16(2.14)	121
Unit Donation - Small $(p = \$0.50)$	$0.60\ (0.49)$	1.07 (1.62)	146
B. Intermediate schemes			
Info $(p = \$3.50)$	0.43 (0.50)	1.12(1.86)	150
Info + Unit frame $(p = \$3.50)$	0.38(0.49)	1.15(2.06)	132
Info + Restricted $(p = \$3.50)$	0.27(0.44)	1.34(2.10)	146

Table 2.3: Donations

Note: Standard deviations in parentheses.

donors) in each of the six treatment groups. For the treatments with unit framing, the mean dollar donation corresponds to the average number of nutritional rations donated times the unit price. Mean donations vary between \$0.95 for the money donation scheme and \$1.34 for the money donation with unit cost information and a restricted choice set. The variance within each treatment is considerable and donations range from the lower limit of \$0.00 (no donation) to the the upper limit of \$7.00. The share of donors varies between 26% for the large-sized and 60% for the small-sized unit donation scheme. The benchmark of the standard money donation is at 47%. These descriptive statistics point to possibly significant extensive-margin effects across solicitation schemes.

Figure 2.1 presents the cumulative distribution of donations for the three pure scheme treatments. The solid black line refers to the 152 donation decisions under the standard money donation scheme. As reported in Table 2.3, we see that slightly more than half of the subjects chose not to donate to the charity. For positive donations, there are clear focal points of contributions at full dollar amounts and, less pronounced, at half dollars. In other words, donors do not make use of the unrestricted nature of the donation space, with some exceptions between \$0.00 and \$1.00. Unsurprisingly, lower money donations are more frequently observed than higher ones.

The dashed blue line shows the cumulative distribution of 121 donation decisions under the large-sized unit donation scheme. There are only three possible donation levels under this scheme: No donation (\$0.00), one week of nutrition (\$3.50), or two weeks of nutrition (\$7.00). 74% of subjects chose not to donate under this scheme while 18% chose to provide one week of nutrition and about



Figure 2.1: Cumulative Distribution of Donations in the Pure Scheme Treatments

7% to provide two weeks. The dashed orange line presents the cumulative distribution of 146 donation decisions under the small-sized unit donation scheme. There are fifteen possible donation levels for subjects, ranging from zero to fourteen days of nutrition. Here, around 40% of subjects chose not to donate. For positive donations, slightly more than 16% chose to provide a single day of nutrition at a cost of \$0.50 and almost 23% chose two days. More days of nutrition are less common and their relative frequency under the small-sized unit donation scheme visually does not differ much from that under the money donation scheme.

We proceed in Section 2.3.1 by first comparing the mean donations (including non-donors) across treatments. Afterwards, we investigate the behavior of potential donors at the extensive margin (Section 2.3.2).

2.3.1 Donations

We observed in Table 2.3 that the differences in the mean donations between treatments are small (at most 5.6% of the endowment of \$7) and that the within-treatment variances are high. More specifically, the differences between the pure money and pure unit donation treatments amount to \$0.21 for the one-week and to \$0.12 for the one-day rations. Both differences are insignificant in a two-sided *t*-test (p = 0.380 and p = 0.546, respectively).

	Pure schemes		Pure and intermediate schemes with large unit size			
	(1)	(2)	(3)	(4)	(5)	(6)
Unit Donation – Large	$0.202 \\ (0.231)$	0.217 (0.232)	_	—	_	_
Unit Donation – Small	$\begin{array}{c} 0.110 \\ (0.183) \end{array}$	$\begin{array}{c} 0.104 \\ (0.182) \end{array}$	_	_	_	—
Price information	_	_	$\begin{array}{c} 0.216 \\ (0.185) \end{array}$	$\begin{array}{c} 0.237 \\ (0.186) \end{array}$	$\begin{array}{c} 0.167 \\ (0.196) \end{array}$	$\begin{array}{c} 0.180 \\ (0.197) \end{array}$
Unit frame	_	_	$\begin{array}{c} 0.117 \\ (0.175) \end{array}$	$\begin{array}{c} 0.149 \\ (0.177) \end{array}$	$\begin{array}{c} 0.221 \\ (0.237) \end{array}$	$\begin{array}{c} 0.270 \\ (0.239) \end{array}$
Restricted choice set	—	_	-0.070 (0.174)	-0.111 (0.176)	$\begin{array}{c} 0.029 \\ (0.228) \end{array}$	$\begin{array}{c} 0.005 \\ (0.232) \end{array}$
Unit frame \times restricted	—	—	_	—	-0.215 (0.351)	-0.251 (0.357)
$\begin{array}{c} \text{Controls}^a \\ \text{Observations} \\ \text{R}^2 \end{array}$	No 419 0.002	Yes 408 0.032	No 701 0.004	Yes 686 0.018	No 701 0.004	Yes 686 0.019

Table 2.4: Effect on Donations (OLS Regression)

Notes: Robust standard errors are in parentheses, *p < 0.1, **p < 0.05, ***p < 0.01. The dependent variable is the amount of money the charity receives and the pure money donation scheme always serves as baseline. In columns 1 and 2, only the treatements with pure solicitation schemes are considered. In columns 3 to 6, all treatments except the unit donation scheme with a small unit size are considered.

^aControls include gender, age, whether the individual has a college degree, whether the individual has children and whether online crowdworking is the individual's primary source of income.

In columns 1 and 2 of Table 2.4, we report estimation results from regressing the monetary amount donated on the treatment dummies, using the pure money donation scheme as baseline and only considering the pure solicitation schemes. Even after controlling for available covariates, the differences between the pure solicitation schemes remain very small and insignificant (see column 2). Thus, we do not find evidence that applying a unit donation scheme instead of a money donation scheme significantly affects the average amount of money received, irrespective of whether a small or large unit size is employed.

For the large unit size, we use our intermediate treatments to identify how specific characteristics of a unit donation scheme affect giving. In particular, we regress individuals' donations on a dummy for each of the three characteristics of a unit donation scheme (price information, unit framing, and a restricted choice set). This allows us to test whether the insignificant difference between the unit and money donation scheme masks countervailing effects of single characteristics. The estimation results are reported in columns 3 and 4 of Table 2.5. We do not find any evidence for such countervailing effects: All coefficients are small and insignificantly different from zero. The same holds true if we additionally include the interaction of using a unit frame and restricting the choice set (see columns 5 and 6), which can be identified due to the selection of our intermediate treatments.

2.3.2 Propensity to Give

As already noted, the differences at the extensive margin merit attention. While in the pure money scheme, about 47% of subjects donate, only about 26% of subjects decide to give under the pure unit scheme with a one-week ration at price of \$3.50 as single unit (p < 0.001, χ^2 -test). If the unit presented to subjects is instead a one-day nutritional ration at a unit price of \$0.50, the propensity to donate is about 13 percentage points higher than under the pure money donation scheme (p = 0.026, χ^2 -test).

Regression results from a linear probability model are presented in Table 2.5 and confirm these findings.⁵ In columns 1 and 2, we regress the binary variable of whether an individual donated on the type of the solicitation scheme, only considering pure schemes and using the pure money donation scheme as baseline. A unit donation scheme with a large unit size (a one-week ration at a price of \$3.50) is estimated to decrease the propensity to give by about 22 percentage points compared to a pure money donation scheme, whereas a unit donation scheme with a small unit size (a one-day ration at a price of \$0.50) is estimated to increase the propensity to give by 11 percentage points when including controls. Hence, applying a unit donation scheme affects the propensity to give, but the direction of the effect depends on the unit size.

Why do we find such substantial effects on the extensive margin but no significant differences in mean donations? As Figure 2.1 reveals, the large-sized unit donation scheme decreases the share of individuals who donate but also encourages individuals to choose a higher donation level than they would have chosen under a money donation scheme. For example, the mass of individuals who give more than \$1.00 but less than \$3.50 under the money donation scheme seems to entirely shift to the donation level of \$3.50. In the case of the small-sized unit donation scheme, the positive impact on the extensive margin does not translate into substantially higher mean donations since the increase is mainly driven by

⁵Results are robust to using a probit model instead.

	Pure schemes		Pure and intermediate schemes with large unit size			
	(1)	(2)	(3)	(4)	(5)	(6)
Unit Donation – Large	-0.217^{***} (0.057)	-0.219^{***} (0.057)	—	—	—	—
Unit Donation – Small	0.129^{**} (0.057)	$\begin{array}{c} 0.113^{**} \\ (0.057) \end{array}$	_	_	_	_
Price information	_	_	-0.050 (0.054)	-0.050 (0.054)	-0.040 (0.057)	-0.041 (0.058)
Unit frame	_	_	-0.033 (0.040)	-0.027 (0.041)	-0.055 (0.059)	-0.047 (0.059)
Restricted choice set	_	—	-0.146^{***} (0.040)	-0.157^{***} (0.040)	-0.166^{***} (0.055)	-0.177^{***} (0.056)
Unit frame \times restricted	_	_	_	_	$\begin{array}{c} 0.044 \\ (0.080) \end{array}$	$\begin{array}{c} 0.043 \\ (0.081) \end{array}$
$\frac{\text{Controls}^a}{\text{Observations}}$ R^2	No 419 0.077	Yes 408 0.109	No 701 0.032	Yes 686 0.045	No 701 0.033	Yes 686 0.046

Table 2.5: Effect on the Propensity to Give (Linear Probability Model)

Notes: Robust standard errors are in parentheses, *p < 0.1, **p < 0.05, ***p < 0.01. The dependent variable is whether a subject has donated and the pure money donation scheme always serves as baseline. In columns 1 and 2, only the treatements with pure solicitation schemes are considered. In columns 3 to 6, all treatments except the unit donation scheme with a small unit size are considered.

 a Controls include gender, age, whether the individual has a college degree, whether the individual has children and whether online crowdworking is the individual's primary source of income.

additional small donations of one unit, i.e. \$0.50. Beyond donations of \$0.50, the cumulative distribution function looks similar to that of the money donation treatment.

Due to the inclusion of the intermediate treatments, we are able to attribute differences in the propensity to give between the pure money and the large-sized unit donation scheme to a particular characteristic. Analogously to the procedure in Section 2.3.1, we regress the binary variable of whether a subject donated on a dummy for each of the three characteristics. The estimation results are reported in columns 3 and 4 of Table 2.5. Neither the unit frame nor the information on the effectiveness of a donation significantly affects the propensity to give. By contrast, restricting the choice set significantly reduces the propensity to give by about 15 percentage points. In columns 5 and 6, we additionally allow for an interaction between the restriction of the choice set and the framing in units. The impact of restricting choices is slightly but not significantly smaller in absolute terms if a unit frame is already in place (p = 0.59). These results are robust to controlling for sociodemographic characteristics and are consistent with pairwise comparisons.

2.4 Discussion and Conclusion

Donation schemes can be designed in terms of physical units to fund rather than the amount of money to give. Does this design of the ask affect individuals' giving behavior? The popularity of unit donation schemes among fundraisers suggests that it should, and that the scheme's performance justifies the complications of the design, such as additional information provision and the reduced freedom in how the funds can be used. To address this research question, we conducted an online experiment in which we tested different solicitation schemes. While we do not find evidence that unit donation schemes affect the amount of money raised, we show that they alter the propensity to give. The direction of this effect depends on the unit size of the charitable good. If the unit size is small, a unit donation scheme attracts more donors than a competing money donation scheme. If the unit size is large, unit donations can deter donors. A unit donation scheme with a small unit size can therefore be an effective strategy for a charity to expand its donor base.

An interesting question is why the effect on the extensive margin reverses when the unit size becomes small. The negative impact of larger-sized units on the extensive margin of giving, relative to the money donation scheme, is unsurprising. In fact, from the intermediate treatments, we can pinpoint that it is the restriction of the choice set that is mostly responsible for the drop in donors. Since the price of becoming a donor increases from almost zero (\$0.01) in a virtually continuous choice set to \$3.50 in a restricted choice set, this demand side response is expected. However, the same logic would apply to the smallersized unit. There, the price of becoming a donor increases from almost zero (\$0.01) in a virtually continuous choice set to \$0.50 (the price of a one-day nutritional ration). As a result, we would expect the magnitude of the effect at the extensive margin to be smaller, but the predicted direction would be the same.

While we do not have intermediate treatments for the small unit size that would allow us to disentangle the exact characteristic responsible for that increase, a closer look at the distributions of the donations under the pure solicitation schemes (see Figure 2.1) offers a plausible explanation. As observed earlier, the distribution of donation amounts under the money donation scheme has focal points at \$0, \$1.00, and \$2.00. This suggests that a substantial share of subjects behaves as if the range of donations available is restricted to integers of dollar amounts. In such a world, the minimum donation, and hence the *perceived* price of becoming a donor, is \$1.00. Offering to fund a charitable good with a price below the smallest focal point of the unrestricted distribution of giving reduces the perceived price of becoming a donor, resulting in a higher propensity to give. Another possible explanation is that the small unit size acts like a low suggested donation amount. Edwards and List (2014) show that suggesting an amount that is below the average donation can increase the propensity to give. Finally, in a world in which "penny donations" carry a stigma or are considered unproductive, a smaller-sized unit justifies small donation amounts. These explanations are in line with the observation that the positive effect on the extensive margin is mainly driven by additional small donations of \$0.5 (as discussed in Section 2.3.2).

Our results from the intermediate treatments (for the large unit size) offer interesting insights into the relevance of framing, choice restrictions, and information provision. First, simply rephrasing the ask from giving money to funding units does not affect giving behavior. Second, we show that restrictions of the choice set can have large behavioral consequences that should be taken into account when designing experiments or fundraising campaigns. This evidence is in line with the finding that a minimum donation amount reduces the propensity to give in the absence of extrinsic incentives to give (Cartwright and Mirza, 2019). It also matches well with the result that large suggested donation amounts discourage giving (Adena et al., 2014). Finally, providing explicit information about the per unit price of the charitable good did not significantly affect the propensity to give or overall donations in our experiment. Previous experiments have used different formats when providing information about the effectiveness of a donation and either did not find an impact on the aggregate (Karlan and Wood, 2017) or identified a significant increase in donations (Latour and Manrai, 1989; Cryder et al., 2013).

An interesting avenue for future research is to explore the role of the unit size in more detail. While we show that the unit size matters for the impact of applying a unit donation scheme, it is unclear whether the effect on the extensive margin monotonically increases with the unit size. Furthermore, it would be interesting to know at which point the effect reverses and to which extent it depends on the distribution of donations under unrestricted choices.

Chapter 3

Subsidizing Unit Donations: Matches, Rebates, and Discounts Compared^{*}

with Johannes Diederich, Catherine C. Eckel, Timo Goeschl, and Philip J. Grossman

3.1 Introduction

Subsidies are a common way of incentivizing charitable giving. They typically take the shape of rebates, in which a third party (e.g., the government) refunds a fraction r of the gift back to the donor; or the shape of matches, in which a third party (e.g., a generous donor) supplements each donation at a rate m, such that the charity receives a total of (1 + m) times the original donation. Both rebates and matches have been extensively studied and several key findings have emerged in the literature (see Vesterlund, 2016; Epperson and Reif, 2019,

^{*}An earlier version of this chapter has been published as a working paper (Diederich et al., 2020). We thank René Bekkers, Christian Conrad, Mark Ottoni-Wilhelm, and Konrad Stahl for their helpful comments. We also thank participants at the 5th Science of Philanthropy Initiative Conference, Indianapolis, the 24th Annual Conference of the European Association of Environmental and Resource Economists, Manchester, the 34th Annual Congress of the European Economic Association, Manchester, and the Annual Conference of the German Economic Association, Leipzig, as well as seminar and workshop audiences at Heidelberg University, the University of Bonn, the University of Mannheim, the Centre for European Economic Research Mannheim, and the Karlsruhe Institute of Technology. We thank Sign of Hope e.V. for cooperation and Woodrow Ahn for language assistance in developing the donation question.

for comprehensive reviews). Probably the most notable result is that although rebates and matches imply the same price of giving if the corresponding subsidy rates r and m satisfy $r = \frac{m}{m+1}$, overall donations received by the charity are higher under matches than under equivalent rebates (Eckel and Grossman, 2003, 2006a,b, 2008a, 2017; Davis et al., 2005; Lukas et al., 2010; Bekkers, 2015; Gandullia and Lezzi, 2018; Gandullia, 2019).¹ Another finding is that matching subsidies often significantly increase private contributions net of the subsidy compared to a no subsidy condition without a lead donor (Eckel et al., 2007; Karlan and List, 2007; Gneezy et al., 2014; Huck et al., 2015; Eckel and Grossman, 2017).²

The literature has established these findings in a setting in which individuals are asked to decide how much money to give to a charity, arguably the most common scheme for soliciting donations. We refer to this as a *money donation* scheme. Yet another frequently applied strategy is to frame the donor's choice variable not in terms of money, but in terms of physical units of a charitable good awaiting funding. A prominent example that has attracted donors from all over the world is *ShareTheMeal*, a smartphone app and initiative of the UN World Food Programme, which is used to provide food to children in need. Donors for ShareTheMeal do not simply choose an amount of money to give. Instead, they are informed that feeding one child for a day costs \$0.80 and are then asked to indicate the number of feeding days ("meals") that they would like to fund.³ We refer to this alternative scheme as a *unit donation* scheme.

Do the key findings about the effects of matches and rebates in money donations generalize to the alternative unit donation scheme? In this paper, we examine the effect of subsidies on unit donations by conducting an online field experiment. We asked 558 subjects how many units of a charitable good they

¹In this literature, rebates are realized without any delay. If a delay is involved (as is the case for tax deductions), time preferences need to be considered. Furthermore, we are aware of only one paper that finds the same level of charity receipts between rebates and matches, but it does so under a choice architecture that does not resemble a typical donation decision: In Davis (2006), subjects do not decide how much to donate (checkbook giving) but how much the charity receives (charity receipts). This choice architecture, motivated by an investigation of the causal mechanism that underpins the standard result of non-equivalence, makes it difficult to compare his results with ours.

 $^{^{2}}$ There are counterexamples to this finding, however (e.g., Karlan et al., 2011; Helms Mc-Carty et al., 2018).

³Similar food provision campaigns are the "100 Thousand Meals" appeal of the Salvation Army Australia or the "Help with $\in 2$ " campaign of Misereor, the German Catholic Bishops' Organisation for Development Cooperation.

would like to provide to a predetermined charity (Sign of Hope e.V.), funded out of their reward for answering an unrelated online survey. The decision variable is framed in quantities of nutritional packages provided for malnourished children. The unsubsidized price is 0.50 per package. In the baseline, no subsidy is offered. The main treatments differ across three subsidy types and two subsidy rates. The first type, the rebate, is offered at a rate of either 33% or 50% such that a third party refunds to the subject about one third or one half of the reward she spent on nutritional packages. The second type, the match, is offered at a rate of 0.5 (1:2) or 1 (1:1) such that a third party adds a nutritional package for either every two or each package donated. The third subsidy type is novel for charitable giving and takes the form of a price discount of either 33% or 50%. This subsidy type is without a direct parallel in money donations.

The contribution of this paper is threefold. First, we define unit donations as a separate class of charitable donations distinct from money donations. Second, we investigate how rebates and matches perform in a setting of unit donations and compare the results to the established literature on money donations. Based on between-subjects evidence, our core result is that matches and rebates are equally effective in incentivizing private net donations and in generating total charity receipts. In other words, we do not replicate the superiority of matching subsidies observed in the case of money donations. Third, we check whether, in a setting of unit donations, the discount subsidy offers an attractive alternative to these subsidy types. We find that discounts are equally effective as matches and rebates when considering net donations or charity receipts. This may well be good news for charities: In a world in which subsidy types perform equally well, charities enjoy additional degrees of freedom in campaign design. At the same time, the different subsidy types perform differently when disaggregated into the extensive and the intensive margin of giving: Rebates are more effective than matches in attracting donors, but matches result in larger donations. Under discounts, the likelihood to give is lower than under rebates, and on both margins, behavior corresponds to that under matches. We conclude that if attracting donors is a secondary objective of a fundraising campaign that uses unit donations, rebates merit additional consideration. New donors offer the possibility of an ongoing income stream for charities, since previous donors are more likely to give in the future (Eckel and Grossman, 2008a; Landry et al., 2010).

The remainder of the paper is organized as follows. In a background section

(Section 3.2), we contrast money and unit donations, explain the mechanics of subsidizing the latter, and review the relevant literature. Section 3.3 describes our experimental design, followed by a presentation of our main results (Section 3.4) and a discussion of potential explanations (Section 3.5). Section 3.6 concludes.

3.2 Background

3.2.1 Unit vs. Money Donations

For our purposes, we define a money donation scheme as a solicitation scheme in which potential donors are asked to decide how much money to give to a charity. It is arguably the most common scheme for solicting donations. Academic papers in the lineage of the now classic donation models (Bergstrom et al., 1986; Andreoni, 1988, 1989) capture its main features by generally assuming a linear production technology for the charitable public good and normalizing the perunit price of both the private and the public good to one. In such models, the prospective donor i's choice is to divide her endowment w_i (in dollars) between private consumption x_i (in dollars) and giving g_i (in dollars) to the charitable good, G. Under a money donation scheme, therefore, the donor's choice variable g_i is denominated in terms of monetary expenditures.

By contrast, we define a *unit donation* scheme as a solicitation scheme that frames the donor's choice variable in terms of physical units of a charitable good awaiting funding. Unit donation schemes have a popularity that extends beyond the food programs mentioned above. Development aid agencies, for example, promote child sponsorships by fixing the monthly donation for the sponsorship – usually around 35 – and prospective donors choose the number of child-months to sponsor rather than the amount of money to donate. Similarly, fundraising drives for biodiversity conservation or reforestation programs let donors indicate the number of acres or trees to fund.⁴ In unit donation schemes, the price of a unit of the charitable good G is no longer implicit. Instead, the fundraiser states an explicit price p and asks how many discrete units g_i the potential donor would

⁴For instance, in the *Monarch Butterfly Habitat Exchange* program of the Environmental Defense Fund, donors sponsor acres of milkweed habitat for 35 per acre. In the *Plant A Tree* program of the Jewish National Fund, donors choose the number of trees to be planted at 18 a tree.
3.2 Background

like to fund. In this respect, the setting resembles early models of the private provision of public goods that are explicit about units and prices (e.g., Warr, 1983). Under a unit donation scheme, therefore, the donor's choice variable g_i is denominated in terms of the quantity of the charitable good funded.

Although under both schemes donors eventually provide money, there are important differences between unit and money donations. First, donors' choice sets differ. Under a unit donation scheme, the units of the charitable good to be provided are typically indivisible, which introduces an element of discreteness that is largely absent in the virtually continuous money donations. Second, the information provided to prospective donors differs. By stating the per-unit price of the charitable good, unit donation schemes make statements about the charity's marginal cost of production, whereas money donation schemes frequently provide little information on the cost structure of producing the charitable good. While information on the share of fundraising and overhead costs is increasingly available to donors (Ribar and Wilhelm, 2002; Meer, 2014), information on the impact of a contribution (or the absence thereof) can substantially affect donations (Bekkers and Wiepking, 2011; Lewis and Small, 2019). Third, the framing of the choice differs. By asking for the number of physical units of the charitable good, unit donation schemes emphasize how a donation generates specific outcomes for recipients. As a result, the motive of giving to create an impact (Duncan, 2004) might become more relevant for the donation decision.

Diederich et al. (2020) compare the two donation schemes in an experimental study and show that the choice of the donation scheme significantly affects the likelihood of receiving donations. The direction of the effect depends on the size of a physical unit: A unit donation scheme attracts more donors than the equivalent money donation scheme if the unit size is small (daily nutritional rations at a price of \$0.50) but fewer donors if the unit size is large (weekly rations at a price of \$3.50). For the large unit size, the difference is primarily driven by the restricted choice set under the unit donation scheme.

3.2.2 Subsidizing Unit Donations

Subsidizing unit donations involves some small but important differences compared to subsidizing money donations. In unit donations, rebates can be applied by refunding a fraction of the donor's provision costs back to the donor. If, for example, a unit of the charitable good costs \$0.50 and a 50% rebate is offered, the donor receives \$0.25 back for each unit funded. Matches can be applied to unit donations by providing supplementary units of the charitable good. If, for example, a 1:1 match is offered for tree plantings, the third party funds one additional tree for each tree funded by the donor. Due to the indivisibility of units, matching payments by the third party are restricted to complete units of the charitable good. This introduces some discontinuity in the matching payment if the matching rate is not an integer: For example, at a matching rate of 0.5 (1:2) every second tree funded by the donor induces one tree funded by the third party. However, for a donation of only one tree, there is no additional funding by the third party. This is in contrast to the continuous choice in money donations, in which the matching rate of 0.5, a donation of any dollar amount induces a matching payment of 0.5 times this amount).

The transferability of results from money to unit donations is therefore not only a matter of framing effects: When matches consist of supplementary units and rebates are refunded costs, rebates and matches are also no longer theoretically equivalent. This is particularly evident at the extensive margin of becoming a donor: The smallest positive donation is to fund one unit of the charitable good. Given a unit price p, this implies a minimum expense of p required under matches. In contrast, rebates provide a refund on the donation given and, at subsidy rate of r, the cost of becoming a donor is p(1-r) < p. As a result, rebates are potentially more effective in attracting donors. An additional difference comes into play when subsidy rates take non-integer values: The change in the matching payment due to a one unit increase in the donation depends on the donation level. In contrast, under rebates any increase in the donation proportionally increases the subsidy payment, as is the case for both subsidy types under money donations. In sum, there are not only structural differences between money and unit donations; there are also reasons to expect that subsidies perform differently under the two schemes.

In a way, unit donation schemes resemble the shopping experience for private goods. For example, WorldVision provides a comprehensive gift catalog where donors can choose the number of units of various gifts that are associated with explicit prices.⁵ In this regard, a matching subsidy is similar to bonus packs or

⁵https://donate.worldvision.org/giftcatalog (accessed on March 31, 2021)

"buy one get two" offers, whereas a rebate is comparable to coupons that provide an instant refund at checkout or cash-back programs (although the latter usually involve a time delay between the payment and the refund). However, with a well-defined good and an explicit price, there is a third option available, which is a direct price discount. Similar to stores that advertise price reductions, a charity can announce the discounted price of a gift and reveal that the gap to the original price is provided by a large donor or a governmental grant. Hence, if a unit of the charitable good costs \$0.50 and a 50% discount is offered, the donor can fund one unit at a price of \$0.25 while being informed that the remaining \$0.25 are funded by an external party.

A discount at rate d is theoretically equivalent to a rebate at rate r. However, two small differences exist that could cause different behavior.⁶ First, rebate subsidies are paid to the donor whereas discount subsidies are paid to the charity. Second, in comparison to rebates (and matches), discounts obviate the need for donors to calculate the effective price of giving. In an online charity gift shop like the one by WorldVision, the rebate would take effect as instant refund upon checkout whereas the discount is applied to the advertised prices during "shopping." Furthermore, evidence for private goods shows that consumers may indeed respond different to rebates and direct price discounts (Davis and Millner, 2005), which makes it crucial to distinguish between both subsidy types in our research.

3.2.3 Related Literature

We are not aware of any previous study that conducts a clean comparison between subsidy types under a pure unit donation scheme. At the same time, there are parallels with a number of papers studying charitable giving. Like our study, Meier (2007) and Gneezy et al. (2014), for example, feature discrete choice sets. However, both frame donations in money, rather than physical quantities, and focus on matches only, yielding results that align with the wider money donation literature. A different parallel is with Lewis and Small (2019) who also provide subjects with information about the cost of a unit of impact and test

⁶If rebates are realized with a delay, a third difference comes into play: The expenses at the time of the donation are larger under a rebate than under a discount. In this paper, rebates are realized without delay as it allows us to compare our rebate to its money donation counterpart in the experimental literature that brought about the seminal result of matches outperforming rebates (e.g., Eckel and Grossman, 2003, 2006b; Davis et al., 2005).

different framings of the information. They find that a cheaper unit price leads to lower donations, an effect that is eliminated or reversed if the price is framed in units-per-dollar rather than dollars-per-unit. Yet, donations in their study are again framed in terms of money, rather than physical quantities, and the authors do not compare different subsidy types. Also relevant is a literature in marketing that experimentally compares product promotion strategies such as bonus packs (which are similar to matching subsidies) and price promotions (which are similar to our discount subsidy if they explicitly state the effective price). The papers in this literature provide mixed evidence (see, e.g., Sinha and Smith, 2000; Mishra and Mishra, 2011; Chen et al., 2012), with bonus packs either being superior, equivalent, or inferior to price promotions. More in line with the money donations literature are Davis and Millner (2005), who find that matches outperform rebates also for private goods and that direct price discounts have an effect in between. While our focus on charitable giving sets our paper apart from this literature, its setting of unit donations offers the opportunity to study price discounts, a tool from private product promotion, in the context of charitable donations.

The paper probably closest to the focus of ours is Kesternich et al. (2016). The authors compare the effectiveness of rebate and matching subsidies in the context of carbon offsetting: When buying their ticket(s) online, clients of a long distance bus operator decide whether to offset the carbon emissions from their travel at a given price per kilogram of emissions. Rebates are found to increase the likelihood to offset while matches only do so at certain matching rates and to a lesser extent. However, the overall contributions net of the subsidy are higher under matches. Key differences to our study are the binary decision format and the use of an impure public good for which the size of giving is tied to the private good. Both limit the comparability to our setup. A few other studies implicitly employ an experimental design soliciting unit donations to an environmental public good (Löschel et al., 2013; Diederich and Goeschl, 2014, 2017, 2018), but they do not compare subsidy types.⁷

⁷Weakly related to a unit donation scheme are so called "buy-one give-one" business models (see, e.g., Marquis and Park, 2014; Hamby, 2016) where for each product purchased the selling company donates a similar product. However, in these models, the donation is tied to the consumption of a private good. We are not aware of any paper introducing or comparing subsidies in that context.

3.3 Experimental Design

3.3.1 Donation Appeal

We adapt the real-donation dictator game introduced by Eckel and Grossman (1996) and subsequently applied to compare subsidy types (Eckel and Grossman, 2003; Davis and Millner, 2005; Davis, 2006; Eckel and Grossman, 2006a,b). In the standard version of the game, subjects decide how much of their money endowment to hold and how much to pass to a charity. In our variant of the game, subjects decide how many units of the charitable good to fund at a given nominal price.

Our variant of the game requires a charitable good or service that is easily quantifiable. We approached a relief organization, Sign of Hope e.V., which frequently uses various forms of unit donation schemes in their fundraising campaigns. Among their activities, we chose the treatment of malnourished children in a certain area of South Sudan as this service offered practical units and prices for our experiment. The children were treated in two "bush clinics" operated by the relief organization at the time of the experiment. Treating one child for one month using a special nutritional paste and high energy cookies requires a donation of US\$15. We divided this number into practical units of nutritional packages per child and day, which implies a "price" of \$0.50 per package.

The donation appeal was part of an online survey and participants used their reward for completing the survey (\$2) to make any donations. The donation appeal introduced the charity, the charitable good, and its marginal provision cost to the charity. We also provided a link to the charity's web page and informed subjects about a transparency award the charity had won to increase trust in the charity (Adena et al., 2019). The final part of the donation appeal was treatment specific. Table 3.1 shows the seven treatment conditions. In the control condition, no subsidy was applied and subjects chose how many packages to fund at a price of \$0.50. The remaining six treatment conditions follow a 3×2 factorial design with one factor being the subsidy type (rebate, match, or discount) and the other factor being the effective price (\$0.33 or \$0.25) implied by the level of the subsidy. In the instructions, we framed the rebate conditions as 33% (50%) rebate and stated that while providing packages would cost the subject \$0.50 apiece, a rebate of \$0.17 (\$0.25) per package would be added to the subject's final reward at the end of the experiment. For the matching conditions,

Subsidy type	Subsidy rate	Nominal unit price	Effective unit price	Ν
No subsidy		0.50	0.50	83
Rebate	$33\%\ 1:2\ 33\%$	\$0.50	\$0.33	71
Match		\$0.50	\$0.33	85
Discount		\$0.33	\$0.33	90
Rebate	$50\%\ 1:1\ 50\%$	\$0.50	0.25	58
Match		\$0.50	0.25	80
Discount		\$0.25	0.25	91

Table 3.1: Treatment Conditions

instructions stated that for every two packages (each package) that the subject provided at a nominal cost of \$0.50 apiece, one package would be matched at no additional cost to the subject. As a result, the charity would receive the combined number of packages. For the discount conditions, instructions stated that the subject would be able to provide packages for \$0.33 (\$0.25) instead of \$0.50 apiece. Hence, the nominal price corresponded to the effective price. For all subsidy types, instructions noted that the subsidy, i.e., the rebate, the matched units, or the money needed to reduce the nominal price, was provided by "a third party." This was a truthful yet indefinite reference to the research budget involved. Subjects chose the desired number of packages from a drop-down menu. The exact wording of each treatment can be found in Table 3.2.⁸

3.3.2 Experimental Protocol

We conduct the experiment online recruiting U.S. residents from the online labor market, Amazon's Mechanical Turk (AMT).⁹ In the case of money donations, online field experiments based on AMT (Gandullia and Lezzi, 2018; Gandullia,

⁸Figure C.1 in Appendix C.3 shows a screenshot of the complete donation appeal.

⁹AMT provides several benefits to researchers, among them fast and easy access to subjects, a diverse subject pool, and low costs (Paolacci et al., 2010; Mason and Suri, 2012). Several papers have examined the suitability of AMT for experimental research and have found encouraging results (Paolacci et al., 2010; Ross et al., 2199; Mason and Suri, 2012; Rand, 2012). Results in these papers highlight a high internal consistency of self-reported demographics, an incentive-compatibility of earnings, and a "spammer"-free workforce from the built-in reputation system. They also present and review results from successful replications of standard experimental games in AMT (e.g., Paolacci et al., 2010; Rand, 2012). In implementing our experiment, we followed the suggestions for researchers in that literature and the Guidelines for Academic Requesters on AMT (WeAreDynamo, 2014).

 Table 3.2:
 Final Part of Donation Appeal Wording by Treatment

Treatment	Wording
No subsidy	In this survey, you may use all, part, or none of your reward of \$2.00 for this HIT to provide these nutrition packages. Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted from your reward.
33% rebate	[Same text as in no subsidy condition]
	Upon completion of the survey, a third party has agreed to fund a 33% rebate for each package you provide. The rebate (\$0.17 per package provided) will be added to your reward.
1:2 match	[Same text as in no subsidy condition]
	A third party has agreed to match every two packages you provide , at no additional cost to you. So, for example, if you choose to provide 2 packages, Sign of Hope will receive 3.
33% discount	In this survey, you will be able to provide these nutritional packages for \$0.33 apiece (a third party will fund the remaining \$0.17). You may use all, part, or none of your reward of \$2.00 for this HIT to provide packages. Thus, you may choose any number between 0 and 6 packages. \$0.33 per package will be subtracted from your reward.
50% rebate	[Same text as in no subsidy condition]
	Upon completion of the survey, a third party has agreed to fund a 50% rebate for each package you provide. The rebate (\$0.25 per package provided) will be added to your reward.
1:1 match	[Same text as in no subsidy condition]
	A third party has agreed to match each package you provide , at no additional cost to you. So, for example, if you choose to provide 2 packages, Sign of Hope will receive 4.
50% discount	In this survey, you will be able to provide these nutritional packages for \$0.25 apiece (a third party will fund the remaining \$0.25). You may use all, part, or none of your reward of \$2.00 for this HIT to provide packages. Thus, you may choose any number between 0 and 8 packages. \$0.25 per package will be subtracted from your reward.

2019) and not based on AMT (Bekkers, 2015) have been successfully used to replicate the superiority of matches over rebates. Gandullia and Lezzi (2018) and Gandullia (2019) use the same endowment level and subsidy rates as we do. Our task was posted five times on the MTurk task queue between July and October 2015, resulting in five online sessions. Interested workers were informed that they would earn \$2 for answering a 20-minutes academic survey on several topics. The payment is rather high when compared to the average hourly wage of about \$3.1 to \$3.5 per worker on AMT (Hara et al., 2018). Each worker was only able to particpate once. Donations were mentioned as one of the topics, but the real-donation dictator game was not particularly salient compared to other survey elements. As a result, it is unlikely that subjects considered the donation task as the main subject of investigation. Interested workers followed a link which directed them to the survey containing the experiment on Qualtrics. Having followed the link to the survey platform, interested workers read and confirmed an informed consent page about the research study.

The experimental survey consisted of four parts: (1) the donation appeal, (2) a questionnaire on various topics, (3) a low-stake version of the Eckel-Grossman risk task (Eckel and Grossman, 2002, 2008b),¹⁰ and (4) a 5-item manipulation check questionnaire comparable to Eckel and Grossman (2003, 2006b) and Davis and Millner (2005). Parts (1) and (2) were presented in random order. Hence, a subject encountered the donation appeal either before or after the questionnaire. One of the treatment conditions was drawn at random and presented to the subject (between-subjects design).¹¹ The questionnaire of part (2) consisted of questions on sociodemographics, employment, and religious beliefs, as well as

¹⁰We opt for the Eckel-Grossman Risk Task because of its simplicity and quickness. A sample of AMT workers is likely to exhibit larger heterogeneity in numeracy than a standard laboratory sample of students. The Eckel-Grossman task has been shown to produce better results with people with low mathematical skills (Dave et al., 2010). Stakes start out at \$0.28 for the sure option and end up at \$0.02 and \$0.70 for the most risky gamble.

¹¹A different sample of 113 subjects received the treatment conditions in a within-subjects (WS) design to investigate how the results differ if individuals are forced to directly compare different subsidy types. Unlike in the between-subjects design, matches and discounts are more effective in providing the charitable good than rebates, including a significant crowding-in of net donations for those two subsidy types. Our analysis suggests that those different results for the WS sample are mostly an artifact of the WS design, which we consider less externally valid: When confronted with all possible subsidies, subjects seem to make a single decision at the extensive margin of giving across all subsidy types with a similar rate and then respond to the subsidy type mostly at the intensive margin. We provide more details about the WS design and the associated results in Appendix C.4.

current ambient environmental conditions and the Ten Item Personality Inventory (TIPI), which is a standard one-minute version of more extensive multi-item instruments to assess the Big Five personality dimensions (Gosling et al., 2003; Ehrhart et al., 2009). After completion of all survey parts, a unique code was shown that the subject had to enter into the survey task window on AMT to receive payment.

In total, we have 613 observations of participants starting the survey and 599 completed records. Incomplete records were dropped from the analysis.¹² The obvious concern that some subjects may fraudulently use multiple accounts to participate more than once is generally seen as a minor problem in online experiments (Horton et al., 2011; Paolacci et al., 2010).¹³ We nevertheless follow the common approach to exclude 40 subjects with duplicate Internet Protocol addresses from the analysis. Including them does not change the results. We also dropped one subject who indicated an age below 18 in the questionnaire, despite having confirmed an age above 18 when agreeing to the informed consent statement. This leaves us with a sample of 558 subjects (see Table 3.1 for the allocation of subjects across treatments). Average payouts were \$1.79 (net of donations and including an average of \$0.30 additional payment for the risk task). Subjects took on average 8.38 minutes to complete the experiment.

3.4 Results

Variables elicited in the questionnaire suggest a diverse sample of subjects (see Table C.2 in Appendix C.2): Slightly less than half of subjects are female, and slightly less than half graduated from college. About one-third of subjects are married, and about the same share has children under age 16 living in the house-hold. Both age and income are well spread, with the median age in category 26–34 and the median yearly income in category US\$40,000–49,999. Separate F-tests suggest that the characteristics are well-balanced across our treatment groups. Only one out of 35 comparisons is significant, and two additional ones

¹²Among the complete observations, three subjects had restarted the survey and hence created an incomplete duplicate record. We kept the complete observations of these three subjects after making sure that they had not encountered a treatment condition in their first attempt and gave the same answers in the survey.

¹³In the case of AMT, having multiple accounts is forbidden by Amazon's Terms of Service (Mason and Suri, 2012) and creating an account requires a unique credit card number (Paolacci et al., 2010).

reach marginal significance.

Answers to the manipulation check questions indicate that on average, subjects clearly understood instructions and procedures, felt that their anonymity was preserved, trusted the experimenters and the charity, and found the recipients of the donations worth supporting (Table C.2 in Appendix C.2). With the exception of the last item, the answers to the manipulation check questions do not significantly differ across treatments.

In Table 3.3, we present descriptive results for donations observed in the experimental treatments. Panel A reports mean values and standard deviations. Column 1 shows the average number of nutritional packages that subjects selected to donate in their version of the donation appeal. Hence, column 1 represents the units purchased before any rebate or matching subsidy while accounting for subsidized nominal prices in the discount conditions. If multiplied by the nominal price, column 1 would correspond to out-of-pocket expenses that are frequently denoted as "checkbook giving" in standard money donation experiments. Column 2 reports individual net donations in dollars that result from subjects' choices after all subsidies are accounted for. That is, column 2 is column 1 evaluated at the (discounted) nominal price minus any rebates. Column 3 refers to the mean number of nutritional packages the charity "receives," based on subjects' choices, that is, column 1 plus any matched packages. If we multiplied column 3 by \$0.50 for all treatments, we would obtain gross charity receipts in dollars, a common focus in money donation experiments. Because of perfect collinearity of both receipts measures, we only use charity receipts expressed in units in the following analysis, with the exception of Figure 3.1 where we use charity receipts expressed in dollars to illustrate its composition. Columns 4 and 5 show the intensive and the extensive margin of giving, respectively. Column 4 reports mean charity receipts conditional on being a donor while column 5 reports the fraction of donors. For each variable, panels B to D report *p*-values of pairwise comparison tests between treatments.

In the remainder of this section, we present the results from our analysis. We first focus on rebates and matches (Section 3.4.1) and subsequently turn to the effectiveness of discount subsidies (Section 3.4.2). We discuss potential explanations for our results in Section 3.5.

Treatments		Donation variable					
Condition	Nominal unit price (\$)	Effective unit price (\$)	Indivi- dual choice (units)	Net dona- tion (\$)	Charity receipt, uncond. (units)	Charity receipt, cond. (units)	Prob. of dona- tion
			(1)	(2)	(3)	(4)	(5)
A. Mean value	es (S.D.)						
No subsidy	0.50	0.50	$1.169 \\ (1.413)$	$\begin{array}{c} 0.584 \\ (0.706) \end{array}$	$1.169 \\ (1.413)$	$2.256 \\ (1.177)$	$\begin{array}{c} 0.518 \ (0.503) \end{array}$
33% rebate	0.50	0.33	$1.690 \\ (1.545)$	$\begin{array}{c} 0.558 \\ (0.510) \end{array}$	$1.690 \\ (1.545)$	$2.400 \\ (1.294)$	$0.704 \\ (0.460)$
1:2 match	0.50	0.33	$1.059 \\ (1.339)$	$\begin{array}{c} 0.529 \\ (0.670) \end{array}$	$1.506 \\ (2.021)$	$3.048 \\ (1.886)$	$0.494 \\ (0.503)$
33% discount	0.33	0.33	$1.478 \\ (1.973)$	$0.488 \\ (0.651)$	$1.478 \\ (1.973)$	$2.771 \\ 1.927$	$\begin{array}{c} 0.533 \\ (0.502) \end{array}$
50% rebate	0.50	0.25	1.931 (1.705)	0.483 (0.426)	1.931 (1.705)	2.732 (1.379)	0.707 (0.459)
1:1 match	0.50	0.25	1.113 (1.253)	0.556 (0.626)	2.225 (2.506)	3.787 (2.176)	0.588 (0.495)
50% discount	0.25	0.25	2.143 (2.831)	0.536 (0.708)	2.143 (2.831)	3.545 (2.879)	0.604 (0.492)
B. Tests of sub	bsidy types:	<i>p</i> -values					<u> </u>
B1. At effective 33% rebate vs. 33% rebate vs. 1:2 match vs.	ve price of \$. 1:2 match . 33% discou 33% discour	8 <i>0.33</i> unt nt	$\begin{array}{c} 0.01 \\ 0.44 \\ 0.10 \end{array}$	$0.76 \\ 0.44 \\ 0.68$	$0.52 \\ 0.44 \\ 0.93$	$0.06 \\ 0.27 \\ 0.49$	$\begin{array}{c} 0.01 \\ 0.03 \\ 0.60 \end{array}$
B2. At effective price of \$0.25 50% rebate vs. 1:1 match 50% rebate vs. 50% discount 1:1 match vs. 50% discount		$\begin{array}{c} 0.00 \\ 0.57 \\ 0.00 \end{array}$	$0.41 \\ 0.57 \\ 0.84$	$0.41 \\ 0.57 \\ 0.84$	$0.01 \\ 0.07 \\ 0.63$	$0.15 \\ 0.20 \\ 0.82$	
C. Tests of subsidized prices: p-values							
50% vs. 33% r 1:1 vs. 1:2 mat 50% vs. 33% d	ebate tch liscount		$0.41 \\ 0.79 \\ 0.07$	$\begin{array}{c} 0.36 \\ 0.79 \\ 0.64 \end{array}$	$0.41 \\ 0.05 \\ 0.07$	$0.24 \\ 0.09 \\ 0.11$	$0.97 \\ 0.23 \\ 0.34$
D. Tests of subsidized vs. unsubsidized prices: p-values							
D1. Low subsi 33% rebate vs. 1:2 match vs. 33% discount	<i>dy rate</i> . no subsidy no subsidy vs. no subsi	, dy	$0.03 \\ 0.61 \\ 0.24$	$0.79 \\ 0.61 \\ 0.35$	$0.03 \\ 0.21 \\ 0.24$	$0.58 \\ 0.02 \\ 0.12$	$0.02 \\ 0.76 \\ 0.84$
D2. High subs 50% rebate vs. 1:1 match vs. 50% discount v	<i>idy rate</i> no subsidy no subsidy vs. no subsi	dv	$0.01 \\ 0.79 \\ 0.00$	$0.29 \\ 0.79 \\ 0.65$	$0.01 \\ 0.00 \\ 0.00$	$0.09 \\ 0.00 \\ 0.00$	$0.03 \\ 0.37 \\ 0.25$

Table 3.3:Descriptive Results

Notes: Panel A shows mean values of the donation variables for each treatment (standard deviations in parentheses). Column 1 reports the number of packages that subjects selected to give at the nominal price. Column 2 shows the net dollar contribution implied by subjects' choices, i.e., column 1 evaluated at the nominal price minus the rebate (if any). Column 3 reports the overall number of packages received by the charity, i.e., column 1 plus matched units (if any). Column 4 reports the same measure as column 3 but conditional on giving (intensive margin). Column 5 reports the share of subjects who donated at least one package (extensive margin). Panels B to D show pairwise tests between treatment conditions. Panel B compares subsidy types conditional on the effective price. Panel C compares the two subsidized prices, 0.25 and 0.33, conditional on subsidy type. Panel D compares the unsubsidized price with the subsidized price arising from the low subsidy rate for each subsidy type. Columns 1 to 4 in panels B to D report *p*-values of two-tailed *t*-tests with unequal variances. Column 5 report *p*-values of Pearson χ^2 tests for binary data.

3.4.1 Rebates versus Matches

Three main results follow from columns 1 to 3 of Table 3.3. First, column 3 shows that the charity received an average of about 1.7 (1.9) units per subject in the 33% (50%) rebate condition and about 1.5 (2.2) units per subject in the 1:2 (1:1) matching condition. At both effective prices, the levels of charity receipts do not significantly differ between the two subsidy types (p = 0.52 and p = 0.41).

Result 1. (Charity receipts) Charity receipts do not significantly differ between rebate and matching subsidies.

Second, column 3 also shows that charity receipts significantly increase in the subsidy level, either from introducing a subsidy (p < 0.05 for a 33% rebate, 50% rebate, and 1:1 match) or from increasing the subsidy rate (p = 0.05 in case of the match).

Result 2. (Law of demand) Charity receipts significantly decrease in the price.

Third, column 2 indicates that net donations exhibit a roughly constant share of around one quarter of the endowment across all treatment conditions. Thus, neither the introduction of a subsidy at any rate nor an increase in the subsidy rate results in significant changes of subjects' own contributions net of the subsidy $(p \ge 0.29$ for pairwise comparisons). Note that in order to achieve the same level of charity receipts and own net donations, subjects need to select more units to donate under a rebate than under a match (since the match is paid on top of the units selected). This is exactly what we observe in column 1: The average number of units selected is at least 0.5 units larger ($p \le 0.01$ at both effective prices).

Result 3. (Net donations) There is no evidence for crowding-in or crowding-out of net donations by rebate or matching subsidies of any level.

Result 3 has an important implication: It implies that the increase in charity receipts (Result 2) is entirely driven by the additional money provided as subsidy payment by the third party, instead of being driven by individuals actually giving more (see Figure 3.1 for an illustration). This finding does not generally hold in the money donation literature. Several papers find that matches significantly increase net donations compared to a no subsidy condition without lead donor (Eckel et al., 2007; Karlan and List, 2007; Gneezy et al., 2014; Huck et al., 2015;



Figure 3.1: Charity Receipts (Divided into Net Donations and Subsidy Payments) by Treatment

Notes: The figure shows the average charity receipts in dollars (divided into average net donations and corresponding subsidy payments) by treatment. This corresponds to charity receipts in units (column 3 of Table 3.3) multiplied by the (unsubsidized) unit price of \$0.50.

Eckel and Grossman, 2017). However, there are also some papers that do not find a significant effect on net donations (Lukas et al., 2010; Karlan et al., 2011; Helms McCarty et al., 2018). Evidence for rebates is scarcer and rather points in the opposite direction, i.e., that rebates crowd out net donations (Eckel and Grossman, 2003, 2008a; Gandullia and Lezzi, 2018; Gandullia, 2019).

The most substantial difference between our findings and that of the money donation literature is the equality of rebates and matches regarding charity receipts (Result 1). The well-established finding in the context of money donations is that charity receipts under matches exceed those under rebates (Eckel and Grossman, 2003, 2006a,b, 2008a, 2017; Davis et al., 2005; Lukas et al., 2010; Bekkers, 2015; Gandullia and Lezzi, 2018; Gandullia, 2019), while "checkbook giving," which corresponds to column 1 of Table 3.3 multiplied by the nominal price, is often roughly the same under both subsidy types. We therefore now examine whether our results are robust to controlling for available covariates. For this purpose, we estimate an Ordered Probit Model with the individual choice as dependent variable and use it to analyze the effect of the different subsidies on the level of charity receipts. The details of the model are presented in Appendix $C.1.^{14}$

In Panel A of Table 3.4, columns 1 and 2, we present the results in the form of the average marginal effects on charity receipts without and with controlling for covariates. For example, offering a 33% rebate is estimated to increase average charity receipts per individual by about 0.5 packages compared to not offering any subsidy (column 1, Rebate), whereas increasing the subsidy rate from 33% to 50% has no significant effect in the case of the rebate (column 1, Rebate \times low price). Analogously to Table 3.3, the predicted levels of charity receipts are compared pairwise across subsidy types in Panel B, holding the effective price constant. The estimates confirm Result 1 and Result 2. We repeat the same exercise for predicted levels of net donations. In line with Result 3, we neither find significant differences in net donations between subsidy types at the same effective price nor any evidence for crowding-in or crowding-out at any conventional significance level when changing the price of giving due to a specific subsidy type.¹⁵

Having observed that charity receipts do not differ between rebates and matches, we ask whether this result masks heterogeneities in the "conversion rates" of the experimental donation call (the extensive margin of giving) and the conditional level of charity receipts demanded by donors (the intensive margin of giving). As discussed in Section 3.1, rebates decrease the minimum net expense required to become a donor, making them potentially more effective at the extensive margin than matches. Indeed, results in column 5 of Table 3.3 confirm that rebates attract a larger share of donors than matches. In particular, the differences amount to roughly 21 percentage points (70.4% vs. 49.4%) and 12 percentage points (70.7% vs. 58.8%) in the case of the high and the low effective price (p = 0.01) but not at the low effective price (p = 0.15).¹⁶ We take this as evi-

¹⁶The fact that the difference in the extensive margin is more pronounced for the low subsidy rate is not surprising, since for the 1:2 match the first unit donated does not result in a matching

¹⁴Whereas a common approach in the literature is to estimate a Tobit Model with the monetary value that the individual has chosen to give, charity receipts, or their logarithmized value as dependent variable, the discrete nature of our donation decision makes it an unsuitable choice to model our data. This is supported by conditional moment tests significantly rejecting the assumption of normally distributed error terms for the Tobit Model with charity receipts or logarithmized charity receipts as dependent variable (p < 0.01). We nevertheless run different Tobit specifications and OLS regressions as robustness checks and find similar results (available from the authors upon request).

¹⁵Results are available from the authors upon request.

	Charity Receipts, unconditional (units)		Probability of donation (binary)		Charity Receipts, conditional (units)		
	(1)	(2)	(3)	(4)	(5)	(6)	
A. Marginal effects							
Rebate	0.546^{**} (0.237)	0.533^{**} (0.257)	0.186^{**} (0.077)	0.220^{***} (0.081)	$\begin{array}{c} 0.091 \\ (0.256) \end{array}$	-0.039 (0.27)	
Match	$\begin{array}{c} 0.324 \\ (0.261) \end{array}$	$0.248 \\ (0.291)$	-0.024 (0.077)	-0.011 (0.086)	0.799^{**} (0.359)	0.488 (0.392)	
Discount	$\begin{array}{c} 0.306 \\ (0.247) \end{array}$	$\begin{array}{c} 0.231 \\ (0.272) \end{array}$	$\begin{array}{c} 0.015 \\ (0.076) \end{array}$	$\begin{array}{c} 0.012 \\ (0.084) \end{array}$	$\begin{array}{c} 0.537 \ (0.332) \end{array}$	$\begin{array}{c} 0.316 \ (0.357) \end{array}$	
Rebate \times low price	$0.218 \\ (0.277)$	$\begin{array}{c} 0.218 \ (0.302) \end{array}$	$\begin{array}{c} 0.003 \\ (0.081) \end{array}$	-0.018 (0.088)	$0.326 \\ (0.266)$	$\begin{array}{c} 0.342 \\ (0.279) \end{array}$	
Match \times low price	0.888^{**} (0.373)	0.992^{**} (0.405)	$0.093 \\ (0.077)$	$0.104 \\ (0.084)$	0.783^{*} (0.448)	0.782^{*} (0.466)	
Discount \times low price	0.585^{*} (0.326)	0.613^{*} (0.366)	$\begin{array}{c} 0.071 \\ (0.073) \end{array}$	$0.057 \\ (0.084)$	$\begin{array}{c} 0.571 \\ (0.448) \end{array}$	$\begin{array}{c} 0.807 \\ (0.503) \end{array}$	
B. Tests of subsidy types: p-va	B. Tests of subsidy types: p-values						
B1. At effective price of \$0.33							
33% rebate vs. 1:2 match	0.43	0.35	0.01	0.00	0.05	0.17	
33% rebate vs. 33% discount	0.37	0.30	0.02	0.01	0.17	0.31	
1:2 match vs. 33% discount	0.95	0.96	0.60	0.79	0.53	0.69	
B2. At effective price of \$0.25							
50% rebate vs. 1:1 match	0.23	0.22	0.14	0.21	0.00	0.01	
50% rebate vs. $50%$ discount	0.71	0.81	0.19	0.13	0.09	0.08	
1:1 match vs. 50% discount	0.43	0.37	0.82	0.77	0.33	0.78	
$Covariates^a$	No	Yes	No	Yes	No	Yes	
Log likelihood	-801.28	-592.51	-372.07	-250.92	-427.37	-314.94	
Observations	558	428	558	428	326	256	

 Table 3.4:
 Estimation Results

Notes: (1)–(2): Ordered Probit with the number of packages selected by the individual as dependent variable. (3)–(4): Probit for whether or not a donation was made. (5)–(6): Ordered Probit conditional on being a donor, with the number of packages selected by the individual as dependent variable. (1)–(2) and (5)–(6) treat a single observation with 5 selected packages as if it were 6 selected packages.

Panel A presents average marginal effects. Standard errors reported in parentheses, *p < 0.1, **p < 0.05, ***p < 0.01. For (1)–(2) and (5)–(6), marginal effects refer to the average change in expected charity receipts over all individuals or donors only, respectively. For each individual considered, the change is calculated by taking the difference in expected charity receipts between receiving a particular subsidy at the low rate (rebate, match, discount) and not receiving any subsidy or between receiving a particular subsidy at the high rate (rebate × low price, match × low price, discount × low price) and receiving the same subsidy at the low rate.

Panel B presents *p*-values for the pairwise comparison of treatment effects (subsidy treatment vs. no subsidy) between subsidy types, based on the average marginal effects.

^aCovariates include gender, marital status, the Big Five personality dimensions, risk preferences, categorical variables for age, income, residential environment, and religion, and dummies for whether the individual holds a college degree, whether children under the age of 16 live in the household, whether the individual is a registered voter, whether the individual frequently attends religious services, whether the individual works for a not-for-profit organization and task order. Likelihood ratio tests reject that their coefficients in model (2), (4) and (6) are jointly zero (p < 0.01, p < 0.01 and p < 0.05, respectively). dence that the equality result for the level of charity receipts is partly driven by the fact that rebate subsidies are more effective at the extensive margin.

Result 4. (Extensive margin) Rebates are more effective in attracting donors than matches.

Turning to the intensive margin, column 4 of Table 3.3, shows that conditional charity receipts under both match conditions significantly exceed the corresponding values in the rebate conditions (3.0 vs. 2.4 units and 3.8 vs. 2.7 units, respectively; p = 0.06 and p = 0.01).

Result 5. (Intensive margin) Charity receipts per donor are higher under matching than under rebate subsidies.

Comparing matching and rebate treatments in column 4 and 5 of Table 3.3 to the control reinforces the view that the channel through which rebates raise unconditional charity receipts primarily is the extensive margin whereas matches unfold their impact through the intensive margin. For the rebate, introducing the low subsidy rate increases the share of donors in column 5 from 51.8% to 70.4% (p = 0.02) compared to the no subsidy condition, while the intensive margin is not significantly affected (p = 0.58). In contrast, for the match, introducing the low subsidy rate increases mean conditional charity receipts in column 4 from 2.3 to 3.0 units (p = 0.02) compared to the no subsidy condition, while the extensive margin is unaffected (p = 0.76).

Again, we supplement the descriptive results by estimating appropriate parametric models. Columns 3 and 4 of Table 3.4 refer to a Probit without and with covariates, respectively, while columns 5 and 6 capture the intensive margin by estimating an Ordered Probit Model for donors only. The latter is set up analogously to the Ordered Probit Model used above. If we assumed that after controlling for observable characteristics, the error terms between the decisions to donate and how much to donate are uncorrelated, we could interpret these two models jointly as a Two-Part model. The parametric estimation confirms results 4 and 5, but for the intensive margin, only the difference between matches and rebates at the high subsidy rate remains significant when covariates are included.

payment. Consequently, the minimum expense required to become a donor is larger than for the equivalent rebate while the impact of the action is the same: a single nutritional package received by the charity. As a result, not only the costs but also the effective prices at the margin of becoming a donor differ, further decreasing the relative attractiveness of the match.

One concern regarding the comparison of rebate and matching subsidies in our experiment might be that differences in the budget constraints for charity receipts could drive some of the results. Under a rebate, the highest possible number of packages received by the charity is always four, since the donor must fully fund each selected unit at a nominal price of \$0.5 before receiving the refund. In contrast, the matching subsidy applies on top of the selected packages: If under a 1:1 match a donor decides to spend her whole endowment of \$2 to fund four packages, then the charity receives eight packages.¹⁷ Similar differences apply to almost all laboratory experiments comparing rebates and matches in the money donation literature, as they also endow subjects with a limited amount of money. In the money donation literature, it is shown that the higher effectiveness of matches observed in laboratory studies also holds in field experiments where subjects use their own income (Eckel and Grossman, 2008a, 2017). If the budget constraint mattered in our design, the results could understate the effectiveness of rebates compared to matches for situations in which the budget constraint is more loose or non-binding. This implies that rebates might be even more effective than matches in such situations.

To provide a robustness check on this matter, we revisit Result 1 and Result 5 by recoding subjects' choices in order to equalize budget constraints. In our data, a total of 29.5% of subjects give the maximum amount under rebates, compared to 10.9% in the matching conditions. For each condition, we set all charity receipts above four packages to four packages (the maximum level of charity receipts under rebates). Detailed results are presented in Table C.4 in Appendix C.2. Although rebates now provide the highest average number of packages received by the charity, the difference to matches is not significant at the high subsidy rate (p = 0.65) and only marginally significant at the low subsidy rate (p = 0.09). Hence, Result 1 survives the robustness check. In contrast, the difference on the intensive margin (Result 5) vanishes after censoring charity receipt at four packages. A possible explanation is that matches create larger conditional donations only in settings where the budget constraint is binding for a sufficiently large share of individuals.

 $^{^{17}}$ See Table C.3 in Appendix C.2 for the detailed choice set by treatment.

3.4.2 Discount Subsidies

Subsidies that put a simple price reduction on nominal prices turn out to be as effective as rebate and matching subsidies that produce equivalent effective prices. Charity receipts in column 3 and net donations in column 2 of Table 3.3 do not significantly differ from those under the other two subsidy types ($p \ge 0.44$ for each pairwise comparison). Hence, the increased salience of the effective price under discounts does not seem to affect demand, and this alternative subsidy type does not lend itself to a more effective subsidy. Instead, the selected number of units in column 1 under a discount is statistically indistinguishable from that selected under a rebate (p = 0.44 and p = 0.57) but by an amount higher than under the corresponding match (p = 0.10 and p < 0.005), which approximately makes up for the additional units provided as matching payment. In line with the law of demand, charity receipts increase in the subsidy level by applying the 50% discount instead of the no subsidy condition (p < 0.01) or the %33 discount (p = 0.07). There is again no evidence for crowding-in or -out. Our Ordered Probit estimates in Table 3.4, columns 1–2, confirm these findings.

Result 6. (Discounts) The discount subsidy produces the same level of charity receipts and net donations as rebates and matches. Increasing the subsidy rate increases charity receipts, without crowding-in or crowding-out net donations.

As in the previous section, we can differentiate the behavior into the extensive and the intensive margin. At the low subsidy rate, the likelihood to give under the discount is significantly lower than under the rebate ($p \leq 0.03$, column 5 of Table 3.3 and columns 3–4 of Table 3.4). Since we would expect the responses to rebates and discounts to be similar at the extensive margin, this difference may hint towards a behavioral bias in the response to an equivalent decrease in the cost of becoming a donor. At the intensive margin, there is only some marginally significant difference between discounts and rebates at the high subsidy rate, shown in column 4 of Table 3.3 and columns 5–6 of Table 3.4. In comparison to the matching subsidies, there are neither significant differences at the extensive nor at the intensive margin.

3.5 Discussion

In our experiment, we find equivalence between matches and rebates as subsidybased incentives to donors and an equivalence with price discounts. This equivalence under a unit donation scheme contrasts with the existing literature that has examined subsidy types under a money donation scheme and has generally found matches outperforming rebates. This includes papers that also use an online experimental methodology (Bekkers, 2015; Gandullia and Lezzi, 2018; Gandullia, 2019) of which two recruit from a similar subject pool and use the same endowment level as we do (Gandullia and Lezzi, 2018; Gandullia, 2019). A closer parallel exists with experimental evidence comparing product promotions in the marketing literature. For private goods, matches (bonus packs) and price promotions sometimes perform equally well (Sinha and Smith, 2000; Hardesty and Bearden, 2003; Chen et al., 2012). For charitable goods, however, our finding is unusual.

To guide our intuition about this result, note that as discussed in Section 3.2.2, rebates and matches are no longer theoretically equivalent under a unit donation scheme. When donors face the choice architecture of a unit donation scheme, the minimum net expense required to become a donor is lower under rebates than under matches. This does not hold for a money donation scheme. As a result, the behavior on the extensive margin might be a crucial factor to explain why matches do not outperform rebates in our setting. In line with this reasoning, our results on the extensive margin differ from those obtained in the context of money donations schemes. While we find that rebates attract more donors than matches, Bekkers (2015) finds the opposite by using similar subsidy rates in a standard money donation choice architecture. Furthermore, Gandullia and Lezzi (2018) use the same online population, subsidy rates, and endowment levels as we do but focus on a standard money donation scheme. In their experiment, both rebates and matches increase the fraction of donors compared to a no-subsidy condition and effect sizes between the different subsidy types are similar.

Additional evidence that the lower cost of becoming a donor under rebates might drive the results comes from a simple recoding exercise. In our experiment, the minimum positive net donation under a match amounts to 0.50. Under a 33% (50%) rebate, 25% (22%) of donors give less than 0.50. Recoding those subjects as non-donors eliminates any significant difference at the extensive margin, and matches now lead to higher charity receipts than rebates (1.506 vs. 1.437 units at the low subsidy rate and 2.225 vs. 1.707 units at the high subsidy rate), yet differences between the two subsidy types remain statistically insignificant (p = 0.817 and p = 0.167). Hence, equalizing the cost of becoming a donor ex post moves the subsidy comparison towards the standard result from money donations, namely that matches outperform rebates.

The importance of the extensive margin to explain our results is also in line with findings by Diederich et al. (2020). The authors show that using unit instead of money donation schemes affects the propensity to give, with the primary driver being the discrete choice set under unit donation schemes. This characteristic is also responsible for the different cost of becoming a donor between rebates and matches in our experiment. While the discreteness thus appears to be an important factor to explain our results, we cannot exclude the possibility that the other distinct characteristics of unit donation schemes discussed in Section 3.2.1 - i.e., the additional information on the effectiveness of a donation and the framing in terms of physical units – may also play a role.¹⁸ Additional research is required to quantify the relative effects of the different characteristics on the responsiveness to subsidies. One possible approach would be to introduce one characteristic at a time, similar to Diederich et al. (2020), and investigate whether and how the effectiveness of certain subsidy types changes.

Regarding the comparison of the rebate and discount subsidies, the equivalence on the aggregate level accords with our expectations. However, our data offers some evidence that rebates are more effective in attracting donors. Although we should await future research to confirm the robustness of this finding, such a difference is surprising given that both subsidies imply the same price of becoming a donor. One speculative explanation is that a donor has the feeling of providing the whole unit of the charitable good herself under the rebate but only a fraction of the unit under the discount. In this case, the donor might derive lower warm glow utility from becoming a donor under the discount than under the rebate.

An interesting question for future research is how the effectiveness of the

¹⁸As pointed out by an anonymous referee, a potential explanation for a different response to rebates under a unit instead of a money donation scheme is fungibility: Due to the unit framing, the donation is in terms of physical units whereas the rebate is in terms of money, which might abate the feeling that money is taken away from the recipient.

different subsidy types under a unit donation scheme depends on the level of the unsubsidized unit price. At a given subsidy rate, a larger unsubsidized price increases the absolute difference in the minimum expense required to become a donor between rebates and matches. As a result, the differences on the extensive margin might become more pronounced, which in turn might be sufficient to make the rebate raise more money than the match. In contrast, reducing the unit size might move results closer to what has been found for money donation schemes.

Future research could also help to verify the generalizability of our results, for example, with respect to the absolute size of the earned endowment and the relative sizes of unit price and endowment. At \$2, the earned endowment is small in absolute terms, even though it is large in the context of the experimental population we recruit. Gandullia and Lezzi (2018) and Gandullia (2019) use the same endowment level with a similar subject pool and replicate the standard finding of matches outperforming rebates under a money donation scheme. These results hint at generalizability, but more research is needed for the specific case examined in the present paper. At \$0.50, the unsubsidized unit price of the charitable good is a quarter of the earned endowment and therefore restricts the room for variation in the donation decision, potentially limiting the scope for identifying differences. Although we still find significant differences between the subsidy and no subsidy conditions as well as between the subsidy types on the extensive and intensive margin, the equivalence of rebates and matches regarding charity receipts merits further examination.

3.6 Conclusion

In this paper, we define a class of donations in which donors are asked to choose the number of discrete units of the charitable good to fund instead of the amount of money to give. We call the former a unit donation and the latter a money donation. We present empirical evidence from an online field experiment designed to analyze how different subsidy types affect unit donations. In doing so, we focus on the two prevalent subsidy types, rebates and matches, as well as a subsidy type that is novel to charitable giving and framed as a simple price discount. The latter can be applied since for unit donations, each physical unit has a well-defined price that can be explicitly reduced. The results remarkably differ from the well-established findings for money donations. Matching subsidies do not outperform rebates but are equally effective in raising funds. Yet matching and rebate subsidies create different responses at the extensive and intensive margin of giving. While rebates significantly increase the fraction of donors, matches produce larger donations. The significantly higher likelihood to give under rebates compared to matches is in contrast to the money donation literature and appears to be one reason why rebates catch up with matches in the unit donation setting of our experiment. Price discounts raise similar levels of funds as rebates and matches. None of the subsidy types significantly affects net donations.

Our results underline the relevance of the decision environment when soliciting donations and, thus, have important implications for practitioners. First, charities that employ unit donations in their fundraising efforts cannot rely on the insights from the existing literature on subsidizing money donations. Second, whether it is useful to apply a certain type of subsidy to unit donations depends on a charity's objectives. Previous research has shown that individuals that donated once are more likely to give in the future. If the charity desires to maximize the set of donors, our evidence suggests that a rebate is preferable over a match. If the charity instead seeks to maximize charity receipts, the choice of the subsidy type seems to be irrelevant, offering some additional degrees of freedom to charities in their campaign design. Third, in cases where funds are not tied to being used as a subsidy, subsidizing unit donations is not necessarily beneficial as on the aggregate it may not crowd in private giving.

Chapter 4

Do Beliefs About Lobbying Affect Pro-environmental Behavior? Experimental Evidence^{*}

4.1 Introduction

Lobbying is an established part of the political process and builds on different strategies that are applied by interest groups to influence political decisionmaking. Previous research provides evidence that lobbying can be effective in exerting influence (e.g., Gawande et al., 2006; Markussen and Svendsen, 2005; Igan and Mishra, 2014; Giger and Klüver, 2016) and in securing substantial returns for the lobbying party (de Figueiredo and Silverman, 2006; Kang, 2016). Such achievements are paid for by the billions of dollars spent on lobbying activities (OECD, 2012). Given this evidence, it is not surprising that people's perception of lobbying is rather negative. An often held belief is that it negatively affects policy-making and primarily benefits businesses and politicians instead of citizens (Epperson et al., 2019).

^{*}I am grateful to Wolfgang Habla, Wladislaw Mill, Alice Solda, Matthias Stelter, Ulrich Wagner, Israel Waichman, and audiences at the 25th EAERE Annual Conference, the VfS Annual Conference, Heidelberg University, the University of Mannheim, and the ZEW – Leibniz Centre for European Economic Research for helpful comments and suggestions. Funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) – Project-ID 139943784 – SFB 884.

The goal of this paper is to investigate whether such a negative perception of lobbying has direct consequences for individual behavior. I focus on the context of environmental protection, where lobbying is often associated with the prevention of stricter regulations (Grey, 2018; Meng and Rode, 2019). On the one hand, a negative perception might discourage individuals from engaging in pro-environmental behavior. Experimental studies have found that a large share of individuals conditionally cooperate in public good games (e.g., Keser and Van Winden, 2000; Fischbacher et al., 2001; Croson and Shang, 2008; Fischbacher and Gächter, 2010), i.e., they contribute less if others contribute less. As environmental protection is a public good, this suggests that the more lobbying is expected to decrease the environmental contributions of other actors, the lower an individual's willingness to behave pro-environmentally. On the other hand, expecting lobbying to prevent environmental regulations could also increase individuals' contributions, e.g., if an individual tries to make up for what others fail to provide. This behavior could be rationalized with a marginal utility from contributing that is decreasing in the size of the public good.

Since it is unlikely that individuals are perfectly informed about the impact of lobbying, it is their *belief* about the impact and not the impact itself that is of primary importance for the behavioral response to lobbying. In general, beliefs are endogenous and are likely to be correlated with a large set of other factors that also affect individual behavior. I use an experimental approach to randomly shift beliefs about lobbying and identify their causal impact on pro-environmental behavior. The experiment is implemented in a survey with a heterogeneous sample from Germany. It mainly builds on the following two treatment groups. Participants in the first group (Group Positive) read three statements that suggest lobbying promotes climate protection, whereas participants in the second group (Group Negative) read three statements that suggest lobbying hinders climate protection. Afterward, participants answer questions about the impact of lobbying on climate protection and have the opportunity to contribute to the reduction of carbon emissions via a donation to the climate protection organization atmosfair. The donation decision serves as a measure of individuals' pro-environmental behavior. A follow-up survey about seven months after the experiment offers additional information on the persistence of beliefs and stated pro-environmental behaviors.

While the experiment successfully induces a gap in the beliefs of the two

groups, I find mixed evidence on whether this difference in beliefs affects proenvironmental behavior. Only the measure based on the stated behaviors shows a significant impact of being confronted with negative instead of positive statements, suggesting that negative beliefs reduce pro-environmental behavior. The estimated treatment effect on the observed contribution to the reduction of carbon emissions points in the same direction but is not significant at any conventional level.

This paper contributes to a large literature on the determinants of proenvironmental behavior and, more generally, the voluntary contribution to public goods. Most closely related is a strand of this literature that investigates the effect of others' contributions on own behavior. For example, Schultz et al. (2007) and Goldstein et al. (2008) show how providing information on the descriptive norm (i.e., what others do) induces individuals to adjust their conservation behavior (reusing towels in hotel rooms) towards the presented norm. Similarly, descriptive norms have been successfully used in fostering energy or water conservation (e.g., Allcott, 2011; Ferraro et al., 2011; Ferraro and Price, 2013; Allcott and Rogers, 2014).¹

In the context of laboratory experiments, a common finding is that a substantial share of individuals conditionally cooperate in public good games (e.g., Keser and Van Winden, 2000; Fischbacher et al., 2001; Croson and Shang, 2008): Individuals give more if others give more. Some of these experiments analyze the role of beliefs in explaining observed behaviors (Fischbacher and Gächter, 2010; Smith, 2013). In this paper, the considered belief does not directly concern the contribution decision of other individuals but the impact of lobbying on climate protection. Nevertheless, the underlying mechanism might be similar: Expecting lobbying to decrease the level of climate protection (e.g., from firms, governments, or individuals) might discourage contributions on the individual level.

Fairness concerns could play a crucial role in such a behavioral response: An unfair distribution of costs has been found to be associated with a lower acceptance of environmental policies (Maestre-Andrés et al., 2019) and a reduction in the willingness to contribute (Andor et al., 2018). In particular, Andor et al. (2018) show that households' (hypothetical) willingness-to-pay for increasing the share of renewable energy is higher if energy-intensive companies are not exempt

¹See Farrow et al. (2017) for a literature review on the impact of norms on pro-environmental behavior.

from bearing the costs.

My paper also extends the literature on the consequences of lobbying by investigating a potential spillover effect that has received little attention so far. Several studies investigate whether and how lobbying influences political decisionmaking (e.g., Igan and Mishra, 2014; Giger and Klüver, 2016; Bertrand et al., 2020; for a review, see de Figueiredo and Richter, 2014), including the context of environmental policy (Markussen and Svendsen, 2005; Gullberg, 2008; Kang, 2016; Meng and Rode, 2019). For example, Kang (2016) estimates that lobbying by the energy sector during the 110th Congress had a small but significant impact on the enactment of policies. However, little is known about whether – in addition to a direct impact of lobbying – the public perception of lobbying might cause behavioral adjustments on the individual level. This spillover can be a relevant factor in assessing the welfare effects of lobbying, particularly in the context of environmental protection, where massive action of all actors is needed to tackle global challenges.

The paper proceeds as follows. Section 4.2 describes the experiment and provides detailed information about the sample. Section 4.3 presents the results by first focusing on the effectiveness of the belief manipulation and subsequently turning to the impact on pro-environmental behavior. Section 4.4 concludes.

4.2 Experiment

4.2.1 The German Internet Panel

The experiment was implemented in the German Internet Panel (GIP), a probability-based online panel of the general population in Germany aged 16–75.² The panel includes both the online and the offline population by recruiting participants face-to-face and providing equipment to households that do not have access to the internet or a computer. Surveys are conducted on a bi-monthly basis and take 20 to 25 minutes. Participants usually receive a reward of 4 euro for each completed survey and an annual bonus of up to 10 euro if all surveys within a one-year period have been answered. A detailed description of the GIP, including the recruitment process, can be found in Blom et al. (2015).

²The German Internet Panel is funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) – Project-ID 139943784 – Collaborative Research Center 884 "Political Economy of Reforms" (SFB 884).

In this paper, I use data from the German Internet Panel waves 38, 41, 43, 44, and 48 (Blom et al., 2019, 2020a, 2020b, 2021a, 2021b). The main experiment was implemented in wave 44, which took place in November 2019. The data sets of wave 38 (November 2018), 41 (May 2019), and 43 (September 2019) are used to obtain baseline beliefs and sociodemographic characteristics, while wave 48 (July 2020) contains a follow-up survey, which is described in more detail in Section 4.2.3.

4.2.2 Experimental Design

The experiment in wave 44 consisted of three main parts. The goal of the first part was to shift individuals' beliefs about the impact of lobbying. After reading a short general introduction about the topic of lobbying, participants were randomly assigned to one of three experimental groups. The first group (Control) directly continued with the second part of the experiment described below. The second group (Group Positive) was confronted with three statements arguing that lobbying (i) improves environmental policy by providing crucial information, (ii) allows nonprofit organizations to achieve a stricter climate policy, and (iii) is sufficiently transparent. The third group (Group Negative) was confronted with three statements arguing that lobbying (i) prevents environmental regulations, (ii) leads to a climate policy in the interest of companies, and (iii) lacks transparency. The wording of these statements is presented in Appendix Table D.1. Participants were asked to read all presented statements and mark those they agree with. Afterward, they indicated how many of the statements they had heard of before. The instructions did not discuss whether the statements are true.

In the second part of the experiment, participants answered up to five questions about the impact of lobbying.³ The last of these questions elicited the belief about the impact of lobbying on the level of climate protection and hence is the focus of this paper.⁴ The translated question reads as follows:

"How does lobbying affect the level of climate protection in the European Union? Lobbying leads to"

 $^{^3\}mathrm{The}$ number of questions varied between 4 and 5 due to the conditioning on previous answers.

⁴The other questions focused on the actual or potential impact of lobbying on the European Union's climate policy.

The answer categories were (2) "much more climate protection," (1) "rather more," (0) "neither more nor less," (-1) "rather less," and (-2) "much less climate protection." Alternatively, participants could select the answer "I don't know." The same question had been included in a survey one year before the experiment (wave 38, November 2018),⁵ which allows me to control for baseline beliefs when investigating whether the treatment had the intended effect.

The final part of the experiment was separated from the belief elicitation by six unrelated questions and was restricted to participants from Group Positive and Group Negative. It informed participants that they receive an additional payment of 4 euro for answering the survey, with the opportunity to donate all or part of this extra money to the climate protection organization *atmosfair*. atmosfair allows individuals to reduce carbon emissions by donating to climate projects, which typically focus on renewable energies in third world countries.⁶ Participants received information about the organization and entered the amount they would like to give on a separate screen.⁷ For simplicity, only integer amounts between 0 euro and 4 euro were possible. Any money not donated to atmosfair was transferred onto the participants' study accounts and paid out together with the cumulative earnings since November 2019 in April 2020.⁸ In line with previous research (Löschel et al., 2013; Diederich and Goeschl, 2014, 2017, 2018; Goeschl et al., 2020), the donation to reduce carbon emissions is used as a measure of pro-environmental, or more specifically, pro-climate behavior.

Participants of the GIP can generally choose from three different modes of payment for the cumulative earnings on their study account. The earnings are either (i) transferred to their bank account, (ii) provided in the form of an Amazon voucher, or (iii) donated in equal portions to three charities, i.e., the German Red Cross, SOS-Kinderdorf e.V., and the World Wildlife Fund For Nature (WWF). Participants have to make this decision when being recruited into the GIP but can change the mode of payment at any time.⁹ As donating to WWF might be

 $^{^{5}}$ The wording of the question in wave 38 was the same except that the phrase "Lobbying leads to ..." was not included. This phrase was added in wave 44 to avoid any possible confusion about the interpretation of the answer categories.

⁶See https://www.atmosfair.de/en/.

⁷See Appendix Table D.2 for the detailed instructions.

 $^{^{8}}$ To eliminate the impact of time preference on the donation decision, donations to *atmosfair* were paid out in April 2020 as well.

⁹In October 2019 (the last payout of cumulative earnings before the experiment), 15 percent of GIP participants with a positive balance on their study account donated their earnings, 32 percent got an Amazon voucher, and 53 percent received a bank transfer.

a substitute for a contribution to climate protection via atmosfair,¹⁰ I use the sum of money donated to atmosfair and WWF in April 2020 as an alternative measure of pro-environmental behavior. The data on the payouts of the cumulative earnings is only available via the On-Site Data Access (ODA) facilities of the GIP for data protection reasons.

4.2.3 Follow-Up

A follow-up survey was implemented in the GIP about seven months after the main experiment (wave 48, July 2020). The survey did not differ across treatments and served two main purposes. First, by including the belief elicitation question from wave 44, it allows me to test whether a potential difference in beliefs induced by the experiment persists over time. Second, it provides additional measures of pro-environmental behavior in a stated format. This is relevant as a donation to atmosfair does not capture all forms of pro-environmental behavior. The follow-up survey elicited how often individuals use a car in a typical week and whether they had conducted the following actions at least once in the past six months: (i) taking a plane, (ii) purchasing local, organic products, (iii) bringing their own bag for shopping, (iv) considering a product's sustainability in the purchasing decision, (v) donating to an environmental organization, (vi) volunteering for an environmental cause, (vii) protesting for more environmental or climate protection, and (viii) signing a petition for more environmental or climate protection. These questions were inspired by previous studies that rely on pro-environmental behaviors (e.g., Videras et al., 2012; Meyer, 2015; Binder et al., 2020).

Kormos and Gifford (2014) and Lange and Dewitte (2019) discuss the advantages and disadvantages of stated measures in the context of pro-environmental behavior. Major advantages are the low costs of data collection and the great flexibility, which makes it relatively easy to consider a wide range of behaviors. Major disadvantages are the risk of over-reporting and limited memory.¹¹ In

 $^{^{10}{\}rm Grieder}$ et al. (2020), for example, use a donation to WWF as a proxy for pro-environmental behavior.

¹¹Two other disadvantages frequently discussed are social desirability bias and the ambiguity of answer categories or questions. Empirical research has found only very limited support for the former (see, e.g., Milfont, 2009; Chao and Lam, 2011). To avoid the latter, the answer categories of the questions used in this paper have a simple yes-no format or refer to a precise unit of measurement (the number of days per week). Furthermore, each question asks about a



Figure 4.1: Treatment Assignment in Wave 44 (November 2019)

Notes: The figure illustrates the assignment of participants in wave 44. The numbers include incomplete survey responses which are subsequently removed. The final sample consists of 3,071 individuals (1,014 in Control, 1,031 in Group Positive, and 1,026 in Group Negative). Throughout the paper, I mainly focus on participants in Group Positive and Group Negative, who completed all three parts of the experiment described in Section 4.2.2.

their meta-analysis, Kormos and Gifford (2014) find a strong correlation between stated and observed pro-environmental behavior. However, the authors also emphasize that a large fraction of variance remains unexplained when using stated behavior as a predictor of observed behavior. Although this might be less of a concern when the main focus is on the difference between two groups rather than the level of pro-environmental behavior, the weaknesses of stated measures should be kept in mind when interpreting the results.

4.2.4 Sample

The sample for this survey experiment is a subset of the individuals who participated in wave 44 (November 2019) of the GIP. The assignment procedure was as follows (see Figure 4.1 for an illustration). Participants were eligible for the experiment if they indicated a well-defined belief about the impact of lobbying in wave 38 (November 2018). Hence, participants who did not complete the survey of wave 38 (909 individuals), did not answer the belief elicitation question in wave 38 (3 individuals), or selected the answer "I don't know" (551 individuals) were excluded from the survey experiment. Three thousand ninety-one from the

specific behavior.

3,117 eligible participants reached the first part of the survey experiment and were, with equal probabilities, assigned either to Control, Group Positive, or Group Negative. The randomization was stratified by the answer to the belief elicitation question in wave 38.¹² Twenty participants (3 out of 1,017 in Control, 5 out of 1,036 in Group Positive, and 12 out of 1,038 in Group Negative) are subsequently removed since they did not complete the survey or did not receive the belief elicitation question due to a technical problem. Hence, the final sample of the survey experiment consists of 3,071 individuals from the general population of Germany. About 91 percent of these individuals also provide complete information on their pro-environmental behaviors and beliefs in the follow-up survey. The attrition does not significantly differ across the experimental groups (χ^2 -test, p = 0.67).

Throughout the paper, I mainly focus on observations from Group Positive and Group Negative since participants who were assigned to the control group did not have the opportunity to donate money to atmosfair. I will only use these observations to learn more about how beliefs adjusted in response to the positive and negative statements compared to not reading any statement.

Table 4.1 presents summary statistics by experimental group and *p*-values from testing the balance of each covariate (see Appendix Table D.3 for a detailed description of each variable used throughout this study). Overall, the covariates are well-balanced, including the distribution of baseline beliefs (Belef 2018). The only significant differences are in the percentage of females (p = 0.02), fulltime employed individuals (p = 0.03) and participants with an income below 1000 euro (p = 0.09). Although the experimental design prevents systematic selection and hence the differences are by chance, I will show that controlling for covariates does not affect the results. An interesting insight from Table 4.1 is that at baseline, the majority tends to believe that lobbying decreases the level of climate protection.

¹²A participant was assigned to the experimental group that, at the time of the assignment, had the fewest individuals with the same baseline belief as the participant. In case of a tie, the assignment was randomized with equal probabilities for the respective groups. Note that for technical reasons, the number of individuals with a given baseline belief still differs by more than one across the different groups. Such inaccuracies could occur, for example, when several individuals were assigned at the same time.

	•			
Variable	Control	Group Positive	Group Negative	F-test p -value
Belief 2018				
Much more climate protection	0.01	0.02	0.01	0.98
witch more enhance protection	(0.12)	(0.12)	(0.12)	0.00
Bather more	0.25	0.12)	0.12)	0 99
Rather more	(0.43)	(0.43)	(0.43)	0.00
Neither more nor less	0.17	0.16	0.16	0.97
Nervice more nor less	(0.37)	(0.37)	(0.37)	0.51
Bather less	0.35	0.35	0.36	0.97
	(0.48)	(0.48)	(0.48)	0.01
Much less climate protection	0.22	0.22	0.22	0.94
inden less enniate protection	(0.41)	(0.42)	(0.41)	0.01
Female	0.42	0.48	0.44	0.02
1 childre	(0.49)	(0.50)	(0.50)	0.02
$A \sigma e^a$	51 35	52.32	51.82	0.37
1180	(15, 14)	(15, 83)	(15,31)	0.01
$Married^a$	0.61	0.64	0.62	0.42
Wallied	(0.49)	(0.48)	(0.48)	0.12
Employment	(0.10)	(0.10)	(0.10)	
Full time	0.47	0.43	0.48	0.03
i un onne	(0.50)	(0.50)	(0.50)	0.00
Part time	0.14	0.14	0.12	0.54
	(0.34)	(0.35)	(0.33)	0.01
Betired	(0.34)	(0.35)	(0.33)	0.54
neurea	(0.43)	(0.44)	(0.42)	0.04
Other	0.16	0.18	0.16	0.36
Other	(0.36)	(0.38)	(0.36)	0.00
Internet $usage^a$	0.91	0.91	0.91	0.90
internet usage	(0.28)	(0.28)	(0.29)	0.00
Recruited 2018	0.41	0.20)	0.23	0.84
	(0.49)	(0.49)	(0.49)	0.01
Payment mode (October $2019)^a$	(0.10)	(0.10)	(0.10)	
Donation	0.13	0.14	0.14	0.60
Donation	(0.13)	(0.34)	(0.35)	0.00
Amazon voucher	0.31	(0.34)	0.30	0.45
Amazon vouener	(0.46)	(0.47)	(0.46)	0.40
Bank transfer	0.56	0.54	0.55	0.49
Dank transfer	(0.50)	(0.54)	(0.50)	0.45
$Pro-environmental^a$	(0.90)	(0.00)	(0.00)	0.90
1 to environmental	(0.43)	(0.24)	(0.24)	0.50
Income ^a	(0.10)	(0.12)	(0.12)	
< €1000	0.16	0.19	0.15	0.09
	(0.37)	(0.39)	(0.36)	0.00
[€1000 €2000)	0.31	0.32	0.32	0.68
[01000, 02000)	(0.46)	(0.47)	(0.47)	0.00
[€ 2000 € 3000)	0.28	0.26	0.20	0.35
	(0.20)	(0.20)	(0.25)	0.00
>€3000	0.45	0.23	0.25	0.54
~ 0000	(0.23)	(0.23)	(0.43)	0.04
	(0.40)	(0.42)	(0.40)	
Observations	1014	1031	1026	

 Table 4.1:
 Summary Statistics

Notes: The table reports the averages by treatment group with standard deviations in parentheses. The last column presents *p*-values from regressing each variable on treatment dummies and conducting an *F*-test for the joint significance of the regressors.

^aNot available for all participants. In particular, the number of missing observations amounts to 4 (age), 1 (married), 8 (internet usage), 16 (payment mode), 132 (proenvironmental), and 479 (income).



How does lobbying affect the level of climate protection in the European Union? Lobbying leads to ...

Figure 4.2: Distribution of Beliefs in 2019

Notes: The figure shows the distribution of responses to the belief elicitation question in the survey experiment (November 2019) by treatment group. The (translated) wording of the belief elicitation question is presented at the top of the figure. A χ^2 -test confirms that the distributions of the responses significantly differ between the two treatment groups (*p*-value < 0.001).

4.3 Results

4.3.1 Beliefs

As intended, the experiment induced a substantial gap in beliefs. In particular, participants in Group Negative assessed the impact of lobbying on climate protection more negatively¹³ than participants in Group Positive. Figure 4.2 presents the distributions of beliefs elicited as part of the survey experiment (November 2019) for each of the two treatment groups. In both groups, the majority believed that lobbying has a negative impact. However, this is much more pronounced in Group Negative (74 percent) than in Group Positive (55 percent). The percentage of participants who expected lobbying to increase climate protection is instead larger in Group Positive (19 percent) than in Group Negative (8 percent). A χ^2 -test confirms that the distributions of beliefs significantly differ between the two treatment groups (*p*-value < 0.001).

¹³Throughout the paper, the terms "negative" and "positive" refer to the direction of the impact of lobbying on climate protection when used in this context. They do not express whether an individual thinks that a certain impact is desirable or not.



Figure 4.3: Beliefs over Time

Notes: The figure shows the average likert score for the response to the belief elicitation question by experimental group over time, with the likert scale ranging from -2 (lobbying leads to much less climate protection) to 2 (lobbying leads to much more climate protection). The response "I don't know" is treated as neither expecting a positive nor a negative impact. The error bars represent 95 percent confidence intervals.

Figure 4.3 visualizes the beliefs across the two treatment groups over time and in comparison to the control group. For this purpose, I treat the response "I don't know" as neither expecting a positive nor a negative impact and report the average likert score for each experimental group. This procedure simplifies the presentation of the data but does not affect the conclusions (see Figure D.1 and Figure D.2 in the Appendix for the comparison of the full distributions). In line with Figure 4.2, the average likert score from the response to the belief elicitation question in the survey experiment (November 2019) is significantly higher for individuals in Group Positive than in Group Negative. When we compare these averages to the control group, we observe that reading positive statements significantly increased the average likert score compared to not receiving any statement (p < 0.001, t-test). In contrast, the average likert score is not significantly different between Group Negative and the control group (p = 0.53, t-test). This asymmetric belief adjustment might be explained by the fact that participants were less familiar with the positive than with the negative statements (p < 0.001, t-test). In particular, participants in Group Negative were familiar with on average 2.1 of the three presented statements, while in Group

Positive the average only amounts to 1.5 statements.

In contrast to the strong short-term impact of the belief manipulation, beliefs elicited in the follow-up survey seven months after the experiment do not significantly differ across the experimental groups. Hence, the belief manipulation did not create a persistent effect.

The successful manipulation of beliefs in the short term and the lack of persistence are robust to controlling for available covariates (see Appendix Table D.4 for estimation results from ordered logit regressions). Furthermore, Appendix Figure D.3 illustrates that the impact of the belief manipulation is similar across the different baseline beliefs.

4.3.2 Pro-environmental Behavior

Since the treatment affected beliefs, the next question is whether the exogenous shift in beliefs translated into differences in participants' pro-environmental behavior. I first focus on the main outcome variable, i.e., the contribution to the reduction of carbon emissions via atmosfair. This contribution decision is arguably the most reliable measure available as it constitutes an observed behavior at the time when beliefs differed. Afterward, I turn to the other available outcome variables, which mostly consist of stated behaviors (see Appendix Table D.5 for an overview of the outcome variables and their summary statistics by treatment group). Since these variables concern behaviors in the months after the experimental intervention, the lack of persistence in the belief manipulation might limit the scope to identify differences, particularly if beliefs across the treatment groups converged quickly.

In the survey experiment, about 66 percent of participants contributed to the climate protection organization atmosfair, and the average contribution over all participants was 2.35 euro. In general, the correlation between contributions and individuals' baseline beliefs about the impact of lobbying on climate protection is significantly negative (r = -0.12). However, the correlation might be misleading as it does not account for factors that jointly determine beliefs and pro-environmental behavior.

Figure 4.4 shows the distribution of contributions to reducing carbon emissions by treatment group. In each group, more than 50 percent of the participants contributed the maximum amount of 4 euro, and another large percentage



Figure 4.4: Contributions to CO_2 Reduction via atmosfair

Notes: The figure shows the percentage of subjects who chose a particular contribution to atmosfair in the survey experiment (November 2019) by treatment group. The average contribution does not significantly differ across the two treatment groups (p = 0.22, *t*-test).

of more than 30 percent did not contribute at all. Less extreme contributions between 1 and 3 euro are rare. The average contribution amounts to 2.40 euro in Group Positive and 2.30 euro in Group Negative. Although individuals in Group Negative gave slightly less, the difference between the two groups is not significantly different (p = 0.22, t-test).

Table 4.2 reports estimation results from regressing individual contributions on a dummy for whether the participant was assigned to Group Negative and different sets of covariates. The estimated coefficient of receiving negative (instead of positive) statements is always slightly negative but never significantly different from zero. Hence, based on the main outcome variable, there is little evidence that the exogenous shift in beliefs affected individuals' pro-environmental behavior.

Next, I consider the other measures of pro-environmental behavior, starting with the sum of donations to environmental charities in the context of the GIP. This outcome variable considers the contribution to the reduction of carbon emissions via atmosfair and the money transferred to WWF (which is one-third of the participant's cumulative earnings if the participants decided to donate the cumulative earnings since November 2019 in April 2020). The average donation
	\mathbf{C}	Contribution						
Variable	(1)	(2)	(3)					
Group Negative	-0.10 (0.08)	-0.09 (0.08)	-0.09 (0.09)					
Belief 2018	Yes	Yes	Yes					
Female		Yes	Yes					
Age		Yes	Yes					
Married		Yes	Yes					
Employment		Yes	Yes					
Internet usage		Yes	Yes					
Recruited 2018		Yes	Yes					
Payment mode		Yes	Yes					
Income			Yes					
Pro-environmental			Yes					
Observations	2057	2042	1687					

 Table 4.2:
 Estimation Results – Pro-environmental Behavior

Notes: Robust standard errors are in parentheses. Each column corresponds to a separate OLS regression with the contribution to the reduction of carbon emissions via atmosfair as dependent variable. *p < 0.1, **p < 0.05, ***p < 0.01.

to both charities amounts to 3.05 euro in Group Negative and 3.06 euro in Group Positive. Hence, the absolute difference in the behavior between the two groups becomes smaller and remains insignificant after accounting for a potential substitution between the donation to atmosfair and the donation to WWF. For both outcome variables, the standardized effect of being assigned to Group Negative is illustrated in the upper part of Figure 4.5.

The lower part of Figure 4.5 shows the standardized effects for the stated proenvironmental behaviors. Since I measure a wide range of behaviors, I create an index from the separate measures following the procedure proposed by Kling et al. (2007). In particular, I normalize each variable such that it has a mean of zero and a standard deviation of 1 in Group Positive. Afterward, I code each variable such that a higher value represents more pro-environmental behavior and calculate the average of the normalized variables for each individual to obtain the stated behavior index. The major advantage of using the index is that it increases the power to detect differences if each outcome variable is affected in the same direction (Kling et al., 2007).

As shown in Figure 4.5, negative beliefs (caused by reading negative statements) significantly reduced the stated behavior index by about 0.13 standard



Figure 4.5: Standardized Treatment Effects

Notes: The figure shows the standardized treatment effects of being assigned to Group Negative instead of Group Positive for different outcome variables. The effects are standardized by the standard deviation of the respective outcome variable in Group Positive and are obtained from regressing the outcome variable on treatment assignment and indicator variables to control for baseline beliefs. The error bars represent 95 percent confidence intervals (based on heteroskedasticity robust standard errors). The red arrows indicate the directions that represent less pro-environmental behavior.

deviations, suggesting that a more negative belief leads to less pro-environmental behavior. A look at the separate behaviors included in the index reveals that the effect is substantially driven by a large and significant standardized effect on the likelihood of taking a plane. In non-standardized terms, the treatment is estimated to double the likelihood of taking a plane from 3.5 to 7 percent. However, all stated behaviors point into the same direction (i.e., less pro-environmental behavior in Group Negative). The effect on the stated behavior index and the effect on taking a plane remain significant after controlling for the number of considered outcome variables in Figure 4.5 via the Holm multiple hypothesis correction (Holm, 1979). In particular, the corrected *p*-values amount to 0.06 and 0.01, respectively. Hence, in contrast to the observed behavior, I do find evidence of a causal impact of beliefs about lobbying on stated pro-environmental behavior.

4.4 Concluding Remarks

In this paper, I have investigated whether individuals' beliefs about the impact of lobbying on climate protection affect pro-environmental behavior. To identify the causal effect of beliefs, I have employed a survey experiment with a large heterogeneous sample from Germany. In the experiment, individuals were randomly induced to expect a more negative or more positive impact of lobbying on the level of climate protection.

The results offer some evidence of a causal impact of beliefs on proenvironmental behavior. Individuals who received a negative (instead of a positive) shock to their beliefs reported significantly less pro-environmental behavior. However, the significance is mainly driven by a single behavior (taking a plane). Furthermore, I do not find an effect on observed pro-environmental behavior, i.e., the contribution to the reduction of carbon emissions: The corresponding estimate points in the same direction but is insignificant.

One concern might be that the difference in beliefs caused by the experimental treatments was insufficient to trigger a strong behavioral response. Figure 4.3 reveals that the difference in beliefs between Group Positive and Group Negative after the treamtent is similar to the shift in beliefs that naturally occurred in the control group within the course of one year (November 2018 until November 2019). In this regard, the manipulation appears to be well suited to assess the consequence of a realistic belief change. Nevertheless, it is possible that a more extreme difference in beliefs would have resulted in a significant effect even for the observed pro-environmental behavior.

The above considerations suggest that additional research is needed before concluding about whether the perception of lobbying affects individual behavior. If a causal link exists, the question of why individuals hold such negative beliefs gains importance. The experimental data shows that in November 2018, about 57 percent of the participants in the control group believed that lobbying decreases the level of climate protection in the European Union, and this share has increased over time (in the absence of any belief manipulation). If there is a lack of information that leads to overly pessimistic beliefs, increasing the transparency of lobbying activities could help correct beliefs and mitigate adverse spillover effects. For example, the European Parliament, Council, and Commission have recently agreed upon establishing a reinforced Transparency Register in order to "strengthen transparent and ethical interest representation" (European Comission, 2020).

Appendices

A Appendix to Chapter 1

A.1 Mapping of Paper and Pre-analysis Plan

We conducted the experiment as described in our pre-analysis plan. As the corona pandemic prevented us from running some planned experimental sessions, our sample size of 330 subjects is smaller than the planned sample of 500 subjects. Similarly, the observation periods used in our main specification (August 1, 2019, until March 16, 2020, in Mannheim and October 1, 2019, until March 19, 2020, in Bonn) are shorter than pre-specified due to the shut down of the university canteens to fight the spread of the coronavirus (see Section 1.2.4 for details). Table A.1 maps the hypotheses in our pre-analysis plan to the analyses in the paper.¹⁴

 $^{^{14}\}mathrm{Note}$ that in the pre-analysis plan, we used the term *meat* to refer to the meat from terrestrial animals.

 Table A.1:
 Mapping of Paper and Pre-analysis Plan

Description	Pre-analysis plan	Paper
Info avoidance	Hypothesis 3	Results are presented in Section 1.4.1 of the paper.
Average info effect in lab	Hypothesis 1a	Results are presented in Section 1.4.2 of the paper.
Average info effect in field	Hypothesis 1b	We pre-specified to test this hypothesis with the propen- sity to eat meat as well as the propensity to eat meat from terrestrial animals as outcome variable. While the former was planned as robustness check, we use it in our main analysis as it is directly comparable to the out- come variable in the lab. Results for the propensity to eat meat are presented in Section 1.4.3. Results for the propensity to eat meat from terrestrial animals are pre- sented in Table A.22 (Appendix A.8) and discussed in Section 1.4.5. In cases where the data is insufficient to determine whether a meal contains meat for sure, we planned to code it as containing meat if it usually does. This turned out to be difficult, however, as ambiguous food items were almost always offered with and without meat. Therefore, we consider two outcome variables: the share of meals that certainly contain meat and the share of meals that certainly do not contain meat.
Average info effect for avoiders in lab	Hypothesis 4a	Results are presented in Section 1.4.4 of the paper.
Average info effect for avoiders in field	Hypothesis 4b	We code outcome variables as for Hypothesis 1b. Results for the propensity to eat meat are presented in Section 1.4.4. Results for the propensity to eat meat from ter- restrial animals are presented in Table A.29 (Appendix A.8).
Heterogeneous info effect by info de- mand in lab	Hypothesis 2a	We pre-specified to test the heterogeneity of the info effect by splitting subjects into two groups based on the median of the WTP distribution. The median WTP interval is $[0, 0.5)$ euro. We planned to allocate subjects with the median WTP such that the groups are as balanced as possible in terms of sample size. This implies a split of subjects into those with WTP < 0.5 (67 percent) and those with WTP ≥ 0.5 (33 percent). The alternative allocation of subjects with the median WTP leads to the split into information avoiders (WTP < 0, 30 percent) and seekers (WTP $\geq 0, 70$ percent). Since the alternative split has almost the same balance but a clearer interpretation (avoiders vs. seekers), we use it in our main analysis (see Section 1.4.4). Results for the other split are presented in Table A.27 (Appendix A.8).
Heterogeneous info effect by info de- mand in field	Hypothesis 2b	We code outcome variables as for Hypothesis 1b and split subjects as for Hypothesis 2a. Results for the compar- ison of the effect for information avoiders and seekers using the propensity to eat meat as outcome variable are presented in Section 1.4.4. Results for the propensity to eat meat from terrestrial animals as outcome variable are presented in Table A.29 (Appendix A.8) and those for the alternative split in Table A.28 (Appendix A.8).

A.2 Theoretical Predictions about Heterogeneity

A variety of theoretical models are able to explain information avoidance in a moral context (see, e.g., Rabin, 1995; Feiler, 2014; Grossman and van der Weele, 2017; Hestermann et al., 2020; Serra-Garcia and Szech, 2020). In the following, we use two examples, Hestermann et al. (2020) and Grossman and van der Weele (2017), to illustrate that these models make ambiguous predictions about treatment effect heterogeneity. In particular, we illustrate that the effect of information on individuals who avoid costless information (information avoiders) and those who do not avoid costless information (information seekers) depends on the model parameters. Similar ambiguity can also be shown for other theoretical models (Rabin, 1995; Feiler, 2014; Serra-Garcia and Szech, 2020).¹⁵

A.2.1 Self-Deception Model by Hestermann et al. (2020)

We start with the theoretical model by Hestermann et al. (2020), who explicitly focus on the context of meat consumption. In their model, an individual has two selves: Self 0 can obtain an information signal about the state of the world. The information signal is either *bad news* and suggests that animals in the meat production suffer much (x_H) or good news suggesting that animals suffer little (x_L) . After receiving the information signal, Self 0 decides what information to transmit to Self 1. Based on the transmitted information, Self 1 forms a belief \tilde{x} about the true state of the world and chooses a level of meat consumption cthat maximizes the utility function

$$\max_{c \in \mathbb{R}_+} U(c) - (p + w\tilde{x})c,$$

where $U(\cdot)$ represents the taste for meat, p is the price of meat, and w is the individual's feeling of empathy towards animals. In general, an individual faces the tradeoff between consuming meat and not harming animals. The key feature of the model is to allow for self-deception. In particular, Self 0's utility from meat consumption depends on Self 1's belief and can be expressed as $U(c^*(\tilde{x})) - (p + w\tilde{x})c^*(\tilde{x})$, where $c^*(\tilde{x})$ denotes the optimal consumption level of Self 1 given

¹⁵In the case of the basic model by Rabin (1995) and a binary perfect information signal, avoiders always respond to the information that the action creates social harm and seekers either respond in the same way or less (in absolute terms). Under the assumption of fixed-belief behaviorally equivalent utility functions (see Definition 3 in Rabin, 1995), information avoiders and seekers react in exactly the same way to receiving information.

belief \tilde{x} . As a result, Self 0 does not necessarily transmit the actual information received but might distort the information to lower the moral costs from meat consumption. However, distorting information creates some cognitive cost, k, and Self 1 takes into account Self 0's tendency to engage in self-deception when forming beliefs (see equations 4 and 5 in Hestermann et al., 2020). The avoidance of information allows Self 0 to keep Self 1 uninformed without engaging in costly self-deception.

In the following, we investigate how information avoiders and information seekers react to receiving bad news, i.e., the information that animals suffer.¹⁶ We use examples of different parameter values to show that the model does not provide unanimous conclusions about treatment effect heterogeneity by information demand. In particular, we show that information avoiders have larger treatment effects than information seekers for some parameter values, while the reverse holds true for other parameter values.

Assume $U(c) = \sqrt{c}$, p = 0.1, $x_L = 0$ and $x_H = 0.9$. Under these assumptions, an individual with a moderate empathy towards animals (w = 0.3) and a high cost of self-deception (k = 0.8) engages in partial self-deception. Nevertheless, one can show that the individual prefers to receive costless information over not receiving it and can hence be classified as an information seeker. Bad news decrease the individual's meat consumption compared to the case without information (by about 0.3 units).

Compare this to an individual with the same cost of self-deception (k = 0.8)but a lower empathy towards animals (w = 0.1). This individual also engages in partial self-deception (with a slightly lower tendency of being self-deceptive). However, the individual does not prefer to receive costless information over not receiving it and is thus an information avoider. Comparing the consumption level without information to the consumption level under bad news suggests a negative effect of information on consumption (by about 2.2 units). Hence, in this example, the information avoider responds more strongly to information than the information seeker.¹⁷

¹⁶In our experiment, bad news correspond to watching the video about intensive farming, which induced subjects to adjust their beliefs towards worse living conditions of pigs on average (see Section A.4).

¹⁷At first sight, it might be counterintuitive that an individual with a lower empathy towards animals is more responsive to information on animal welfare. The main reason is that a lower empathy leads to generally higher levels of meat consumption. Given the concavity of the utility function, this can result in a larger absolute change in meat consumption when beliefs

A Appendix to Chapter 1

However, for other parameter values, information avoiders are less responsive to information than information seekers. Consider an individual with a low empathy towards animals (w = 0.1) and a low cost of self-deception (k = 0.1). The equilibrium strategy of this individual is to engage in full self-deception. The individual also avoids costless information about the true state of the world as this allows to stay uninformed without incurring the cost of self-deception. Since the individual engages in full self-deception, receiving bad news does not affect beliefs and thus does not have any impact on meat consumption.

A.2.2 Self-Image Model by Grossman and van der Weele (2017)

Another model of information avoidance is proposed by Grossman and van der Weele (2017). In their model, individuals avoid information to protect their selfimage despite behaving selfishly. Individuals have again two selves: A "decision maker self" decides whether to receive information about the state of the world and whether to behave prosocially. Depending on the state of the world, taking a costly action a improves social welfare or not. An "observer self" observes the equilibrium choices under the endogenous information state and uses them to infer the social preferences of the decision self. The utility function maximized by the decision self is given by

$$u(\theta, a, I, \sigma) = a(\theta E[W|\sigma] - c) - kI + \mu E[\theta|\sigma, a],$$

where a is the binary choice to behave prosocially or not, $\theta E[W|\sigma]$ and c are the expected benefit and the cost of behaving prosocially, respectively, μ is the selfimage concern, and $E[\theta|\sigma, a]$ is the expected social preference based on the chosen actions. In our setting, the preference parameter θ could represent the empathy towards animals, $E[W|\sigma]$ could capture the expected increase in animal welfare from not eating meat, a the decision to eat meat and c the utility from meat consumption. In the following, we use the terminology of the original model.

The authors assume that individuals differ in their preferences as follows. A fraction ε of the population does not have social preferences ($\theta = 0$) and does not care about self-image ($\mu = 0$). These *nonsocial* individuals will attend to costless information and behave selfishly if $c \ge 0$. All other individuals care about their self-image ($\mu > 0$) and have a social preference θ which is drawn

change.

from a distribution with positive support on [0, 1]. For individuals with a low θ , it can be attractive to avoid information as it allows them to differentiate themselves from the nonsocial individuals without choosing the prosocial action.

The authors show the existence of a semiseparating equilibrium with k = 0 in which (i) nonsocial types attend to information and choose a = 0, (ii) individuals with a social preference parameter below some cutoff θ^* avoid information and behave selfishly, and (iii) individuals with a social preference parameter above the cutoff attend to information and choose the prosocial action if it improves social welfare. Several parameter combinations are possible to establish this equilibrium. In the following, we show that within this set of possible parameter combinations the prediction on how information seekers (which includes nonsocial agents) and information avoiders react to information is ambiguous.

To identify the information effect, we compare the behavior of each type if no one receives information and if everyone receives *bad news* in the sense that a costly action *a* improves welfare. We focus on semiseparating equilibria with some cutoff value θ^* , where all individuals with θ equal or above the cutoff choose a = 1 and all individuals with θ below the cutoff choose a = 0. Since the selfimage from an action does not only depend on the own action but also on the behavior of all others, we extend the simulation code published by Grossman and van der Weele (2017) to calculate the heterogeneous information effects under different parameter values.

As a starting point, we use the parameter values introduced in Section 5.3 of the paper by Grossman and van der Weele (2017), i.e., $\varepsilon = 0.1$, p = 0.5, k = 0, w = 1.65, $\mu = 0.95$, c = 1, θ has a truncated normal distribution on [0, 1] with a mean of 0.5 and standard deviation of 0.1. Under these assumptions, receiving bad news reduces the share of selfish behavior by 0.46 percentage points for information avoiders and 0.82 percentage points for information seekers. If we instead assume a slightly larger share of nonsocial agents ($\varepsilon = 0.2$), bad news reduce the share of selfish behavior by 0.88 percentage points for information avoiders and only 0.61 for information seekers. The effect sizes change mainly because the nonsocial agents do not react to information and now make up a larger share of the information seekers. Furthermore, the larger share of nonsocial agents increases the self-image cost of choosing the selfish action under full information and thus lowers the cutoff value of θ at which a social agent will choose the prosocial action when informed about a welfare impact. It is also possible to select parameter values such that information avoiders are predicted to not react to bad news at all (initial parameter values but with c =1.25). In this case, the costs of behaving prosocially are too high for information avoiders. Hence, they all prefer a lower self-image from behaving selfishly under full information instead of taking the costly prosocial action.

A.3 Estimation Details

A.3.1 AIPW Estimator

Let T_i denote whether individual *i* receives treatment $(T_i = 1)$ or not $(T_i = 0)$. The AIPW estimator can be expressed as:

$$\hat{\tau}_{AIPW} = \frac{1}{N} \sum_{i=1}^{N} \frac{T_i Y_i - (T_i - \hat{p}(WTP_i)) m_1(\boldsymbol{X}_i, \hat{\boldsymbol{\beta}}_1)}{\hat{p}(WTP_i)} - \frac{1}{N} \sum_{i=1}^{N} \frac{(1 - T_i) Y_i + (T_i - \hat{p}(WTP_i)) m_0(\boldsymbol{X}_i, \hat{\boldsymbol{\beta}}_0)}{1 - \hat{p}(WTP_i)},$$
(A.1)

where WTP_i is the WTP of the individual, Y_i is the individual's propensity to eat meat, and $\hat{p}(.)$ is the estimated propensity of being treated which we know only depends on the WTP. $m_0(\mathbf{X}_i, \hat{\boldsymbol{\beta}}_0)$ and $m_1(\mathbf{X}_i, \hat{\boldsymbol{\beta}}_1)$ are the predicted levels of the outcome variable for being in the control group and being treated, respectively. When the outcome variable is binary or fractional, we use a logit model to obtain predicted outcome levels. Otherwise, we apply OLS. The parameters $\hat{\boldsymbol{\beta}}_t$ are estimated based on the data from subjects with T = t. The "augmentation" of the inverse probability weighting estimator makes the AIPW estimator doubly robust (Lunceford and Davidian, 2004), which means that it is consistent if at least the propensity score model or the outcome models are correctly specified.

A.3.2 IPW Estimator Based on True Propensity Scores

The IPW estimator based on the true propensity scores can be expressed as

$$\hat{\tau}_{IPW} = \left(\sum_{i=1}^{N} \frac{T_i}{p(WTP_i)}\right)^{-1} \sum_{i=1}^{N} \frac{T_i Y_i}{p(WTP_i)} - \left(\sum_{i=1}^{N} \frac{1 - T_i}{1 - p(WTP_i)}\right)^{-1} \sum_{i=1}^{N} \frac{(1 - T_i) Y_i}{1 - p(WTP_i)} = \hat{\mu}_{T=1} - \hat{\mu}_{T=0},$$
(A.2)

where p(.) is the true propensity of being treated that follows directly from the experimental design. In contrast to the estimator presented in equation 1.1, we replace the factor $\frac{1}{N}$ by the terms $\left(\sum_{i=1}^{N} \frac{T_i}{p(WTP_i)}\right)^{-1}$ and $\left(\sum_{i=1}^{N} \frac{1-T_i}{1-p(WTP_i)}\right)^{-1}$ to normalize the weights, which ensures that they add up to one. When using our

empirical propensity score estimates, this normalization becomes redundant.

A.4 Beliefs

We elicited beliefs about the living conditions of pigs in intensive farming as part of the questionnaire before subjects completed the multiple price list. The answer categories range from (1) very good to (5) very bad. On average, living conditions were expected to be bad with a likert score of about 4.62. 70 percent of subjects (230 out of 330) selected the most negative answer and less than 2 percent (6 out of 330) assessed the conditions as good or very good (see top left graph of Figure A.1). To evaluate whether the information signal had an impact on beliefs, we asked subjects who received the information to assess the living conditions again after watching the video about intensive farming. For these subjects, the average likert score increased from 4.66 to 4.86 and a rank sum test confirms that the change in beliefs is statistically significant (p < 0.001).

In addition, we also elicited beliefs about the living conditions of chicken to investigate whether individuals used the information to make inference about the living conditions of other animals. The conditions were again expected to be bad ex ante (average likert score of 4.66, distribution of responses presented in top right graph of Figure A.1), but we do not find a significant adjustment of beliefs after watching the video about intensive farming (p = 0.64, rank sum test). Figures A.1 and A.2 also show the distributions of responses to other questions which were related to intensive farming and meat consumption but were only asked before eliciting subjects' WTP for information.



Figure A.1: Responses Related to Intensive Farming

Notes: The figure shows the distributions of responses for different questions related to intensive farming. All questions were asked before eliciting subjects' WTP for information. Each graph is based on 330 observations.



Figure A.2: Responses Related to Meat Consumption

Notes: The figure shows the distributions of responses for different questions related to meat consumption. All questions were asked before eliciting subjects' WTP for information. Each graph is based on 330 observations.

A.5 Robustness Checks

Since we conduct a controlled experiment, we know the probability of receiving the information on intensive farming for each subject and can use the true instead of the empirical propensity scores when estimating the effect of information (see equation A.2 in Appendix A.3.2). Tables A.2 to A.5 show that the estimation results from this alternative strategy are very similar to our main findings presented in the paper. We also ensure that our results are robust to directly specifying a model for the outcome variable via OLS regressions (see Table A.6 and Table A.7).

Furthermore, we investigate whether there are important differences in the results for the two locations, Bonn and Mannheim. Figure A.3 shows that the demand curves for information are almost exactly the same across the two locations. To estimate location-specific information effects, we apply the IPW and WLS estimator separately to the data from Mannheim and Bonn, respectively. Since this is demanding in terms of sample size, results should only be taken as suggestive evidence. The estimates from the field do not significantly differ between the locations (see Table A.10 and Table A.11). In the laboratory, it seems that information seekers $(WTP \ge 0)$ in Mannheim respond less than information seekers in Bonn (see Table A.9) and the difference between the estimates is weakly significant for each specification (p < 0.1). The response of information avoiders (WTP < 0) is instead very similar. This pattern leads to a smaller point estimate for the average information effect in Mannheim (see Table A.8), which results in a marginally significant difference compared to the point estimate from Bonn in the case of the WLS estimator (p < 0.1). Hence, there is suggestive evidence of a difference between locations for a subgroup of individuals in the laboratory, but this difference does not show up in the field.

	Dependent variable: Choosing voucher for meal with meat						
	IPW	W	WLS		PW		
	(1)	(2)	(3)	(4)	(5)		
Info effect	-0.168^{***} (0.055)	-0.156^{***} (0.045)	-0.143^{***} (0.045)	-0.153^{***} (0.044)	-0.135^{***} (0.043)		
Mean~(w/o~info)	0.461	0.454	0.447	0.449	0.441		
Meat-eating habit Additional controls Observations	No No 312	Yes No 312	Yes Yes 312	Yes No 312	Yes Yes 312		

Notes: Standard errors are in parentheses. Mean ($w/o \ info$) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2 to 5, we use indicator variables. Since never eating meat is a perfect predictor of choosing the vegetarian voucher, the expected likelihood for individuals with this meat-eating habit is not predicted via a logit model but directly set to zero for the models in columns 4 and 5. Additional controls are age, gender, degree, location, and WTP. All estimates are based on the true instead of the empirical propensity scores. *p < 0.1, **p < 0.05, ***p < 0.01.

 Table A.3:
 Information Effect in the Field Based on True Propensity Scores

	IPW	WLS		AI	PW
	(1)	(2)	(3)	(4)	(5)
A. Share of meals with meat					
Info effect	-0.086**	-0.059*	-0.067**	-0.060*	-0.066**
	(0.044)	(0.033)	(0.032)	(0.033)	(0.032)
Mean (w/o info)	0.457	0.440	0.444	0.440	0.450
B. Share of meals without meat					
Info effect	0.099^{**}	0.054	0.090^{***}	0.053	0.083^{**}
	(0.043)	(0.036)	(0.034)	(0.036)	(0.035)
Mean (w/o info)	0.402	0.430	0.411	0.429	0.410
Baseline level	No	Yes	Yes	Yes	Yes
Additional controls	No	No	Yes	No	Yes
Observations	261	261	261	261	261

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. All estimates are based on the true instead of the empirical propensity scores.

 $p^* p < 0.1, p^* < 0.05, p^* < 0.01.$

	Dependent variable: Choosing voucher for meal with meat				
	IPW	W	LS		
	(1)	(2)	(3)		
Avoider					
Info effect	-0.309^{***} (0.097)	-0.236^{***} (0.077)	-0.229^{***} (0.073)		
Mean $(w/o info)$	0.530	0.503	0.500		
Seeker					
Info effect	-0.095 (0.066)	-0.100^{*} (0.054)	-0.102^{*} (0.055)		
Mean $(w/o info)$	0.403	0.408	0.409		
Comparison of effects (<i>p</i> -value)	0.07	0.15	0.16		
Meat-eating habit	No	Yes	Yes		
Additional controls	No	No	Yes		
Observations	312	312	312		

Table A.4: Information Effect in the Laboratory Based on True Propensity Scores – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2 to 5, we use indicator variables. Additional controls are age, gender, degree, location, and WTP. All estimates are based on the true instead of the empirical propensity

	IPW	W	LS
	(1)	(2)	(3)
A. Share of meals with meat			
Avoider			
Info effect	-0.106	-0.001	-0.006
	(0.086)	(0.066)	(0.065)
Mean $(w/o info)$	0.453	0.412	0.415
Seeker			
Info effect	-0.084	-0.084**	-0.089**
	(0.052)	(0.037)	(0.037)
Mean $(w/o info)$	0.460	0.460	0.463
Comparison of effects $(p-value)$	0.82	0.27	0.27
B. Share of meals without meat			
Avoider			
Info effect	0.198^{**}	0.048	0.069
	(0.085)	(0.073)	(0.071)
Mean $(w/o info)$	0.377	0.436	0.424
Seeker			
Info effect	0.062	0.073^{*}	0.095^{**}
	(0.052)	(0.039)	(0.039)
Mean $(w/o info)$	0.423	0.417	0.406
Comparison of effects $(p-value)$	0.17	0.76	0.75
Baseline level	No	Yes	Yes
Additional controls	No	No	Yes
Observations	261	261	261

Table A.5: Information Effect in the Field Based on True Propensity Scores – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. All estimates are based on the true instead of the empirical propensity scores.

 $^{*}p<0.1,\ ^{**}p<0.05,\ ^{***}p<0.01.$

	Labor	Laboratory		Field			
	Meal wi	Meal with meat		th meat	Meal w	Meal w/o meat	
	(1)	(2)	(3)	(4)	(5)	(6)	
Info effect	-0.130***	-0.128***	-0.064**	-0.067**	0.088**	0.088***	
	(0.047)	(0.047)	(0.032)	(0.032)	(0.034)	(0.034)	
Mean~(w/o~info)	0.439	0.438	0.446	0.448	0.410	0.410	
Meat-eating habit	Yes	Yes	No	No	No	No	
Baseline level	No	No	Yes	Yes	Yes	Yes	
WTP	Yes	Yes	Yes	Yes	Yes	Yes	
Additional controls	No	Yes	No	Yes	No	Yes	
Observations	312	312	261	261	261	261	

Table A.6: Information Effect Based on OLS

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit and WTP, we use indicator variables. Additional controls are age, gender, degree, and location. All estimates are based on OLS regressions. In columns 1 and 2, the dependent variable is whether the individuals chose the voucher for the meal with meat. In column 3 and 4, the dependent variable is the share of meals with meat. In columns 5 and 6, the dependent variable is the share of meals without meat.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	Laboratory Meal with meat		Field			
			Meal wi	th meat	Meal w/o meat	
	(1)	(2)	(3)	(4)	(5)	(6)
Avoider						
Info effect	-0.271***	-0.260***	-0.022	-0.025	0.101	0.102
	(0.082)	(0.079)	(0.062)	(0.063)	(0.066)	(0.067)
Mean $(w/o info)$	0.522	0.517	0.422	0.423	0.407	0.406
Seeker						
Info effect	-0.071	-0.074	-0.081**	-0.083**	0.083**	0.083**
	(0.057)	(0.056)	(0.037)	(0.036)	(0.039)	(0.039)
Mean $(w/o info)$	0.393	0.395	0.460	0.460	0.413	0.412
Comparison of effects $(p-value)$	0.04	0.05	0.41	0.42	0.81	0.80
Meat-eating habit	Yes	Yes	No	No	No	No
Baseline level	No	No	Yes	Yes	Yes	Yes
WTP	Yes	Yes	Yes	Yes	Yes	Yes
Additional controls	No	Yes	No	Yes	No	Yes
Observations	312	312	261	261	261	261

Table A.7: Information Effect Based on OLS – Avoider and S	eeke
--	------

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit and WTP, we use indicator variables. Additional controls are age, gender, degree, and location. All estimates are based on OLS regressions. In columns 1 and 2, the dependent variable is whether the individuals chose the voucher for the meal with meat. In column 3 and 4, the dependent variable is the share of meals with meat. In columns 5 and 6, the dependent variable is the share of meals without meat.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

Appendices



Figure A.3: Demand for Information by Location

Notes: For each location, the figure shows the percentage of subjects choosing to receive information about intensive farming at a given price from the multiple price list.

	Dependent variable: Choosing voucher for meal with meat							
		Mannheim			Bonn			
	IPW	WLS		/ WLS		IPW	W	LS
	(1)	(2)	(3)	(4)	(5)	(6)		
Info effect	-0.104 (0.067)	-0.049 (0.070)	-0.052 (0.067)	-0.248^{***} (0.088)	-0.252^{***} (0.080)	-0.216^{***} (0.073)		
Mean~(w/o~info)	0.424	0.388	0.390	0.495	0.501	0.483		
Meat-eating habit Additional controls Observations	No No 197	Yes No 197	Yes Yes 197	No No 115	Yes No 115	Yes Yes 115		

Table A.8: Information Effect in the Laboratory by Location

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2, 3, 5, and 6, we use indicator variables. Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.

	Dependent variable: Choosing voucher for meal with meat						
		Mannheim	l		Bonn		
	IPW	W	LS	IPW	WLS		
	(1)	(2)	(3)	(4)	(5)	(6)	
Avoider							
Info effect	-0.322***	-0.238***	-0.248***	-0.359**	-0.278*	-0.168	
	(0.108)	(0.091)	(0.093)	(0.156)	(0.143)	(0.120)	
Mean $(w/o info)$	0.377	0.436	0.424	0.377	0.436	0.424	
Seeker							
Info effect	0.001	0.044	0.040	-0.209**	-0.243***	-0.232***	
	(0.083)	(0.091)	(0.091)	(0.105)	(0.093)	(0.088)	
Mean $(w/o info)$	0.346	0.313	0.316	0.476	0.498	0.493	
Comparison of effects $(p$ -value)	0.02	0.03	0.04	0.42	0.84	0.67	
Meat-eating habit	No	Yes	Yes	No	Yes	Yes	
Additional controls	No	No	Yes	No	No	Yes	
Observations	197	197	197	115	115	115	

Table A.9: Information Effect in the Laboratory by Location – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2, 3, 5, and 6, we use indicator variables. Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	Mannheim			Bonn		
	IPW	W	LS	IPW	WLS	
	(1)	(2)	(3)	(4)	(5)	(6)
A. Share of meals with meat						
Info effect	-0.105**	-0.061	-0.059	-0.182***	-0.087	-0.086
	(0.050)	(0.050)	(0.049)	(0.069)	(0.053)	(0.055)
Mean (w/o info)	0.487	0.469	0.467	0.486	0.443	0.442
B. Share of meals without meat						
Info effect	0.127^{***}	0.075	0.071	0.184^{***}	0.120^{*}	0.112
	(0.049)	(0.058)	(0.052)	(0.071)	(0.068)	(0.068)
Mean (w/o info)	0.356	0.380	0.382	0.413	0.439	0.444
Baseline level	No	Yes	Yes	No	Yes	Yes
Additional controls	No	No	Yes	No	No	Yes
Observations	173	173	173	88	88	88

Table A.10: Information Effect in the Field by Location

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP.

 $\tilde{p} < 0.1, \ \tilde{p} < 0.05, \ \tilde{p} < 0.01.$

	Ν	Iannheim	1	Bonn			
	IPW	W	LS	IPW	W	LS	
	(1)	(2)	(3)	(4)	(5)	(6)	
A. Share of meals with meat							
Avoider							
Info effect	-0.120	-0.003	-0.001	-0.245^{**}	-0.034	-0.015 (0.114)	
Mean (w/o info)	(0.000) 0.377	0 436	(0.011) 0.424	(0.117) 0.377	0 436	0 424	
Seeker	0.011	0.100	0.121	0.011	0.100	0.121	
Info effect	-0.097 (0.060)	-0.088 (0.062)	-0.085 (0.061)	-0.159^{*} (0.084)	-0.104^{*} (0.057)	-0.105^{*} (0.058)	
Mean (w/o info)	0.423	0.417	0.406	0.423	0.417	0.406	
Comparison of effects $(p-value)$	0.83	0.38	0.38	0.55	0.59	0.46	
B. Share of meals without meat							
Avoider							
Info effect	0.197**	0.046	0.041	0.363***	0.161	0.138	
	(0.084)	(0.070)	(0.068)	(0.105)	(0.145)	(0.140)	
Mean $(w/o info)$	0.377	0.436	0.424	0.377	0.436	0.424	
Seeker							
Info effect	0.094	0.088	0.084	0.121	0.107	0.104	
	(0.060)	(0.075)	(0.068)	(0.088)	(0.074)	(0.074)	
Mean (w/o info)	0.423	0.417	0.406	0.423	0.417	0.406	
Comparison of effects $(p-value)$	0.32	0.68	0.65	0.08	0.73	0.82	
Baseline level	No	Yes	Yes	No	Yes	Yes	
Additional controls	No	No	Yes	No	No	Yes	
Observations	173	173	173	88	88	88	

Table A.11: Information Effect in the Field by Location – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.

A.6 Instructions for WTP Elicitation

In the following, we present the translated instructions for the WTP elicitation task. The only difference across locations was that in Mannheim, the electronic payment card is called "ecUM" while in Bonn, it is called "MensaCard". Whether the video about the living conditions of pigs in intensive farming was presented as Option A or Option B was randomized, and the comprehension question was adjusted accordingly. Figure A.4 shows a screenshot of the decision screen.

Start	of	the	instructions
No cour c	· · ·	0110	1110 01 010 010 110

In a few minutes you will watch another 360° video. There are two different options available to you.

On the next few pages, we present the two options and explain how to choose one of the options. Only then will you make your decision(s).

Page break

You have the following options to choose from:

Option A: You watch a 360° video about intensive farming of pigs.

If you choose this option, you'll see excerpts from a video by the organization Animal Equality. This video shows the life of pigs in intensive farming, from birth until slaughter. The scenes were recorded between December 2014 and January 2016 in Germany, Italy, Mexico and Spain. According to Animal Equality, all the scenes correspond to standard practice in Europe.

Please note: This video contains scenes in which blood can be seen and which may be shocking.

Option B: You watch the 360° video about the Deutsche Bundesbank again.

If you choose this option, you will again see the excerpts from the video of the Deutsche Bundesbank. This video shows the virtual tour through the Bundesbank building and corresponds exactly to the video that you saw earlier. The scenes were recorded in 2019. Most of the rooms shown are part of the official tour of the Bundesbank building.

The two videos are of the same length (approx. 5 minutes).

Page break

You can choose between the two options as follows:

You make a total of 11 decisions, at each of which you choose one of the two options. The only difference between the 11 decisions is for which option you receive a bonus payment and how high this bonus payment is.

One of your 11 decisions is then randomly selected and implemented. This means that you watch the video selected in this decision and receive the corresponding bonus payment in addition to your guaranteed participation fee of \in 5.00. You will also receive the bonus payment in the form of a credit to your ecUM.

Since each of your decisions can be selected, you should carefully consider each decision.

We will now show you an example to illustrate this procedure. You only make your decisions after the example.

Page break

For a better understanding, we will now show you an example. The table in which you will enter your decisions will look like the one below.

You only make your decisions on the next page. You cannot select any options in this table.

[multiple price list without active radio buttons]

Each line of the table contains a decision to be made, which is identified by a decision number. In every decision you make, you either choose Option A or Option B by ticking the corresponding option.

120

Afterwards, one of the decision numbers is randomly selected by the computer and the corresponding decision is implemented. This means that you watch the video selected in this decision and receive the corresponding bonus payment in addition to your guaranteed participation fee of \in 5.00.

Since any decision number can be selected, you should carefully consider all decisions. (The decision numbers E1 and E11 are each selected with a probability of 27.5%, each of other numbers is selected with a probability of 5%).

We would now like you to answer a comprehension question. You cannot proceed until you have correctly answered this question. If something is still unclear to you, please raise your hand and wait for someone to come to your seat.

- 1. What happens if you choose Option B in decision E8 and decision E8 is randomly selected as the decision to be implemented?
 - In addition to your guaranteed participation fee of €5.00, you will receive a bonus payment of €1.00 and watch the 360° video about the Deutsche Bundesbank.
 - In addition to your guaranteed participation fee of €0.00, you will receive a bonus payment of €0.00 and watch the 360° video about the Deutsche Bundesbank.
 - In addition to your guaranteed participation fee of €5.00, you will receive a bonus payment of €0.00 and watch the 360° video about the Deutsche Bundesbank.
 - In addition to your guaranteed participation fee of €5.00, you will receive a bonus payment of €1.00 and watch the 360° video about intensive farming.

Page break

You answered the comprehension question correctly.

Please make your decisions now by selecting your preferred option in each row of the table.

Decision	Option A:	Option B:
Decision	You watch the 360° video about	You watch the 360° video about
number	intensive farming	the Deutsche Bundesbank
	Choose Option A and obtain	Choose Option B and obtain
E1	$ \in 0.00 $ additionally.	$\in 8.00$ additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E2	$\in 0.00$ additionally.	€5.00 additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E3	€0.00 additionally.	€3.00 additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E4	$\in 0.00$ additionally.	$\in 1.00$ additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E5	€0.00 additionally.	$\in 0.50$ additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E6	€0.00 additionally.	€0.00 additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E7	€0.50 additionally.	€0.00 additionally.
	0	
	Choose Option A and obtain	Choose Option B and obtain
E8	€1.00 additionally.	€0.00 additionally.
D e	Choose Option A and obtain	Choose Option B and obtain
E9	€3.00 additionally.	€0.00 additionally.
D10	Choose Option A and obtain	Choose Option B and obtain
EIO	€5.00 additionally.	€0.00 additionally.
D11	Choose Option A and obtain	Choose Option B and obtain
E11		
1		

End of instructions

Sie haben die Verständnisfrage richtig beantwortet.

Bitte treffen Sie nun Ihre Entscheidungen, indem Sie in jeder Zeile der Tabelle Ihre bevorzugte Option markieren.

Entscheidungs- nummer	Option A: Sie schauen das 360°-Video über die Massentierhaltung	Option B: Sie schauen das 360°-Video über die Deutsche Bundesbank
E1	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 8,00 € zusätzlich erhalten. ○
E2	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 5,00 € zusätzlich erhalten. ○
E3	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 3,00 € zusätzlich erhalten. ○
E4	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 1,00 € zusätzlich erhalten. ○
E5	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 0,50 € zusätzlich erhalten. ○
E6	Option A wählen und 0,00 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○
E7	Option A wählen und 0,50 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○
E8	Option A wählen und 1,00 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○
E9	Option A wählen und 3,00 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○
E10	Option A wählen und 5,00 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○
E11	Option A wählen und 8,00 € zusätzlich erhalten. ○	Option B wählen und 0,00 € zusätzlich erhalten. ○

Figure A.4: Screenshot Multiple Price List

Notes: The figure shows a screenshot of the decision screen that was used to elicit subjects' willingness-to-pay for information on the living conditions of pigs in intensive farming.

A.7 Video Content

Minute	Original	Description	Narrator (translated)
windle	Video	Description	
0.00 - 0.29	0.32 - 1.01	Fully-grown pigs kept individually:	You don't know what you're in for but you're
0.00 0.20	0.02 1.01	Housing facilities are separated by	behind bars. Looking around, you see count-
		bars: Slatted floor (approx. 70%) and	less others like you. Your neighbor on the left
		a connected area in front of the feeding	tries to seek comfort from you. It has been this
		trough	way as long as you can remember.
0.29 - 0.48	1.30 - 1.49	Sow with piglets in farrowing pen:	Many of your siblings won't even survive a
0.20 0.10	1.00 1.10	Slatted floor: Stillbirths/dead piglets	week in that breeding ground for disease
		on the ground: Small lying area with-	week in that breeding ground for disease.
		out litter available	
0:49-1:08	1:50-2:09	Castration of piglet that is conscious	If you do survive that first week, you're en-
			forced to endure a tooth clipping, tail docking,
			and castration. All without any pain relief.
1:09-1:29	2:10-2:30	Piglets are returned to the cage; Crate	
		stall; Slatted floor;	
1:30-2:10	2:31 - 3:11	Crate stall; Slatted floor; Lying area;	Without proper medical care, you return to
		Area with litter	the cage where you were born. A cell that's
			anything but sanitary. Here, you'll never once
			take a single breath of fresh air, never taste
			fresh water, never even see the sun. This pun-
			ishment is worse than that faced by even the
			most heinous offenders in prison. Yet, you're
			guilty of being born and even that was never
			a choice.
2:11-2:35	3:12-3:36	Sows in individual housing; Bar biting	If you happen to be born a girl, you'll likely be
			cycled back through the gauntlet of cages. Ar-
			tificially inseminated and then robbed of your
			children over and over and over again for as
			long as you remain fertile. The same fate
2.26 2.06	2.27 4.07	Pige that are unquitable for breading on	awaits most daughters you will ever have.
2.30-3.00	5.57-4.07	male kept in groups (fattening). More	And if you're deemed unit for breeding of if
		than 20 animals: Slatted floor: Car tire	taken to a fattening farm. Here you live in
		to stay busy: Injured animals	cramped quarters for the next five months
		to stay busy, injured animals	again without any stimulation no fresh air or
			water. The boredom and frustration may drive
			you mad and if not you your siblings In-
			evitably some act out biting and hurting each
			other.
3:07-3:58	4:33-5:24	Slaughterhouse: Electrical stunning of	Even if you survive the fattening farm. vou'll
		the pigs with stunning forceps; Pigs are	soon learn that those first six months of your
		hung up by one leg after being anes-	life will be your last. Your next and final
		thetized; Bleeding the pigs by pierc-	stop is the slaughterhouse. Here, huddled to-
		ing the center of the neck once; Pad-	gether with all the other pigs, you see the bod-
		dling movement of pigs after anesthe-	ies of those that came here moments before,
		sia (typical behavior after anesthesia)	now strung up and bleeding. You watch help-
			lessly as the others are picked off one by one,
			stunned with an electrical current just enough
			to be chained and hoisted up by one leg, until
			finally, it's your turn.
3:59-4:49	5:40-6:30	Bleeding the pigs by piercing the cen-	
		ter of the neck once; Pig with severe	
		cramps falls from the holding device	

Table A.12: Video Content

Notes: In total, the video is 5 minutes long as it additionally shows a black screen for about 11 seconds at the beginning. The black screen was shown for the same duration in the video about the German central bank.

A Appendix to Chapter 1

Minute	Legal situation (translated and condensed)
0:00-0:29	§ 22 Allgemeine Anforderungen an Haltungseinrichtungen für Schweine (TierSchNutztV, Ab-
	schnitt 5 Anforderungen an das Halten von Schweinen)
	Housing facilities need to guarantee that individually held pigs can see the other pigs and all
	pigs are able to lay down, stand up, and adopt a natural position.
	§ 24 Besondere Anforderungen an Haltungseinrichtungen für Jungsauen und Sauen (Tier-
	SchNutztV, Abschnitt 5 Anforderungen an das Halten von Schweinen)
	The lying area of individually hold gilts and sows should at most be partially perforated. Crate
	stalls need to prevent that pigs hurt themselves and guarantee that all pigs can stand up, lay
	down and, stretch out.
0:29-0:48	§ 23 Besondere Anforderungen an Haltungseinrichtungen für Saugferkel (TierSchNutztV, Ab-
	schnitt 5 Anforderungen an das Halten von Schweinen)
	The living area of the piglets need to have safety devices so no piglet gets squashed. Additionally
	it need to be guaranteed that all can suck and relax at the same time. The lying area either
	needs to be thermally insulated and heatable or covered by litter. Perforated floor in the lying
	area must be covered.
	§ 24 Besondere Anforderungen an Haltungseinrichtungen für Jungsauen und Sauen (Tier-
	SchNutztV, Abschnitt 5 Anforderungen an das Halten von Schweinen)
	Crate stalls need to prevent that pigs hurt themselves and guarantee that all pigs can stand up,
	lay down and, stretch out.
0:49-1:08	Richtlinie 2008/120/EG des Rates vom 18. Dezember 2008 über Mindestanforderungen für den
	Schutz von Schweinen (kodifizierte Fassung) (ABl. L 47 vom 18.2.2009, S. 5-13)
	Veterinarians are allowed to shrink the canines of piglets, crop the tail, castrate male pigs, and
	attach nose rings. Cropping the tail or shrinking the canines is only allowed if it is shown that
	the sow's teats or the ears of other pigs have been injured. However, before doing such painful
	interventions other measures need to be taken to prevent behavioral disorders, such as adapting
	unsuitable accommodations.
1:09-1:29	See above.
1:30-2:10	See above.
2:11-2:35	§ 30 Besondere Anforderungen an das Halten von Jungsauen und Sauen (TierSchNutztV, Ab-
	schnitt 5 Anforderungen an das Halten von Schweinen)
	Gilts and sows are supposed to be hold in a group until one week before giving birth. The living
	area here should offer at least 1.85sqm per gilt and 2.50sqm per sow. Gilts or sows that are hold
	individually need to be able to turn around unimpededly during that time. Gilts and sows are
	only allowed to be hold in crate stalls when it is not obvious that this husbandry condition leads
	to negative arousal.
2:36-3:06	§ 26 Allgemeine Anforderungen an das Halten von Schweinen (TierSchNutztV, Abschnitt 5 An-
	forderungen an das Halten von Schweinen)
	Every pig should have access to enough toys that are harmless for health to maintain their ex-
	ploratory behavior.
	§ 29 Besondere Anjoraerungen an aas Halten von Zuchtlaufern und Mastschweinen (Tier-
	SchNutztV, Abschnitt 5 Anforderungen an das Halten von Schweinen)
	Breeding runners and fattening pigs should be hold in groups without frequent changes in the
	composition.
	Richtlinie 2008/120/EG des Rates vom 18. Dezember 2008 über Mindestanforderungen für den
	Schutz von Schweinen (kodifizierte Fassung) (ABl. L 47 vom 18.2.2009, S. 5-13)
	Aggressive as well as injured animals need to be separated from the rest of the group. Sick or
	injured pigs are held individually.
3:07-3:58	§ 12 Betäuben, Schlachten und Töten (Tierschutz-Schlachtverordnung – TierSchlV. Abschnitt 1
	Vorschriften über das Ruhigstellen, Betäuben. Schlachten und Töten von Tieren)
	When slaughtering an animal, the bleeding needs to be started within 20 seconds after the anaes-
	thesia, while the animal still cannot feel anything. Additionally, the owner of the slaughterhouse
	needs to guarantee that if it is necessary the animal is bled by hand.
	veroranung (EG) Nr. 1099/2009 des Rates vom 24. September 2009 über den Schutz von
	Tieren zum Zeitpunkt der Totung
	Animals are only getting killed after being stunned. This state of numbress and unconsciousness
	needs to hold on until the animal is dead. It is not allowed to immobilize an animal by hanging
	it up before it got stunned.
	Using head-only electronic stunning for pigs demands a minimum amperage from 1.3A. The
	whole procedure of stunning, hanging and bleeding an animal needs to be finished before start-
	ing with the next one. Further steps are allowed after checking whether the animal really is
	dead.
3:59-4:49	See above.

Table A.13: Legal Situation

Additional Tables A.8

Table A.14: Comparison by Whether Subjects Purchased Meal after Experiment

	Purchased meal after experiment G		Comparison	
Variable	Yes	No	p-value	
WTP for information on intensive farming $(euro)^a$	0.39 (0.11)	0.01 (0.28)	0.21	
Age (years)	21.95 (0.19)	22.22 (0.51)	0.62	
Female (1:yes, 0:no)	$\begin{array}{c} 0.43 \\ (0.03) \end{array}$	$\begin{array}{c} 0.61 \\ (0.07) \end{array}$	0.01	
Bachelor's degree or higher (1:yes, 0:no)	$\begin{array}{c} 0.28 \\ (0.03) \end{array}$	$\begin{array}{c} 0.35 \ (0.07) \end{array}$	0.31	
Location: Mannheim (1:yes, 0:no)	$\begin{array}{c} 0.64 \\ (0.03) \end{array}$	$\begin{array}{c} 0.48 \\ (0.07) \end{array}$	0.03	
Meat-eating habit (stated frequency)				
Never	$\begin{array}{c} 0.14 \\ (0.02) \end{array}$	$\begin{array}{c} 0.07 \\ (0.04) \end{array}$	0.13	
Several times per year	$\begin{array}{c} 0.10 \\ (0.02) \end{array}$	$\begin{array}{c} 0.17 \\ (0.05) \end{array}$	0.21	
Several times per month	$\begin{array}{c} 0.25 \\ (0.03) \end{array}$	$\begin{array}{c} 0.30 \\ (0.06) \end{array}$	0.47	
Several times per week	$\begin{array}{c} 0.38 \\ (0.03) \end{array}$	$\begin{array}{c} 0.37 \\ (0.07) \end{array}$	0.85	
Daily	$\begin{array}{c} 0.13 \\ (0.02) \end{array}$	$\begin{array}{c} 0.09 \\ (0.04) \end{array}$	0.36	
Received information (1:yes, 0:no)	$\begin{array}{c} 0.53 \\ (0.03) \end{array}$	$\begin{array}{c} 0.50 \\ (0.07) \end{array}$	0.70	
Number of purchases before experiment	$21.79 \\ (1.05)$	$5.69 \\ (2.09)$	0.00	
Observation period before experiment (days)	$ \begin{array}{c} 126.40 \\ (2.37) \end{array} $	$143.65 \\ (6.66)$	0.02	
Observation period after experiment (days)	$ \begin{array}{l} 80.65 \\ (2.37) \end{array} $	$53.76 \\ (4.35)$	0.00	
Share of meals with meat before $\operatorname{experiment}^{b}$	$\begin{array}{c} 0.42 \\ (0.02) \end{array}$	$\begin{array}{c} 0.36 \ (0.07) \end{array}$	0.37	
Share of meals without meat before $\operatorname{experiment}^b$	$\begin{array}{c} 0.43 \\ (0.02) \end{array}$	$0.48 \\ (0.07)$	0.53	
Share of ambiguous meals before experiment ^{b}	$0.15 \\ (0.01)$	0.17 (0.06)	0.79	
Observations	276	54		

Notes: Standard errors are in parentheses. p-values are obtained from regressing the corresponding variable on a dummy for purchasing at least one food item in the post-experimental period, using robust standard errors. ^a The WTP for information is based on the midpoint of the corresponding WTP interval and

only subjects with a WTP that is bounded by the prices in the multiple price list are considered (261 in the first and 51 in the second group) ^bThese values are only available for subjects with a pre-experimental purchase (263 in the first

and 29 in the second group).

A Appendix to Chapter 1

	Before w	reighting	After we	eighting	Comparison
Variable	No info	Info	No info	Info	<i>p</i> -value
WTP for information on int. farm. (euro)	-0.07 (0.15)	0.69 (0.14)	$0.32 \\ (0.10)$	0.32 (0.10)	1.00
Age (years)	22.06 (0.27)	21.92 (0.26)	22.18 (0.29)	21.82 (0.25)	0.33
Female (1:yes, 0:no)	$0.45 \\ (0.04)$	$0.45 \\ (0.04)$	$0.42 \\ (0.04)$	$\begin{array}{c} 0.47 \\ (0.04) \end{array}$	0.34
Bachelor's degree or higher (1:yes, 0:no)	$\begin{array}{c} 0.33 \ (0.04) \end{array}$	$\begin{array}{c} 0.25 \\ (0.03) \end{array}$	$\begin{array}{c} 0.35 \ (0.04) \end{array}$	$0.24 \\ (0.03)$	0.02
Location: Mannheim (1:yes, 0:no)	$0.64 \\ (0.04)$	$0.63 \\ (0.04)$	$0.63 \\ (0.04)$	$0.62 \\ (0.04)$	0.91
Meat-eating habit (stated frequency)					
Never	$0.11 \\ (0.03)$	$\begin{array}{c} 0.15 \ (0.03) \end{array}$	$0.09 \\ (0.02)$	$\begin{array}{c} 0.16 \\ (0.03) \end{array}$	0.05
Several times per year	$\begin{array}{c} 0.07 \\ (0.02) \end{array}$	$0.14 \\ (0.03)$	$\begin{array}{c} 0.07 \\ (0.02) \end{array}$	$\begin{array}{c} 0.13 \ (0.03) \end{array}$	0.07
Several times per month	$0.30 \\ (0.04)$	0.22 (0.03)	0.29 (0.04)	0.24 (0.04)	0.32
Several times per week	0.41 (0.04)	$0.35 \\ (0.04)$	0.45 (0.04)	0.33 (0.04)	0.03
Daily	0.11 (0.03)	0.14 (0.03)	0.10 (0.02)	0.14 (0.03)	0.34
Number of purchases before exp.	20.03 (1.53)	18.95 (1.38)	19.97 (1.58)	$18.95 \\ (1.37)$	0.63
Observation period before exp. (days)	130.24 (3.42)	128.20 (3.23)	129.62 (3.40)	127.58 (3.12)	0.66
Observation period after exp. (days)	76.27 (3.25)	77.81 (3.13)	76.46 (3.38)	78.11 (3.12)	0.72
Share of meals with meat before \exp^{a}	0.43 (0.03)	0.40 (0.03)	0.44 (0.03)	0.38 (0.02)	0.08
Share of meals without meat before \exp^{a}	0.40 (0.03)	0.46 (0.03)	0.40 (0.02)	0.47 (0.02)	0.05
Share of ambiguous meals before $\exp.^a$	0.17 (0.02)	0.14 (0.02)	0.16 (0.02)	0.15 (0.02)	0.75
Observations	151	161	151	161	

 Table A.15:
 Comparison of Treatment Groups

Notes: Standard errors are in parentheses. Comparisons between the treatment and control group are based

on the IPW estimator applied to the corresponding variable of interest. ^aThese values are only available for subjects who purchased a meal before the experiment (135 and 141, respectively).

	(1)	(2)	(3)
Demographics			
Age (years)	0.02	0.02	0.03
	(0.03)	(0.03)	(0.03)
Female (1:yes, 0:no)	-1.11***	-0.75***	-0.75***
	(0.24)	(0.24)	(0.23)
Bachelor's degree or higher (1:yes, 0:no)	-0.04	0.00	0.10
	(0.25)	(0.25)	(0.23)
Location: Mannheim (1:yes, 0:no)	0.04	0.05	0.08
	(0.22)	(0.21)	(0.20)
Meat-eating habit (0:never, 4:daily)	-0.20*	-0.23**	0.01
	(0.11)	(0.11)	(0.13)
Videos and virtual reality			
Own VR glasses (1:yes, 0:no)		-0.46**	-0.24
		(0.22)	(0.30)
Used VR glasses to watch a 360° video (1:yes, 0:no)		0.32	0.25
		(0.20)	(0.19)
Tolerance of violence in videos (1:very high, 5:very low)		-0.37***	-0.40***
		(0.10)	(0.10)
Related to intensive farming		. ,	. ,
Evaluation of pigs' living conditions (1:very good, 5:very bad)			-0.11
			(0.23)
Evaluation of chickens' living conditions (1:very good, 5:very bad)			-0.07
			(0.22)
Interest in pigs' living conditions (1:very much, 5:not at all)			0.16
			(0.10)
Relevance of pigs' living conditions (1:very relevant, 5:not at all)			-0.44***
			(0.12)
Informed about pigs' living conditions (1:very well, 5:not at all)			0.10
			(0.14)
Seen pigs' living conditions somewhere			0.61^{**}
10 0			(0.28)
Pigs' capacity to feel pain (0:no capacity, 10:like humans)			-0.01
$\mathbf{S}_{\mathbf{r}} = \mathbf{r}_{\mathbf{r}} + $			(0.04)
Related to meat consumption			
Dishes with meat taste good (1:totally agree, 5:totally disagree)			-0.04
			(0.11)
Dishes w/o meat taste good (1:totally agree, 5:totally disagree)			-0.14
,			(0.12)
Meat is important for health (1:totally agree, 5:totally disagree)			0.14
			(0.10)
Eating meat is normal (1:totally agree, 5:totally disagree)			0.16
mout is normal (instanty agros, showary abagito)			(0.10)
Friends would show understanding for decision to not eat meat			0.08
(1:totally agree 5:totally disagree)			(0.13)
(Inorany asico, onorany disagree)			(0.10)
R^2	0.08	0.13	0.20
Observations	312	312	312

 Table A.16:
 Correlation of WTP and Observable Characteristics

Notes: Robust standard errors are in parentheses. Coefficient estimates are based on OLS regressions with the midpoint of the WTP interval as dependent variable. Only subjects whose WTP is bounded in the multiple price list are included. For variables which are elicited on a likert scale, we only provide the start and end of the scale in parentheses after the variable name.

A Appendix to Chapter 1

	IPW	W	WLS		IPW	
	(1)	(2)	(3)	(4)	(5)	
A. Any meal purchase						
Info effect	$0.006 \\ (0.041)$	$0.008 \\ (0.044)$	$0.007 \\ (0.041)$	$\begin{array}{c} 0.010 \\ (0.036) \end{array}$	$0.009 \\ (0.036)$	
Mean (w/o info)	0.837	0.840	0.841	0.840	0.841	
B. Meal purchases per day						
Info effect	-0.004 (0.019)	-0.003 (0.019)	-0.002 (0.016)	-0.003 (0.018)	-0.002 (0.015)	
Mean (w/o info)	0.158	0.159	0.158	0.159	0.159	
Baseline level	No	Yes	Yes	Yes	Yes	
Additional controls	No	No	Yes	No	Yes	
Observations	312	312	312	312	312	

Table A.17: Likelihood to Purchase Meal in the Field

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. In Panel A, the dependent variable is whether an individual purchased a meal after the experiment. Logit regressions are used to predict outcome levels for the AIPW estimator (columns 4 and 5). In Panel B, the dependent variable is the number of meal purchases per day after the experiment (not counting Sundays). OLS regressions are used to predict outcome levels for the AIPW estimator (columns 4 and 5).

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

Table A.18:	Information	Effect	in the	Field	Controlling	for	Meat	Eating	Habit

	W	'LS	AIPW		
	(1)	(2)	(3)	(4)	
A. Share of meals with meat					
Info effect	-0.060*	-0.067**	-0.062*	-0.069**	
	(0.034)	(0.033)	(0.033)	(0.033)	
Mean (w/o info)	0.447	0.451	0.447	0.449	
B. Share of meals without meat					
Info effect	0.091^{**}	0.098^{**}	0.094^{**}	0.100^{***}	
	(0.044)	(0.042)	(0.037)	(0.037)	
Mean $(w/o info)$	0.403	0.399	0.403	0.402	
Meat-eating habit	Yes	Yes	Yes	Yes	
Additional controls	No	Yes	No	Yes	
Observations	261	261	261	261	

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit, we use indicator variables. Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.01

 Table A.19:
 Information Effect in the Laboratory with Restricted Sample

	Dependent variable: Choosing voucher for meal with meat				
	IPW	WLS		AIPW	
	(1)	(2)	(3)	(4)	(5)
Info effect	-0.193^{***} (0.059)	-0.124^{**} (0.057)	-0.114^{**} (0.051)	-0.127^{**} (0.050)	-0.113^{**} (0.048)
Mean (w/o info)	0.477	0.447	0.441	0.447	0.445
Baseline level from field Additional controls	No No	Yes No	Yes Yes	Yes No	Yes Yes
Observations	261	261	261	261	261

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). The share of purchased meals containing meat before the experiment is used as baseline level. Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.
	IPW	W	LS
	(1)	(2)	(3)
A. Share of meals with meat			
Avoider			
Info effect	-0.153^{**}	-0.006	-0.010
	(0.076)	(0.061)	(0.062)
Mean $(w/o info)$	0.482	0.423	0.425
Seeker			
Info effect	-0.112**	-0.087**	-0.089**
	(0.050)	(0.038)	(0.037)
Mean (w/o info)	0.479	0.466	0.467
Comparison of effects $(p-value)$	0.65	0.26	0.28
B. Share of meals without meat			
Avoider			
Info effect	0.249^{***}	0.073	0.073
	(0.071)	(0.066)	(0.065)
Mean $(w/o info)$	0.331	0.400	0.400
Seeker			
Info effect	0.103**	0.095^{*}	0.095^{*}
	(0.051)	(0.055)	(0.049)
Mean $(w/o info)$	0.393	0.399	0.399
Comparison of effects $(p$ -value)	0.09	0.79	0.79
Baseline level	No	Yes	Yes
Additional controls	No	No	Yes
Observations	261	261	261

Table A.20: Information Effect in the Field – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean $(w/o \ info)$ corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	Dependent variable: Choosing voucher for meal with meat					
	IPW	W	' LS			
	(1)	(2)	(3)			
Avoider						
Info effect	-0.340^{***} (0.109)	-0.174^{*} (0.100)	-0.157^{*} (0.093)			
Mean $(w/o info)$	0.597	0.530	0.517			
Seeker						
Info effect	-0.131^{*} (0.070)	-0.103 (0.065)	-0.097 (0.060)			
Mean (w/o info)	0.426	0.412	0.409			
Comparison of effects $(p-value)$	0.11	0.55	0.58			
Baseline level from field Additional controls Observations	No No 261	Yes No 261	Yes Yes 261			

Table A.21: Information Effect in the Laboratory with Restricted Sample – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). The share of purchased meals containing meat before the experiment is used as baseline level. Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.

IPW	W	LS	AIPW	
(1)	(2)	(3)	(4)	(5)

Table A.22: Information Effect in the Field – Meat from Terrestrial and Aquatic

from terrestrial animals					
Info effect	-0.076**	-0.022	-0.022	-0.022	-0.024
	(0.038)	(0.030)	(0.030)	(0.030)	(0.030)
Mean (w/o info)	0.372	0.348	0.348	0.351	0.350
B. Share of meals without meat from terrestrial animals					
Info effect	0.099^{**}	0.048	0.045	0.046	0.043
	(0.040)	(0.042)	(0.039)	(0.034)	(0.034)
Mean (w/o info)	0.479	0.500	0.502	0.498	0.503
C. Share of meals with meat from aquatic animals					
Info effect	-0.047**	-0.045**	-0.047**	-0.046**	-0.053***
	(0.021)	(0.021)	(0.021)	(0.020)	(0.020)
Mean (w/o info)	0.105	0.104	0.105	0.105	0.109
D. Share of meals without meat					
from aquatic animals					
Info effect	0.033	0.043	0.043	0.045^{*}	0.044^{*}
	(0.030)	(0.033)	(0.031)	(0.026)	(0.025)
Mean (w/o info)	0.822	0.814	0.814	0.814	0.816
Baseline level	No	Yes	Yes	Yes	Yes
Additional controls	No	No	Yes	No	Yes
Observations	261	261	261	261	261

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. $^{*}p<0.1,\,^{**}p<0.05,\,^{***}p<0.01.$

	IPW	W	LS	AI	PW
	(1)	(2)	(3)	(4)	(5)
A. Share of meals with pork					
Info effect	-0.035	-0.021	-0.020	-0.026	-0.027
	(0.026)	(0.025)	(0.025)	(0.025)	(0.025)
Mean (w/o info)	0.186	0.179	0.179	0.182	0.182
B. Share of meals without pork					
Info effect	0.030	0.012	0.009	0.017	0.015
	(0.037)	(0.036)	(0.035)	(0.033)	(0.032)
Mean (w/o info)	0.710	0.716	0.717	0.713	0.717
C. Share of meals with meat					
	0.000**	0.020	0.020	0.091	0.005
Into effect	-0.000^{+1}	-0.032	-0.030	-0.031	-0.025
	(0.029)	(0.031)	(0.030)	(0.027)	(0.027)
Mean (w/o inio)	0.229	0.217	0.216	0.219	0.216
D. Share of meals without meat from other terrestrial animals					
Info effect	0.063^{*}	0.032	0.027	0.030	0.025
	(0.034)	(0.036)	(0.033)	(0.033)	(0.032)
Mean (w/o info)	0.674	0.688	0.691	0.683	0.687
Baseline level	No	Yes	Yes	Yes	Yes
Additional controls	No	No	Yes	No	Yes
Observations	261	261	261	261	261

Table A.23: Information Effect in the Field – Pork and Meat from Other Terrestrial Animals

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.

	Considered days after the experiment:					
	0-3	0-7	> 7	all		
A. Share of meals with meat						
Info effect	-0.173**	-0.118**	-0.033	-0.066**		
	(0.072)	(0.049)	(0.034)	(0.032)		
Mean (w/o info)	0.483	0.451	0.434	0.455		
B. Share of meals without meat						
Info effect	0.178***	0.113**	0.068	0.089**		
	(0.067)	(0.052)	(0.043)	(0.041)		
Mean (w/o info)	0.366	0.400	0.415	0.399		
Baseline level	Yes	Yes	Yes	Yes		
Additional controls	Yes	Yes	Yes	Yes		
Observations	133	199	248	261		

Table A.24: Information Effect in the Field over Time

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. All estimates are based on the WLS estimator. To estimate propensity scores for the period of days 0–3, we merge WTP groups only consisting of treated or untreated subjects with neighboring WTP groups.

*p < 0.1, **p < 0.05, ***p < 0.01.

	Conside	red days a	after the ex	xperiment:
	0-3	0-7	> 7	all
A. Any purchase				
Info effect	-0.041	-0.053	-0.006	0.007
	(0.064)	(0.057)	(0.045)	(0.041)
Mean (w/o info)	0.457	0.674	0.807	0.841
B. Purchases per day				
Info effect	-0.012	-0.019	0.002	-0.002
	(0.031)	(0.025)	(0.016)	(0.016)
Mean (w/o info)	0.209	0.264	0.142	0.158
Baseline level	Yes	Yes	Yes	Yes
Additional controls	Yes	Yes	Yes	Yes
Observations	312	312	312	312

Table A.25: Likelihood to Purchase Food over Time

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. All estimates are based on the WLS estimator. In Panel A, the dependent variable is whether an individual purchased a meal after the experiment. In Panel B, the dependent variable is the number of meal purchases per day after the experiment (not counting Sundays). *p < 0.1, **p < 0.05, ***p < 0.01.

	Considered days after the experiment:				
	0-3	0-7	>7	all	
A. Share of meals with meat					
Info effect	-0.153^{**} (0.078)	-0.132^{**} (0.051)	$\begin{array}{c} 0.005 \ (0.039) \end{array}$	-0.027 (0.034)	
Mean (w/o info)	0.471	0.449	0.415	0.428	
B. Share of meals without meat					
Info effect	0.163^{**} (0.073)	0.119^{**} (0.054)	$0.025 \\ (0.041)$	$\begin{array}{c} 0.040 \\ (0.037) \end{array}$	
Mean (w/o info)	0.381	0.401	0.452	0.438	
Baseline level Additional controls Observations	Yes Yes 127	Yes Yes 186	Yes Yes 186	Yes Yes 186	

Table A.26: Information Effect in the Field over Time for Constant Sample

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results).Additional controls are age, gender, degree, location, and WTP. All estimates are based on the WLS estimator. Only subjects who purchase a meal in the first week after the experiment as well as in the subsequent period (week 2 and following) are considered.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	Dependent variable: Choosing voucher for meal with meat				
	IPW	W	LS		
	(1)	(2)	(3)		
WTP < 0.5					
Info effect	-0.206^{***} (0.067)	-0.164^{***} (0.056)	-0.168^{***} (0.055)		
Mean $(w/o info)$	0.481	0.463	0.462		
$WTP \ge 0.5$					
Info effect	-0.055 (0.096)	-0.009 (0.107)	-0.011 (0.102)		
Mean (w/o info)	0.387	0.359	0.361		
Comparison of effects (<i>p</i> -value)	0.19	0.20	0.18		
Meat-eating habit	No	Yes	Yes		
Additional controls	No	No	Yes		
Observations	312	312	312		

Table A.27: Information Effect in the Laboratory – Low and High WTP

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). To control for the meat-eating habit in columns 2 and 3, we use indicator variables. Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	IPW	W	LS
	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	(3)	
A. Share of meals with meat			
WTP < 0.5			
Info effect	-0.125**	-0.048	-0.052
	(0.050)	(0.040)	(0.040)
Mean $(w/o info)$	0.451	0.419	0.420
$WTP \ge 0.5$			
Info effect	-0.122	-0.094*	-0.093
	(0.075)	(0.056)	(0.056)
Mean (w/o info)	0.538	0.523	0.523
Comparison of effects $(p-value)$	0.97	0.50	0.56
B. Share of meals without meat			
WTP < 0.5			
Info effect	0.167^{***}	0.091^{**}	0.093**
	(0.049)	(0.042)	(0.041)
Mean (w/o info)	0.394	0.423	0.422
$WTP \ge 0.5$			
Info effect	0.107	0.086	0.080
	(0.077)	(0.097)	(0.090)
Mean (w/o info)	0.336	0.350	0.354
Comparison of effects $(p-value)$	0.51	0.97	0.89
Baseline level	No	Yes	Yes
Additional controls	No	No	Yes
Observations	261	261	261

Table A.28: Information Effect in the Field – Low and High WTP

Notes: Standard errors are in parentheses. Mean $(w/o \ info)$ corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP.

p < 0.1, p < 0.05, p < 0.05, p < 0.01.

	IPW	W	LS
	(1)	(2)	(3)
A. Share of meals with meat from terrestrial animals			
Avoider			
Info effect	-0.088 (0.065)	0.032 (0.054)	0.034 (0.054)
Mean (w/o info)	0.336	0.289	0.287
Seeker			
Info effect	-0.070 (0.046)	-0.044 (0.036)	-0.045 (0.036)
Mean (w/o info)	0.387	0.374	0.374
Comparison of effects $(p-value)$	0.82	0.24	0.23
B. Share of meals without meat from terrestrial animals			
Avoider			
Info effect	0.187^{***} (0.068)	0.037 (0.065)	0.031 (0.062)
Mean (w/o info)	0.474	0.531	0.535
Seeker			
Info effect	$\begin{array}{c} 0.061 \\ (0.050) \end{array}$	$\begin{array}{c} 0.053 \\ (0.052) \end{array}$	$\begin{array}{c} 0.051 \\ (0.047) \end{array}$
Mean $(w/o info)$	0.481	0.487	0.488
Comparison of effects $(p-value)$	0.14	0.85	0.79
Baseline level	No	Yes	Yes
Additional controls	No	No	Yes
Observations	261	261	261

Table A.29: Information Effect in the Field for Meat from Terrestrial Animals – Avoider and Seeker

Notes: Standard errors are in parentheses. Mean (w/o info) corresponds to the counterfactual mean outcome in the absence of information (based on the estimation results). Additional controls are age, gender, degree, location, and WTP. *p < 0.1, **p < 0.05, ***p < 0.01.

A.9 Additional Figures





Notes: The figure shows the distribution of prices for meals. The dashed line represents the average price. The minimum and maximum prices are 0.35 and 22.42 euro, respectively.



Figure A.6: Treatment Status by WTP for Information

Notes: The figure shows the allocation of subjects with a given WTP into the different treatment groups.



Figure A.7: Observations in Moving Time Window

Notes: The figure shows the number of subjects who purchase at least one meal in the time window of seven days, with the day on the x-axis as the midpoint $(\pm 3 \text{ days})$. Day 0 represents the day of the experimental session. For the dotted line, the time windows become shorter around zero since they only capture observations before or after the experiment.



Figure A.8: Effect of Information over Time

Notes: The figure shows the average effect of receiving information on the propensity to eat a meal with meat (Panel A) and to eat a meal without meat (Panel B) over time. Day 0 represents the day of the experimental session. The effect is estimated based on a moving time window of seven days with the day on the x-axis as the midpoint (± 3 days). Around day 0, the time window captures only those days before or after the experiment. We use the WLS specification and control for covariates and the meat consumption before day -14 (missing values are imputed as described in Section 1.4.3). To estimate propensity scores, we merge WTP groups only consisting of treated or untreated subjects with neighboring WTP groups.





Notes: The figure shows the 7-day moving average level of meat consumption for subjects that do and those that do not receive information, with the day on the x-axis as midpoint $(\pm 3 \text{ days})$. Day 0 represents the day of the experimental session. Around day 0, the time window captures only those days before or after the experiment. At day 0, we only show the values for the time window that focuses on the post-experimental period. All values are normalized based on the week before the experiment.



Figure A.10: Likelihood to Purchase Food over Time

Notes: The figure shows the average effect of receiving information on the propensity to purchase a meal (Panel A) and the number of purchases per day (Panel B) over time. Day 0 represents the day of the experimental session. The effect is estimated based on a moving time window of seven days with the day on the x-axis as the midpoint (± 3 days). Around day 0, the time window captures only those days before or after the experiment. We use the WLS specification and control for covariates and the meat consumption before day -14 (missing values are imputed as described in Section 1.4.3). To estimate propensity scores, we merge WTP groups only consisting of treated or untreated subjects with neighboring WTP groups.



Notes: The figure shows the information effects for individuals who do and those that do not select into information at a given cost (Panel A and Panel B, respectively). The information effects are based on the WLS estimator with choosing the meat voucher as dependent variable and controlling for the stated meat-eating

habit. Only individuals whose WTP for information is bounded by -8 and 8 euro are considered. Whiskers show standard errors. When the cost is high (low), the estimated information effect for individuals who (do not) select into information is only based on few observation but is shown nevertheless for illustrative purposes.

Panel A. Propensity to consume meat



Panel B. Propensity to not consume meat



Panel C. Information acquisition



Figure A.12: Effect of Information Provision on Meat Consumption in the Field

Notes: Panel A and Panel B show the average effect of making information available at a given cost. The outcome variable is the share of purchased meals with meat and the share of meals without meat, respectively. The effect at a given cost c is calculated as $ATE(WTP \ge c) \frac{1}{N} \sum_{i=1}^{N} \mathbb{1}\{WTP_i \ge c\}$, where WTP_i denotes the willingness-to-pay of subject i, $ATE(WTP \ge c)$ denotes the average treatment effect of receiving information for subjects with a willingness-to-pay for information at least as large as the cost, $\mathbb{1}\{\cdot\}$ denotes the indicator function, and N denotes the number of subjects. Estimates of $ATE(WTP \ge c)$ are based on the WLS estimator, allowing for heterogeneous information effects based on the selection into information at a given cost and controlling for baseline levels. Panel C shows the fraction of subjects acquiring information at a given cost, $\frac{1}{N} \sum_{i=1}^{N} \mathbb{1}\{WTP_i \ge c\}$. Only individuals whose WTP for information is bounded by -8 and 8 euro are considered.

5 8



Panel A. Effects on those who acquire info

0

-0.2

-0.4

Panel B. Effects on those who don't acquire info

Effect on propensity to choose meal without meat (pp) -0.2 -0.4 -0.6 -3 -1 -0.5 0 0.5 1 -5 3 5 8 -8 -5 -3 -1-0.5 0 0.5 1 -8 3 5 8 Cost of information (euro) Cost of information (euro)

Figure A.13: Information Effects by Selection in the Field

Notes: The figure shows the information effects for individuals who do and those that do not select into information at a given cost (left and right panels, respectively). All estimated information effects are based on the WLS estimator controlling for the baseline level before the experiment. In the top panels, the outcome variable is the share of meals with meat. In the bottom panels, the outcome variable is the share of meals without meat. In all cases, only individuals whose WTP for information is bounded by -8 and 8 euro are considered. Whiskers show standard errors. When the cost is high (low), the estimated information effect for individuals who (do not) select into information is only based on few observation but is shown nevertheless for illustrative purposes.

B Appendix to Chapter 2

B.1 Wording of Donation Appeal

Part I:

As part of this survey, each participant will have the **opportunity to support the provision of nutritious food for malnourished children** in the African country of South Sudan.

[Picture of malnourished child]

Decades of civil war have **devastated South Sudan** and **many children are severely malnourished**. The rations of the nutritious food are supplied by Sign of Hope, an accredited relief organization from Germany cooperating with a hospital in South Sudan.

Sign of Hope holds the **certificate for responsible and transparent use of collected donations** ("Spendenzertifikat") **awarded by the "Deutscher Spendenrat**", a German umbrella association for charitable organizations. 86 cents of every dollar they receive go directly into the relief efforts, while the remaining fourteen cents cover their overheads. Learn more about the organization at http://www.sign-ofhope.org.

[Picture of malnourished child]

Part II – Money Donation:

The nutritious food consists of a **specially developed paste** and **energy-rich biscuits** that help children gain weight.

In this survey, you may donate all, part, or none of your reward of \$7.00 for this MTurk HIT to Sign of Hope for providing the nutritious food. Thus, you may choose any amount from \$0 to \$7.00. The amount you choose will be subtracted from your reward.

Please indicate how much money you wish to donate below:

[Numeric field to indicate dollar amount]

Part II – Unit Donation – Large:

One nutritional ration, which feeds one malnourished child for one week, can be provided by the charity for a donation of \$3.50. The nutritious food consists of a specially developed paste and energy-rich biscuits that help children gain weight.

In this survey, you may use all, part, or none of your reward of \$7.00 for this MTurk HIT to provide these nutritional rations. Thus, you may choose a number from 0 to 2 rations. \$3.50 per ration will be sub-tracted from your reward.

Please indicate how many rations you wish to provide below:

[Numeric field to indicate number of rations, restricted to 0, 1, and 2]

<u>Part II – Unit Donation – Small:</u>

One nutritional ration, which feeds one malnourished child for one day, can be provided by the charity for a donation of \$0.50. The nutritious food consists of a specially developed paste and energy-rich biscuits that help children gain weight.

In this survey, you may use all, part, or none of your reward of \$7.00 for this MTurk HIT to provide these nutritional rations. Thus, you may choose a number from 0 to 14 rations. \$0.50 per ration will be sub-tracted from your reward.

Please indicate how many rations you wish to provide below:

[Numeric field to indicate number of rations, restricted to integers between 0 and 14]

<u>Part II – Info:</u>

One nutritional ration, which feeds one malnourished child for one week, can be provided by the charity for a donation of \$3.50. The

nutritious food consists of a **specially developed paste** and **energy-rich biscuits** that help children gain weight.

In this survey, you may donate all, part, or none of your reward of \$7.00 for this MTurk HIT to Sign of Hope for providing the nutritious food. Thus, you may choose any amount from \$0 to \$7.00. The amount you choose will be subtracted from your reward.

Please indicate how much money you wish to donate below:

[Numeric field to indicate dollar amount]

Part II – Info + Frame:

One nutritional ration, which feeds one malnourished child for one week, can be provided by the charity for a donation of \$3.50. The nutritious food consists of a specially developed paste and energy-rich biscuits that help children gain weight.

In this survey, you may use all, part, or none of your reward of \$7.00 for this MTurk HIT to provide these nutritional rations. Thus, you may choose any number from 0 to 2 rations (including fractions). \$3.50 per ration (or the appropriate fraction) will be subtracted from your reward.

Please indicate how many rations you wish to provide below:

[Numeric field to indicate number of rations]

Part II – Info + Restricted:

One nutritional ration, which feeds one malnourished child for one week, can be provided by the charity for a donation of \$3.50. The nutritious food consists of a specially developed paste and energy-rich biscuits that help children gain weight.

In this survey, you may donate all, part, or none of your reward of \$7.00 for this MTurk HIT to Sign of Hope for providing the nutritious food. In

particular, you may choose an amount of \$0, \$3.50, or \$7.00. The amount you choose will be subtracted from your reward.

Please indicate how much money you wish to donate below:

[Numeric field to indicate dollar amount, restricted to \$0, \$3.50, and \$7.00]

B.2 Additional Tables

	v		v			
Treatment	Female	Age (years)	Has children	College graduate	Crowd- worker	N
A. Pure schemes						
Money Donation	$\begin{array}{c} 0.52 \\ (0.50) \end{array}$	$37.20 \\ (10.64)$	$\begin{array}{c} 0.47 \\ (0.50) \end{array}$	$\begin{array}{c} 0.49 \\ (0.50) \end{array}$	$\begin{array}{c} 0.25 \\ (0.43) \end{array}$	152
Unit Donation – Large $(p = \$3.50)$	$\begin{array}{c} 0.52 \\ (0.50) \end{array}$	$37.34 \\ (10.30)$	$\begin{array}{c} 0.50 \\ (0.50) \end{array}$	$\begin{array}{c} 0.53 \ (0.50) \end{array}$	$\begin{array}{c} 0.23 \\ (0.42) \end{array}$	121
Unit Donation – Small $(p = \$0.50)$	$\begin{array}{c} 0.60 \\ (0.49) \end{array}$	$37.42 \\ (10.20)$	$\begin{array}{c} 0.48 \\ (0.50) \end{array}$	$\begin{array}{c} 0.48 \\ (0.50) \end{array}$	$\begin{array}{c} 0.23 \\ (0.42) \end{array}$	146
B. Intermediate schemes						
Info $(p = $3.50)$	$\begin{array}{c} 0.52 \\ (0.50) \end{array}$	$36.71 \\ (10.68)$	$\begin{array}{c} 0.46 \\ (0.50) \end{array}$	$\begin{array}{c} 0.53 \ (0.50) \end{array}$	$\begin{array}{c} 0.28 \\ (0.45) \end{array}$	150
Info + Unit frame $(p = \$3.50)$	$\begin{array}{c} 0.54 \\ (0.50) \end{array}$	$36.33 \\ (10.40)$	$\begin{array}{c} 0.43 \\ (0.50) \end{array}$	$\begin{array}{c} 0.48 \\ (0.50) \end{array}$	$\begin{array}{c} 0.19 \\ (0.39) \end{array}$	132
Info + Restricted $(p = \$3.50)$	$\begin{array}{c} 0.55 \\ (0.50) \end{array}$	$37.65 \\ (11.34)$	$\begin{array}{c} 0.45 \\ (0.50) \end{array}$	$\begin{array}{c} 0.44 \\ (0.50) \end{array}$	$\begin{array}{c} 0.31 \\ (0.47) \end{array}$	146
F-test (p-value)	0.75	0.91	0.93	0.71	0.25	

Table B.1: Summary Statistics by Treatment

Notes: Standard deviations in parentheses. The number of observations per variable does not always correspond to the number of subjects in the treatment group since we did not force subjects to answer the survey questions. The last row shows the p-value of the F-test for the joint significance of the treatment dummies when regressing the respective variable on the treatment dummies and a constant.

C Appendix to Chapter 3

C.1 Ordered Probit Model

Consider an individual, i, who decides how many units, g_i , of the charitable good to provide. Since the individual has a limited endowment and only complete units of the good can be provided, the individual faces a discrete and ordered choice set. For example, subjects who are assigned to the control condition can give 0, 1, 2, 3 or 4 packages. We estimate an Ordered Probit Model with the individual choice as dependent variable, which is subsequently employed to analyze the effect of the different subsidies on the level of charity receipts. This is possible, since, given a subsidy scheme, the individual choice directly translates into a level of charity receipts. The advantage of this procedure will become clear after explaining the model in more detail.

The model is based on a latent variable

$$g_i^* = x_i'\beta + s_i'\gamma + \epsilon_i \tag{C.1}$$

where x_i is a vector of covariates, including a constant, s_i is a vector consisting of a dummy for each subsidy type as well as subsidy type specific dummies for whether the offered subsidy rate is high and therefore the effective price is low (\$0.25), β and γ are vectors of parameters to be estimated and ϵ_i is an i.i.d. standard normally distributed error term. In general, each of the possible choices an individual can make, $g_i \in \{g^1, ..., g^J\}$, is associated with a certain interval of the latent variable:

$$g_i = g^j \qquad \text{if } \alpha_{j-1} < g_i^* \le \alpha_j \qquad \text{for } j = 1, ..., J \qquad (C.2)$$

where α_0 and α_J are set to $-\infty$ and ∞ , respectively, $\alpha_1 = 0$ and $\alpha_2, ..., \alpha_{J-1}$ are threshold parameters to be estimated.

A specific feature of the experimental design that we need to account for in the estimation is that the choice sets differ across treatments. For subjects facing a match, rebate, or no subsidy, each selected unit requires an expenditure of \$0.5. Although in case of the rebate, part of this expenditure is refunded, this refund cannot be donated (similar to most money donation experiments). Since the endowment is \$2, the maximum number of packages that can be selected in

those treatment conditions is four. In contrast, each unit selected in the discount treatments requires an expenditure of only \$0.33 or \$0.25 since the nominal price per unit is discounted upfront. Therefore, subjects can select up to six or eight packages, depending on whether the discount rate is low or high (see Table C.3 in Appendix C.2).

We account for this by adding censoring to the model. Since we do not observe a choice of seven packages in our data, we cannot include this category in the model. Furthermore, only a single subject provided five packages. In our main analysis, we treat this observation as if the subject had donated six packages. Results are similar if we explicitly include the choice category of five packages or omit the observation.¹⁸ Consequently, the choice sets in the following analysis are $g_i \in \{0, 1, 2, 3, 4, 6, 8\}$ for the 50% discount treatment, $g_i \in \{0, 1, 2, 3, 4, 6\}$ for the 33% discount treatment and $g_i \in \{0, 1, 2, 3, 4\}$ for all other treatments. Table C.1 in Appendix C.2 illustrates how the latent variable translates into a certain choice conditional on the treatment.

Let g_i^{max} be the maximum number of packages an individual *i* can give, which depends on the treatment the individual is assigned to. The probability to observe a choice g_i from the set $\{g^1, ..., g^7\} = \{0, 1, 2, 3, 4, 6, 8\}$ is then given by

$$Pr(g_{i} = g^{j} | x_{i}, s_{i}) = \mathbb{1} \{ g^{j} < g_{i}^{max} \} \{ \Phi(\alpha_{j} - x_{i}'\beta - s_{i}'\gamma)$$

- $\Phi(\alpha_{j-1} - x_{i}'\beta - s_{i}'\gamma) \}$
+ $\mathbb{1} \{ g^{j} = g_{i}^{max} \} \{ 1 - \Phi(\alpha_{j-1} - x_{i}'\beta - s_{i}'\gamma) \}$
for $j = 1, ..., 7$ (C.3)

The parameters $\theta = (\beta, \gamma, \alpha_2, \alpha_3, \alpha_4, \alpha_5, \alpha_6)$ are estimated by maximum likelihood, without and with covariates. The covariates include indicator variables for gender, marital status, whether the individual holds a college degree, whether children under the age of 16 live in the household, whether the individual is a registered voter, whether the individual frequently attends religious services, whether the individual works for a not-for-profit organization, and for task order. We also include categorical variables for age, income, residential environment, and religion as well as scores for the Big Five personality dimensions and risk preferences.

To check for misspecification of the model we use the Lagrange Multiplier test

 $^{^{18}\}mathrm{Results}$ are available from the authors upon request

derived by Glewwe (1997). In both model specifications, the Null of normally distributed error terms cannot be rejected (p > 0.35 and p > 0.90, without and with covariates, respectively). We also find little evidence that the results are substantively affected by allowing for heteroscedasticity.¹⁹

We use the estimated coefficients of the model to calculate the average marginal effect of each subsidy on charity receipts. The formulas used to calculate these effects are based on the deterministic relationship between the individual choice and charity receipts. Using the individual choice as dependent variable simplifies the estimation procedure: If the choice sets of any two treatments differ, one is a subset of the other, and the smaller set is simply censored from above. Furthermore, the selected number of packages represents subjects' immediate choice and therefore is probably the most intuitive concept for modeling the decision process. The average marginal effects are calculated as follows. We can explicitly write s_i in equation (C.1) as

$$s_{i} = \begin{pmatrix} rebate_{i} \\ match_{i} \\ discount_{i} \\ rebate_{i} \times low \ price_{i} \\ match_{i} \times low \ price_{i} \\ discount_{i} \times low price_{i} \end{pmatrix},$$

where $rebate_i$, $match_i$, and $discount_i$ are dummies for whether individual *i* faces a particular subsidy type, $rebate_i \times low \ price_i$, $match_i \times low \ price_i$, and $discount_i \times low \ price_i$ are subsidy type specific dummies indicating whether the subsidy rate is high and the effective price is low (\$0.25).

We use the estimated parameters $\hat{\theta}$ and the deterministic relationship between the selected number of packages g_i (individual's choice) and charity receipts cr_i to

¹⁹We expand the model in column 2 of Table 3.4 by modeling the variance as $exp(z'_i\rho)$. We estimate this model with different sets of covariates included in z_i . Set 1 includes age, income, gender, whether the individual frequently attends religious services, and task order. Set 2 additionally contains the Big Five personality dimensions and risk preferences. Set 3 includes all covariates. Only if we use the whole set of covariates to explicitly model heteroscedasticity, the model with homoscedasticity is rejected (p < 0.01). Still, rebates and matches do not significantly differ in the level of charity receipts at the low subsidy rate, but the difference approaches marginal significance at the high subsidy rate (where charity receipts are higher under matches). One should be careful with relying on this expanded model specification as it is sensitive to small changes in the set of covariates included. Results are available from the authors upon request.

calculate the expected level of charity receipts \hat{E}_i that is predicted by the model for each individual under each treatment condition. For example, to receive individual *i*'s expected level of charity receipts under the 50% rebate we set $s_{i,r25} = (1, 0, 0, 1, 0, 0)'$ – the subscript *r*25 indicates the subsidy type and the implied effective price in cents – and calculate the predicted value of the latent variable according to equation (C.1):

$$\hat{g}_{i,r25}^* = x_i'\hat{\beta} + s_{i,r25}'\hat{\gamma}$$

Afterwards, we estimate the expected level of charity receipts by

$$\hat{E}_{i,r25} = \sum_{k=0}^{8} k\hat{P}(cr_i = k | x_i, s_i = s_{i,r25}) = \sum_{k=0}^{4} k\hat{P}(g_i = k | x_i, s_i = s_{i,r25})$$
$$= 4 - \Phi(\hat{a}_4 - \hat{g}_{i,r25}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,r25}^*) - \Phi(\hat{a}_2 - \hat{g}_{i,r25}^*) - \Phi(-\hat{g}_{i,r25}^*)$$

where $\hat{P}(cr_i = k | x_i, s_i = s_{i,r25}) = 0$ for k > 4 since the maximum level of charity receipts under the rebate is four. The second equality then follows from the fact that for all treatment conditions except matches, the individual choice (in physical units) is equal to the level of charity receipts (in physical units). The last equality follows from using equation (C.3) to calculate $\hat{P}(y_i = k | x_i, s_i = s_{i,r25})$. The expected levels of charity receipts for the other six conditions (no subsidy, 33% rebate, 1:2 match, 1:1 match, 33% discount, and 50% discount) are accordingly calculated as

$$\begin{split} \hat{E}_{i,n50} &= 4 - \Phi(\hat{a}_4 - \hat{g}_{i,n50}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,n50}^*) - \Phi(\hat{a}_2 - \hat{g}_{i,n50}^*) - \Phi(-\hat{g}_{i,n50}^*) \\ \hat{E}_{i,r33} &= 4 - \Phi(\hat{a}_4 - \hat{g}_{i,r33}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,r33}^*) - \Phi(\hat{a}_2 - \hat{g}_{i,r33}^*) - \Phi(-\hat{g}_{i,r33}^*) \\ \hat{E}_{i,m33} &= 6 - 2\Phi(\hat{a}_4 - \hat{g}_{i,m33}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,m33}^*) - 2\Phi(\hat{a}_2 - \hat{g}_{i,m33}^*) - \Phi(-\hat{g}_{i,m33}^*) \\ \hat{E}_{i,m25} &= 8 - 2\Phi(\hat{a}_4 - \hat{g}_{i,m25}^*) - 2\Phi(\hat{a}_3 - \hat{g}_{i,m25}^*) - 2\Phi(\hat{a}_2 - \hat{g}_{i,m25}^*) - 2\Phi(-\hat{g}_{i,m25}^*) \\ \hat{E}_{i,d33} &= 6 - 2\Phi(\hat{a}_5 - \hat{g}_{i,d33}^*) - \Phi(\hat{a}_4 - \hat{g}_{i,d33}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,d33}^*) - \Phi(\hat{a}_2 - \hat{g}_{i,d33}^*) \\ - \Phi(-\hat{g}_{i,d33}^*) \\ \hat{E}_{i,d25} &= 8 - 2\Phi(\hat{a}_6 - \hat{g}_{i,d25}^*) - 2\Phi(\hat{a}_5 - \hat{g}_{i,d25}^*) - \Phi(\hat{a}_4 - \hat{g}_{i,d25}^*) - \Phi(\hat{a}_3 - \hat{g}_{i,d25}^*) \\ - \Phi(\hat{a}_2 - \hat{g}_{i,d25}^*) - \Phi(-\hat{g}_{i,d25}^*) \end{split}$$

We use the expected level of charity receipts to calculate average marginal effects (AMEs) for introducing a subsidy type at the low rate (rebate, match, discount)

and for changing the subsidy rate for a specific subsidy type from low to high (rebate \times low price, match \times low price, discount \times low price):

$$AME_{rebate} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,r33} - \hat{E}_{i,n50}$$
$$AME_{match} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,m33} - \hat{E}_{i,n50}$$
$$AME_{discount} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,d33} - \hat{E}_{i,n50}$$
$$AME_{rebate \times low \ price} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,r25} - \hat{E}_{i,r33}$$
$$AME_{match \times low \ price} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,m25} - \hat{E}_{i,m33}$$
$$AME_{discount \times low \ price} = \frac{1}{N} \sum_{i=1}^{N} \hat{E}_{i,d25} - \hat{E}_{i,d33}$$

These average marginal effects are presented in column 1 and 2 of Table 3.4. Standard errors are calculated based on the delta method.

C.2 Additional Figures and Tables

	Individual	choice (g_i) in	
$\begin{array}{c} \text{Latent} \\ \text{variabel} \\ g_i^* \end{array}$	no subsidy control, 33% rebate, 1:2 match, 50% rebate, 1:1 match	33% discount	50% discount
$(-\infty, 0]$	0	0	0
$(0, \alpha_2]$	1	1	1
$(\alpha_2, \alpha_3]$	2	2	2
$(\alpha_3, \alpha_4]$	3	3	3
$(\alpha_4, \alpha_5]$	4	4	4
$(\alpha_5, \alpha_6]$	4	6	6
(α_6,∞)	4	6	8

 Table C.1:
 Latent Variable and Individual Choice

	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	μ 0.45 0.22 0.39	ontrol		33% 1	rebate		1:2 n	ıatch		33% di:	scount		50% r	ebate		1:1 mat	tch.	50%	% disco	unt	F-test
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	aile 0.48 0.50 558 $^{\circ}$ (Years): 0.25 0.44 558 -354 0.27 0.44 558 -34 0.37 0.48 558 -34 0.37 0.48 558 -34 0.37 0.46 558 55 0.34 0.27 554 100 0.33 0.47 554 100 0.34 0.37 558 0.34 0.34 0.37 558 0.90 0.34 0.47 558 0.97 0.34 0.37 558 0.90 0.30 0.47 558 0.90 0.90 0.34 558 0.090 0.90 0.31 0.33 540 0.000 0.10 0.23 0.41 540 0.000 0.21 0.21 0.35 540 0.0000 <	0.45 0.22 0.39	σ		μ	υ		п	σ	~ >	1	0	2	μ	ь Г	1 μ	σ	N	±	υ	N	p-value
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$0.22 \\ 0.39$	0.50	83 (0.49 0	.50	71 0.	48 0	.50 8	15 0.	44 0.	50 5	30 0.	.55 0.	50 5	8 0.4	4 0.5(08 (0.53	0.50	91	0.75
	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.39	5		000	1 1	c F	000		()	1		9	0	0	0	, ,		0 7 0		5	
	$\sum_{5.6 \\ 1.5.6 \\ 5.5.6 \\ 5.5.6 \\ 1.5.6 \\ 1.5.6 \\ 1.5.6 \\ 1.5.6 \\ 1.5.6 \\ 1.5.8 \\ 1.5.6 \\ 1.5.8 \\ 1.5.6 \\ 1$		0.49	200	1.38	49	0. 11	34 0	- 44 - 84 - 80	ç ç Ç Ç	37 0.	48 5 5		. 30 31 0.	470 470	8 0.4 0.4	0.45	80	0.41	0.49	16	0.00
	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.29	0.46	83 (0.30 0	.46	71 0.	34 0	.48	55 O.	29 0.	.46 5	90 0.	24 0.	43 5	8 0.2	3 0.42	2 80	0.35	0.48	91	0.53
	$\label{eq:constraint} {\begin{tabular}{lllllllllllllllllllllllllllllllllll$	0.11	0.31	83	0.04 6	.20	71 0.	0.06	.24 8	.0 .0	08 0.	27 5	90 0.	0.09	28	8 0.1	0 0.3(80	0.08	0.27	91	0.70
	$\begin{array}{c} \mbox{legree} & 0.47 & 0.50 & 558 \\ \mbox{omb}^b (US8); & 0.09 & 0.57 & 558 \\ \mbox{omb}^b (US8); & 0.09 & 0.28 & 540 \\ \mbox{olucion}_{-9}999 & 0.11 & 0.31 & 540 \\ \mbox{olucion}_{-9}9999 & 0.11 & 0.33 & 540 \\ \mbox{olucion}_{-9}9999 & 0.11 & 0.35 & 540 \\ \mbox{olucion}_{-4}9999 & 0.11 & 0.35 & 540 \\ \mbox{olucion}_{-9}9999 & 0.11 & 0.35 & 540 \\ \mbox{olucion}_{-9}9999 & 0.12 & 0.21 & 0.41 & 540 \\ \mbox{olucion}_{-9}9999 & 0.09 & 0.21 & 0.41 & 540 \\ \mbox{olucion}_{-9}999 & 0.09 & 0.21 & 0.21 & 0.41 & 540 \\ \mbox{olucion}_{-9}9999 & 0.09 & 0.21 & 0.21 & 0.21 \\ \mbox{olucion}_{-9}9999 & 0.09 & 0.21 & 0.21 & 0.21 & 0.21 \\ \mbox{olucion}_{-9}9999 & 0.09 & 0.21 & 0.21 & 0.21 & 0.21 & 0.21 & 0.21 \\ \mbox{olucion}_{-9}9999 & 0.09 & 0.21 & $	0.35	0.48	200	0.30	, 46 146	12		.46 46 8	4 	88 8 7 0 0	19 19 19		.31 91	47 77 70	5 i7 ⊃ ⊂	× 0.4	5 79 50	0.41	0.49	90	0.49
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.49	0.50	83	0.46	20	71 0.	49 0	.50	22 00	43 0.	50 9	00	47 0.	50 5	8 0.4	1 0.50	80	0.48	0.50	91	0.97
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0,000 0.09 0.09 0.28 540 $0,000-19,999$ 0.11 0.21 540 $0,000-19,999$ 0.11 0.33 540 $0,000-29,999$ 0.11 0.33 540 $0,000-39,999$ 0.11 0.32 540 $0,000-74,999$ 0.15 0.35 540 $5,0000-74,999$ 0.21 0.41 540 $5,000-99,999$ 0.09 0.28 540																					
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.07	0.26	81 (0.10 6	.31	68 0.	0.7 0	.26 8	34 0.	10 0.	31 8	36 0.	.07 0.	26 5	5 0.1	4 0.3	5 77	0.06	0.23	88	0.62
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.09	0.28	81 C	0 00.0	.29	68 0.	.13 0	.34 8	34 O.	10 0.	.31 8	36 0.	.11 0.	31 5	5 0.1.	3 0.34	1 77	0.11	0.32	88	0.96
	0,000-39,999 0.11 0.32 $54010,000-49,999$ 0.15 0.35 $54010,000-74,999$ 0.21 0.41 $54075,000-99,999$ 0.09 0.28 540	0.14	0.34	81 (0.16 6	.37	68 0.	08 0	388	0.	08 0.	38	36 0.	.07 0.	26 5	5 0.1.	0 03	1 77	0.19	0.40	80	0.20
	10,000-49,999 0.10 0.30 $0.4050,000-74,999$ 0.21 0.41 $54075,000-99,999$ 0.09 0.28 540	0.10	0.30	20	0.12	27.0	89	0 01.0	s 040 s 060	57 0 0	.09 10	20	90 90	.11	10 10 10	000	0.77 		0.10	0.30	x 0 x 0	0.63
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	75,000-99,999 0.09 0.28 540	0.00	777	50 61 61	- 15 7 15	70		0 C	2 0 90.00 000	5 C	- 1 A	50 20 20	0 0	0 1 C	נע 11 ק	0 0	2 0.4.	- 22	11.0	0.07	00	10.0
		0.07	0.26	201	0 600	100	800		280		12 0.	100	0.0	14 0.	1 10 7 10 7 10	100	30.0	- 2-2-	1770	0.25	0 00	0.76
	100,000 0.12 0.32 540	0.15	0.36	81 C	0.19 0	.40	58 0.	11 0	.31 8	4 0.	00 0.	29 8	36 0.	11 0.	31 5	0.0	8 0.27	77 77	0.11	0.32	88	0.52
	sidential environment:																					
	0.20 0.40 558	0.16	0.37	83	0.20 6	.40	71 0.	22 0	.42 8	55 0.	21 0.	.41 5	90 0.	.12 0.	33 5	8 0.2	4 0.4:	3 80	0.22	0.42	91	0.48
	burban $0.51 0.50 558$	0.60	0.49	200	.56 .54	. 20	17 0.0	41 0	30.30	ມີ ເ	50 0.	02 0 0	00	.53 0.	000 000 001	8 4 6	0.0 .0	080	0.53	0.50	91	0.21
First function for a constraint of the c	10an 0.29 0.40 0.58	1.24	0.43	200	1.24	.43	11	00.00	2 0 2 0 2 0	2 C	50 00	3 06		.34 U.	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 0 0 0	1 0.4	100	0.20	0.44	16 00	0.40
	structed voter 0.00 0.04 0.02 t-for-profit 0.06 0.03 558	0.00	0.15	207	1.91	140	0 0		00. 80 80	4, ič ⊃ ⊂	70 70	21 21 20		04	000 200 200	070	0.0	808	0.09	0.02	90 10	0.00
	igions ^d 0.13 0.34 548	20.0	0.26	83	0 010	30	12	0 20	, x , x	; ⊂ 0	15 0	36.95	0 9	0	9 - C 2 - C	10 8	1 0.35	808	0.16	0.37	68	0.38
	ligion:																					
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	heist 0.38 0.49 533	0.37	0.49	81 (0.40 0	.49	65 0.	42 0	.50 8	33 0.	46 0.	50 8	35 0.	32 0.	47 5	7 0.3	3 0.4	7 75	0.32	0.47	87	0.44
$ \begin{array}{l l l l l l l l l l l l l l l l l l l $	gostic 0.08 0.28 533	0.10	0.30	81 51	0.14 6	.35	65 0	0.7 0	.26	33 0.	07 0.	26 8	35 .0	05 0.	23	7 0.0	7 0.2	5 75	0.08	0.27	87	0.76
	man-Catholic 0.12 0.32 533	0.10	0.30	100	0.12		65 . 0.	90.0	2 4 2 7 2 7 2 7 2 7 2 7 2 7 2 7 2 7 2 7 2 7		0. 00.	2007	85 25 0.0	.14 0.	0.0 0 0 0 1 0	7 0.1	5 5 5 6 6 7 6	7 75	0.17	0.38	2 1 2	0.15
	0.005.04116 0.20 0.23 her Christian 0.12 0.22 5.23	71.0	0.36	50		85	00 32 00	0 0 0	277 777 80	2 G	07	04.0 9 0 0 8 0 0	00		100 100 100 100 100 100 100 100 100 10	1.0	0 0 0 0 0	100	0.10	10.0	- 1- 0 0	0.04
$ eq: scale 1-7): \math transform 3.18 \math transform (3.16) $51 \math transform (3.16) $51 \math transform (3.18) $51 \$	her 0.12 0.33 533	0.16	0.37	81 0	0.111	.31	35 0.	080	. 28 . 28	; 0 ; 0	13 0.	191 191 191 192	0.0 0.0	16 0.	37 5	7 0.1.	1 0.31	1 75	0.11	0.32	87	0.78
	Five (scale $1-7$):																					
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	traversion 3.18 1.60 520	3.14	1.55		2.00	14-	67	42	- 69 	- 1 - 1	01	90. 3 0	ις Γου Γου	.41 1.	69 67	4 7 0 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1 7 1	4 1.5	74	3.16	1.60	980 80	0.67
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	greeableness 0.04 1.24 020	4.94 101	1.20	и т - 0 - 1	1 1		000 000			т н т		0 0 1 0 0 0	о и о и	1 07	40 40 10 10 10	0 0 0	0 1 1 0 1 0	1 1 1 1	0.00 1 1 0	1.00	4 C	11.0
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	notional stah. 4.62 1.53 531	4.57	1.38	- 70 - 70 - 70	1 1 1 1	440	0 80 0 80	- 1 1 1 1 1		10 70	42 1.	194 194 194	0 9 9 9	73 1.	1 X C	- 12 - 12 - 12	0.10	292	4.66	1.46	1 9 0 00	0.86
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	tenness 4.67 1.29 532	4.71	1.15	80.4	67	22	- 7 - 7 - 7	77 1	21 2	7	86	212	22	73	20 5	5 4 7	5 1.50	2.0	4.25	1.40	87	0.10
	k pref. (scl. 1–6) 4.06 1.78 554	3.96	1.82	83	1.25	8	71 4.	05 1	.77	4.	18	73 8	39	83 1.	74 5.0	3 4.20	1.75	80	3.94	1.87	68	0.76
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	k order ^e 0.52 0.50 558	0.43	0.50	83	0.52 0	.50	71 0.	45 0	50 8	0. 0	58 0.	50 5	0.0	59 0.	50 5	8 0.55	9 0.50	80	0.53	0.50	91	0.24
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	nipulationon check questions(scale 1-5):																					
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	$arity^f$ 4.58 0.67 551	4.56	0.69	82 4	1.54 0	.70	70 4.	.60 0	.60 8	34 4.	57 0.	.75 8	39 4.	.60 0.	60 5	5 4.6	4 0.55	3 80	4.55	0.79	91	0.97
$ \begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$	1000000000000000000000000000000000000	4.48	0.63	82 4	1.56 G	. 71	70 4.	.46 0	.72 8	34 4.	51 0.	3 97.	39 4.	.64 0.	56 5	5 4.4	7 0.67	7 80	4.41	0.89	91	0.52
$ \begin{array}{rrrrrrrrrr} to the the the the the the the the the the$	$ust exp.^{n}$ 4.04 0.93 549	4.08	0.87	80 4	i.07 1	.02	71 4.	.13 0	.92 8	34 4.	02 0.	3 76.	38 4.	.02 0.	82 5	6 4.0.	5 0.9	5 79	3.93	0.96	91	0.90
eserving recip. ⁷ 4.48 0.82 549 4.47 0.74 81 4.43 0.94 70 4.52 0.75 84 4.40 0.91 89 4.71 0.53 55 4.58 0.63 79 4.32 0.99 91 0.03	ust charity ^{i} 4.13 0.90 550	4.17	0.78	80 4	1.17 0	. 93	71 4.	24 0	89 83.	34 4.	01 1.	.02 8	39 4.	.09 0.	86 5	5 4.1,	8 0.8	7 79	4.08	0.93	91	0.75
	serving recip. ³ 4.48 0.82 549	4.47	0.74	81 4	1.43 G	.94	70 4.	.52 0	.75 8	34 4.	40 0.	3 10.	39 4.	.71 0.	53 5	5 4.5.	8 0.6	3 79	4.32	0.99	91	0.03

C Appendix to Chapter 3

Treatment	Individual choice [units]	Corresponding net donations [\$]	Corresponding charity receipts [units]
No subsidy	$\begin{array}{c} 0 \\ 1 \\ 2 \\ 3 \\ 4 \end{array}$	$\begin{array}{c} 0 \\ 0.5 \\ 1 \\ 1.5 \\ 2 \end{array}$	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \end{array}$
33% rebate	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \end{array}$	$\begin{array}{c} 0 \\ 0.33 \\ 0.66 \\ 0.99 \\ 1.32 \end{array}$	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \end{array}$
1:2 match	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \end{array}$	$\begin{array}{c} 0 \\ 0.5 \\ 1 \\ 1.5 \\ 2 \end{array}$	$egin{array}{c} 0 \ 1 \ 3 \ 4 \ 6 \end{array}$
33% discount	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \ 5 \ 6 \ \end{array}$	$\begin{array}{c} 0 \\ 0.33 \\ 0.66 \\ 0.99 \\ 1.32 \\ 1.65 \\ 1.98 \end{array}$	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \ 5 \ 6 \ \end{array}$
50% rebate	$0 \\ 1 \\ 2 \\ 3 \\ 4$	$0 \\ 0.25 \\ 0.5 \\ 0.75 \\ 1$	$\begin{array}{c} 0\\ 1\\ 2\\ 3\\ 4\end{array}$
1:1 match	$egin{array}{c} 0 \ 1 \ 2 \ 3 \ 4 \end{array}$	$\begin{array}{c} 0 \\ 0.5 \\ 1 \\ 1.5 \\ 2 \end{array}$	$egin{array}{c} 0 \\ 2 \\ 4 \\ 6 \\ 8 \end{array}$
50% discount	$egin{array}{c} 0 \\ 1 \\ 2 \\ 3 \\ 4 \\ 5 \\ 6 \\ 7 \\ 8 \end{array}$	$\begin{array}{c} 0\\ 0.25\\ 0.5\\ 0.75\\ 1\\ 1.25\\ 1.5\\ 1.75\\ 2\end{array}$	$egin{array}{cccc} 0 & 1 & & \ 2 & 3 & & \ 3 & 4 & & \ 5 & 6 & & \ 7 & & \ 8 & & \ \end{array}$

Table C.3: Choice Set by Treatment

Treatment	Charity r	receipts
	unconditional (units)	conditional (units)
	(1)	(2)
A. Mean values (S.D.)		
No subsidy	1.169 (1.413)	2.256 (1.177)
33% rebate	(1.690) (1.545)	2.400 (1 294)
1:2 match	(1.010) 1.271 (1.538)	2.571 (1.192)
33% discount	1.233 (1.446)	2.313 (1.188)
50% rebate	1.931 (1.705)	(1.3732) (1.379)
1:1 match	1.800 (1.702)	3.064 (1.009)
50% discount	1.495 (1.615)	2.473 (1.372)
B. Tests of subsidy types: p-va	lues	. ,
 B1. At effective price of \$0.33 33% rebate vs. 1:2 match 33% rebate vs. 33% discount 1:2 match vs. 33% discount 	$0.09 \\ 0.06 \\ 0.87$	$0.51 \\ 0.73 \\ 0.31$
B2. At effective price of \$0.25 50% rebate vs. 1:1 match 50% rebate vs. 50% discount 1:1 match vs. 50% discount	$0.65 \\ 0.12 \\ 0.23$	$\begin{array}{c} 0.21 \\ 0.36 \\ 0.01 \end{array}$
C. Tests of subsidized prices: p	p-values	
50% vs. 33% rebate 1:1 vs. 1:2 match 50% vs. 33% discount	$0.41 \\ 0.04 \\ 0.25$	$0.24 \\ 0.04 \\ 0.53$
D. Tests of subsidized vs. unsu	ubsidized prices:	<i>p</i> -values
 D1. Low subsidy rate 33% rebate vs. no subsidy 1:2 match vs. no subsidy 33% discount vs. no subsidy 	$0.03 \\ 0.66 \\ 0.77$	$0.58 \\ 0.22 \\ 0.82$
D2. High subsidy rate 50% rebate vs. no subsidy 1:1 match vs. no subsidy 50% discount vs. no subsidy	$0.01 \\ 0.01 \\ 0.16$	$0.09 \\ 0.00 \\ 0.40$

Table C.4: Robustness Check for Charity Receipts Censored at 4 Packages

Notes: Panel A shows mean values of the donation variables for each treatment (standard deviations in parentheses). Column 1 shows unconditional charity receipts with each number of packages above four recoded to four. Column 2 shows the corresponding numbers for charity receipts conditional on being a donor. Shown in panels B and C are p-values of two-tailed t-tests with unequal variances.

C.3 Instructions

In this survey, each participant will have the opportunity to provide nutritional packages for malnourished children in the African country of South Sudan.



Decades of civil war have ravaged South Sudan, and many children are severely malnourished. The nutritional packages will be delivered by Sign of Hope, an accredited relief organization that operates two hospitals in South Sudan.

In 2010, Sign of Hope won the Transparency Award for German non-profit organizations. Eighty cents out of every dollar they receive go directly to relief efforts, while the remaining twenty cents cover their overhead costs. You can read more about the organization on http://www.hoffnungszeichen.de/sign-ofhope-africa.html.



One nutritional package, which feeds one malnourished child for one day, can be provided by the charity for a donation of \$0.50. The package consists of a specially designed paste and high-energy cookies to help the children gain weight.

In this survey, you will be able to provide these nutritional packages for **\$0.25** apiece (a third party will fund the remaining \$0.25). You may use all, part, or none of your reward of \$2.00 for this HIT to provide packages. Thus, you may choose any number between 0 and 8 packages. \$0.25 per package will be subtracted from your reward.

Please indicate your choice below:

•

Figure C.1: Example Donation Appeal, 50% Discount Treatment. The Final Paragraph Differed between Treatments.

C.4 Within-Subjects Design

If the results of the within-subjects (WS) design mirrored the results of the between-subjects (BS) design, the WS variation could be used to learn more about how those results come about. However, as we will show, the results of the WS design substantially differ. Although under these circumstances we rank the external validity of the BS design higher, the WS data can provide insights into subjects' decision process when they are forced to compare different conditions.

In the experiment, 146 subjects were randomly assigned to the WS treatment in which all seven treatment conditions were jointly displayed on the donation call page in random order (Figure C.2). Instructions informed subjects that one of the conditions would be randomly selected through a lottery and implemented. Subjects then entered, for each condition, an integer number indicating their desired number of units. 119 subjects completed the survey of which 6 were subsequently removed because of duplicate IP addresses. Table C.5 shows summary statistics of the sample, including p-values from pairwise comparisons with the sample that was assigned to the BS design.

Table C.6 presents the main results analogously to Table 3.3. Beginning with the unconditional level of charity receipts in column 3, we observe that under a WS design, matches and discounts are more effective in providing the charitable good than rebates. This finding is most pronounced for the low price of \$0.25. Potentially, the discontinuities in the match – the first and third unit funded not resulting in an additional matched unit – discourage giving at the effective price of \$0.33. For net donations, we observe in column 2 that introducing matches and discounts significantly crowds in net donations while an increase in the rebate rate induces crowding-out.

At first glance the discrepancy in results compared to the BS sample might come as a surprise, but a closer look at the extensive and intensive margins in columns 4 and 5 offers a simple explanation for most differences. Unlike in the BS design, we find that for a given effective price, subsidies are equally successful in attracting donors at the extensive margin. We speculate that under a WS design, subjects may not decide whether to donate for each subsidy separately, but rather make a single participation choice across all subsidies with a similar rate and then respond to the subsidy type mostly at the intensive margin. This explanation is also supported by looking at the decision of whether to give at the individual

C Appendix to Chapter 3

In this survey, each participant will have the opportunity to provide nutritional packages for malnourished children in the African country of South Sudan.



Decades of civil war have ravaged South Sudan, and many children are severely malnourished. The nutritional packages will be delivered by Sign of Hope, an accredited relief organization that operates two hospitals in South Sudan.

In 2010, Sign of Hope won the Transparency Award for German non-profit organizations. Eighty cents out of every dollar they receive go directly to relief efforts, while the remaining twenty cents cover their overhead costs. You can read more about the organization on http://www.hoffnungszeichen.de/sign-ofhope-africa.html.



One nutritional package, which feeds one malnourished child for one day, can be provided by the charity for a donation of 0.50. The package consists of a specially designed paste and high-energy cookies to help the children gain weight.

In this survey, you may use all, part, or none of your reward of \$2.00 for this HIT to provide these nutrition packages.

Below, you find seven different price conditions. For each condition, please indicate the number of packages you choose to provide under this condition.

One of the seven price conditions will be implemented in real. This condition will be randomly selected through a lottery after this survey. So, since each case may become real with equal probability, you will want to answer each condition as if it is the condition that will be implemented .

	Your choice:
Condition C: In this condition, you will be able to provide these nutritional packages for \$0.50 apiece . Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted from your reward.	
Condition E: In this condition, you will be able to provide these nutritional packages for \$0.33 apiece (a third party will fund the remaining \$0.17). Thus, you may choose any number between 0 and 6 packages. \$0.33 per package will be subtracted from your reward.	
Condition D: In this condition, you will be able to provide these nutritional packages for $$0.25$ apiece (a third party will fund the remaining \$0.25). Thus, you may choose any number between 0 and 8 packages. $$0.25$ per package will be subtracted from your reward.	
Condition R: In this condition, you will be able to provide these nutritional packages for \$0.50 apiece. Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted form your reward. A third party has agreed to fund a 33% rebate for each package you provide. Upon completion of the survey, the rebate (\$0.71 per package provided) will be added to your reward.	
Condition S: In this condition, you will be able to provide these nutritional packages for \$0.50 apiece. Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted form your reward. A third party has agreed to fund a 50% rebate for each package you provide. Upon completion of the survey, the rebate (\$0.25 per package provided) will be addet by your reward.	
Condition M: In this condition, you will be able to provide these nutritional packages for \$0.50 apiece. Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted from your reward. A third party has agreed to match every two packages you provide, at no additional cost to you. So, for example, if you choose to provide 2 packages, Sign of Hope will receive 3.	
Condition N: In this condition, you will be able to provide these nutritional packages 67 60.50 apiece. Thus, you may choose any number between 0 and 4 packages. \$0.50 per package will be subtracted from your reward. A third party has agreed to match each package you provide, at no additional cost to you. So, for example, if you choose to provide 2 packages, Sign of Hope will receive 4.	

Figure C.2: Donation Appeal in the WS Design

level: 91 and 82 percent of individuals make the same decision of whether to give across the different subsidy types at the high and low subsidy rate, respectively. In contrast, the introduction of a subsidy and the height of its rate seem to be highly relevant for the participation decision, regardless of its type. This behavior is very different from the one observed in the BS design and likely to be affected by demand effects from "nudging" subjects to compare options in the WS design (Charness et al., 2012). We therefore follow the literature and ascribe higher external validity to the results in the between-subjects design.

Table C.5:Summary Statistics for BS Sample, WS Sample, and CombinedSample

	Variable	Comb	ined sa	mple	BS	5 sample	е	WS	5 sampl	e	Comparison
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		Mean	SD	N	Mean	SD	N	Mean	SD	N	<i>p</i> -value
$\begin{array}{l c c c c c c c c c c c c c c c c c c c$	Female	0.48	0.50	671	0.48	0.50	558	0.46	0.50	113	0.70
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Age (years):										
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	18-25	0.25	0.43	671	0.25	0.44	558	0.21	0.41	113	0.34
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	26 - 34	0.38	0.49	671	0.37	0.48	558	0.43	0.50	113	0.23
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	35 - 54	0.29	0.45	671	0.29	0.46	558	0.27	0.45	113	0.68
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	> 55	0.08	0.27	671	0.08	0.27	558	0.08	0.27	113	0.98
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Married	0.33	0.47	667	0.34	0.47	554	0.27	0.45	113	0.20
	$Children^{a}$	0.30	0.46	671	0.30	0.46	558	0.26	0.44	113	0.33
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	College degree	0.48	0.50	670	0.47	0.50	558	0.54	0.50	112	0.19
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Income ^{b} (US\$):										
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	<10.000	0.09	0.28	649	0.09	0.28	540	0.07	0.26	109	0.60
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	10,000 - 19,999	0.11	0.31	649	0.11	0.31	540	0.11	0.31	109	0.98
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	20.000 - 29.999	0.12	0.33	649	0.12	0.33	540	0.14	0.35	109	0.62
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	30.000-39.999	0.13	0.34	649	0.11	0.32	540	0.20	0.40	109	0.01
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	40.000-49.999	0.15	0.35	649	0.15	0.35	540	0.15	0.36	109	0.99
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	50.000 - 74.999	0.20	0.40	649	0.21	0.41	540	0.12	0.33	109	0.03
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	75.000-99.999	0.09	0.29	649	0.09	0.28	540	0.09	0.29	109	0.92
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	> 100,000	0.12	0.32	649	0.12	0.32	540	0.12	0.33	109	0.98
Rural 0.20 0.40 671 0.20 0.40 558 0.19 0.39 113 0.75 Suburban 0.52 0.50 671 0.51 0.50 558 0.54 0.50 113 0.57 Urban 0.29 0.45 671 0.29 0.45 558 0.27 0.45 113 0.73 Registered voter 0.87 0.34 663 0.86 0.34 552 0.88 0.32 111 0.56 Not-for-proft ^c 0.05 0.22 671 0.06 0.23 558 0.03 0.16 113 0.18 Religious ^d 0.13 0.34 659 0.13 0.34 548 0.14 0.34 111 0.87 Religion: $$	Residential environmen	nt:	0.0-	0 - 0	0	0.0-	0 - 0	0			0.000
Suburban 0.52 0.50 671 0.51 0.50 558 0.54 0.50 113 0.57 Urban 0.29 0.45 671 0.29 0.45 558 0.27 0.45 113 0.73 Registered voter 0.87 0.34 663 0.86 0.34 552 0.88 0.32 111 0.56 Not-for-profit ^c 0.05 0.22 671 0.06 0.23 558 0.03 0.16 113 0.18 Religious ^d 0.13 0.34 659 0.13 0.34 548 0.14 0.34 111 0.87 Religion: $$	Rural	0.20	0.40	671	0.20	0.40	558	0.19	0.39	113	0.75
Urban 0.29 0.45 671 0.29 0.45 538 0.27 0.45 113 0.73 Registered voter 0.87 0.34 663 0.86 0.34 552 0.88 0.32 111 0.56 Not-for-profit ^c 0.05 0.22 671 0.06 0.23 558 0.03 0.16 113 0.18 Religious ^d 0.13 0.34 659 0.13 0.34 548 0.14 0.34 111 0.87 Religion: $$	Suburban	0.52	0.50	671	0.51	0.50	558	0.54	0.50	113	0.57
Registered voter 0.87 0.34 663 0.86 0.34 552 0.88 0.32 111 0.56 Not-for-profit ^c 0.05 0.22 671 0.06 0.23 558 0.03 0.16 113 0.18 Religious ^d 0.13 0.34 659 0.13 0.34 548 0.14 0.34 111 0.87 Religion: A A 0.13 0.34 659 0.13 0.34 548 0.14 0.34 111 0.87 Agostic 0.09 0.28 643 0.08 0.28 533 0.16 0.30 110 0.55 Roman-Catholic 0.12 0.32 643 0.12 0.32 533 0.14 0.34 110 0.56 Protestant 0.18 0.38 643 0.18 0.38 533 0.17 0.38 110 0.93 Other Christian 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 626 5.18 1	Urban	0.29	0.45	671	0.29	0.45	558	0.27	0.45	113	0.73
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Registered voter	0.87	0.34	663	0.86	0.34	552	0.88	0.32	111	0.56
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Not-for-profit ^{c}	0.05	0.22	671	0.06	0.23	558	0.03	0.16	113	0.18
Religion: Atheist0.370.486430.380.495330.360.481100.79Agostic0.090.286430.080.285330.100.301100.55Roman-Catholic0.120.326430.120.325330.140.341100.56Protestant0.180.386430.180.385330.170.381100.93Other Christian0.120.336430.130.335330.110.311100.63Other0.120.336430.120.335330.110.311100.63Other0.120.336430.120.335330.120.321100.91Big Five (scale 1-7):Extraversion3.211.626263.181.605203.321.731060.45Agreeableness5.021.236285.041.245234.911.201050.34Conscientiousness5.111.296305.131.305254.971.241050.23Emotional stability4.651.526384.621.535314.791.491070.30Openness4.701.296404.671.295324.831.321080.25Risk pref. (scale 1-6)4.001.796674.061.785543.65 </td <td>$\mathbf{Religious}^d$</td> <td>0.13</td> <td>0.34</td> <td>659</td> <td>0.13</td> <td>0.34</td> <td>548</td> <td>0.14</td> <td>0.34</td> <td>111</td> <td>0.87</td>	$\mathbf{Religious}^d$	0.13	0.34	659	0.13	0.34	548	0.14	0.34	111	0.87
Atheist 0.37 0.48 643 0.38 0.49 533 0.36 0.48 110 0.79 Agostic 0.09 0.28 643 0.08 0.28 533 0.10 0.30 110 0.55 Roman-Catholic 0.12 0.32 643 0.12 0.32 533 0.14 0.34 110 0.56 Protestant 0.18 0.38 643 0.18 0.38 533 0.17 0.38 110 0.93 Other Christian 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.12 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.12 0.33 533 0.12 0.32 110 0.91 Big Five (scale 1-7): $Extraversion$ 3.21 1.62 626 3.18 1.60 520 3.32 1.73 106 0.45 Agreeableness 5.02 1.23 628 5.04 1.24 523 4.91 1.20 105 0.34 Conscientiousness 5.11 1.29 630 5.13 1.30 525 4.97 1.24 105 0.23 Emotional stability 4.65 1.52 638 4.62 1.53 531 4.79 1.49 107 0.30 Openness 4.70 1.29 640 <	Religion:	0.20	0.0-		0.20	0.0-	0 - 0	0.2.2	0.0-		0.01
Agostic 0.09 0.28 643 0.08 0.28 533 0.10 0.30 110 0.55 Roman-Catholic 0.12 0.32 643 0.12 0.32 533 0.14 0.34 110 0.56 Protestant 0.18 0.38 643 0.12 0.32 533 0.14 0.34 110 0.56 Protestant 0.18 0.38 643 0.12 0.32 533 0.17 0.38 110 0.56 Other Christian 0.12 0.33 643 0.13 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.12 0.33 533 0.11 0.31 110 0.63 Other 0.12 0.33 643 0.12 0.33 533 0.12 0.32 110 0.91 Big Five (scale 1–7):Extraversion 3.21 1.62 626 3.18 1.60 520 3.32 1.73 106 0.45 Agreeableness 5.02 1.23 628 5.04 1.24 523 4.91 1.20 105 0.34 Conscientiousness 5.11 1.29 630 5.13 1.30 525 4.97 1.24 105 0.23 Emotional stability 4.65 1.52 638 4.62 1.53 531 4.79 1.49 107 0.30 Openness 4.70 1.29 640	Atheist	0.37	0.48	643	0.38	0 49	533	0.36	0.48	110	0.79
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Agostic	0.09	0.28	643	0.08	0.28	533	0.10	0.30	110	0.55
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Roman-Catholic	0.12	0.32	643	0.12	0.32	533	0.14	0.34	110	0.56
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Protestant	0.12	0.38	643	0.12	0.38	533	0.17	0.38	110	0.93
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Other Christian	0.12	0.33	643	0.13	0.33	533	0.11	0.31	110	0.63
Big Five (scale 1-7): 6.62 6.62 6.62 6.63 6.63 6.63 6.62 6.62 6.62 6.62 6.63 6.63 6.62 6.6	Other	0.12	0.33	643	0.12	0.33	533	0.12	0.32	110	0.00
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Big Five (scale 1-7)	0.12	0.00	010	0.12	0.00	000	0.12	0.02	110	0.01
Agreeableness5.021.236285.041.245234.911.201050.34Conscientiousness5.111.296305.131.305254.971.241050.23Emotional stability4.651.526384.621.535314.791.491070.30Openness4.701.296404.671.295324.831.321080.25Risk pref. (scale 1-6)4.001.796674.061.785543.651.791130.03Task order ^e 0.510.506710.520.505580.460.501130.22	Extraversion	3 21	1.62	626	3 18	1.60	520	3 32	1 73	106	0.45
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Agreeableness	5.02	1.02	628	5.10	1.00 1.24	523	4 91	1.70	105	0.34
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Conscientiousness	5.11	1.29	630	5.13	1.30	525	4 97	1.20	105	0.23
Dependence 4.70 1.29 640 4.67 1.29 532 4.83 1.32 108 0.25 Risk pref. (scale 1-6) 4.00 1.79 667 4.06 1.78 554 3.65 1.79 113 0.03 Task order ^e 0.51 0.50 671 0.52 0.50 558 0.46 0.50 113 0.22	Emotional stability	4 65	1.20 1.52	638	4 62	1.53	531	4 79	1 49	107	0.30
Risk pref. (scale 1-6) 4.00 1.79 667 4.06 1.78 554 3.65 1.79 113 0.03 Task order ^e 0.51 0.50 671 0.52 0.50 558 0.46 0.50 113 0.22	Openness	4 70	1 29	640	4 67	1.00	532	4 83	1.10	108	0.25
Task order0.510.506710.520.505580.460.501130.22	Bisk pref (scale $1-6$)	4.00	1.20	667	4.06	1.20 1.78	554	3.65	1.02	113	0.20
	Task order ^{e}	0.51	0.50	671	0.52	0.50	558	0.00	0.50	113	0.00
Manipulation check questions (scale 1-5):	Manipulation check que	estions (s	cale 1-5	5).	0.02	0.00	000	0.10	0.00	110	0.22
Clarity f 4.56 0.68 663 4.58 0.67 551 4.46 0.70 112 0.11	Clarity ^f	4 56	83.0	663	4 58	0.67	551	4 46	0.70	112	0.11
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	A nonvmity ^g	4.00	0.00	663	4.00	0.07	551	4.40	0.70	112	0.11
The experiment h 4.04 0.03 6.60 4.04 0.02 5.04 4.01 0.02 111 0.72	Trust experiment ^h	4.40	0.13	660	4.43	0.12	5/0	4.40	0.13	111	0.42
Truct caperine $4.04 - 0.35 - 000 - 4.04 - 0.35 - 343 - 4.01 - 0.35 - 111 - 0.75$	Trust experiment	4.04	0.93	662	4.04	0.93	550	4.01	0.95	110	0.13
Deserving recipients j 4 47 0 81 661 4 48 0 82 549 4 42 0 70 112 0 49	Deserving recipients j	4 47	0.80	661	4.13	0.80	549	4 42	0.92	112	0.34

Notes: The last column reports *p*-values from comparing each variable across the two samples based on a χ^2 -test if the variable is binary and a *t*-test with unequal variances if the variable is not binary. ^{*a*} Has children under age 16 living in household. ^{*b*} Household income. ^{*c*} Works for a not-for-profit organization. ^{*d*} Frequently attends religious services. ^{*e*} 1 if the subject encountered the donation task after the questionnaire, 0 if before. ^{*f*} "The instructions, questions, and tasks in this survey were clear and easy to understand." ^{*g*} "The procedures followed in this experiment preserved your anonymity." ^{*h*} "The money you donated to the charity will be given to the charity." ^{*i*} "The charity will use the money to provide the chosen number of nutrition packages." ^{*j*} "The recipients of the donations are deserving of support."

Tr	eatments			Do	nation var	iable	
Condition	Nominal unit price (\$)	Effective unit price (\$)	Indivi- dual choice (units)	Net dona- tion (\$)	Charity receipt, uncond. (units)	Charity receipt, cond. (units)	Prob. of dona- tion
			(1)	(2)	(3)	(4)	(5)
A. Mean value	es (S.D.)						
No subsidy	0.50	0.50	$\begin{array}{c} 0.558 \\ (1.026) \end{array}$	$\begin{array}{c} 0.279 \\ (0.513) \end{array}$	$\begin{array}{c} 0.558 \\ (1.026) \end{array}$	$1.750 \\ (1.105)$	$\begin{array}{c} 0.319 \\ (0.468) \end{array}$
33% rebate	0.50	0.33	$\begin{array}{c} 0.867 \\ (1.278) \end{array}$	$\begin{array}{c} 0.286 \\ (0.422) \end{array}$	$\begin{array}{c} 0.867 \\ (1.278) \end{array}$	$2.130 \\ (1.147)$	$\begin{array}{c} 0.407 \\ (0.493) \end{array}$
1:2 match	0.50	0.33	$\begin{array}{c} 0.699 \\ (1.085) \end{array}$	$\begin{array}{c} 0.350 \\ (0.542) \end{array}$	$\begin{array}{c} 0.965 \ (1.614) \end{array}$	$2.535 \\ (1.695)$	$\begin{array}{c} 0.381 \\ (0.488) \end{array}$
33% discount	0.33	0.33	$\begin{array}{c} 0.982 \\ (1.547) \end{array}$	$\begin{array}{c} 0.324 \\ (0.510) \end{array}$	$\begin{array}{c} 0.982 \\ (1.547) \end{array}$	$2.362 \\ (1.580)$	$\begin{array}{c} 0.416 \\ (0.495) \end{array}$
50% rebate	0.50	0.25	$\begin{array}{c} 0.991 \\ (1.373) \end{array}$	$\begin{array}{c} 0.248 \ (0.343) \end{array}$	$\begin{array}{c} 0.991 \\ (1.373) \end{array}$	$2.196 \\ (1.233)$	$\begin{array}{c} 0.451 \\ (0.500) \end{array}$
1:1 match	0.50	0.25	$0.805 \\ (1.109)$	$0.403 \\ (0.554)$	1.611 (2.218)	3.434 (2.052)	$0.469 \\ (0.501)$
50% discount	0.25	0.25	$1.363 \\ (1.996)$	$\begin{array}{c} 0.341 \\ (0.499) \end{array}$	$1.363 \\ (1.996)$	2.906 (2.003)	$0.469 \\ (0.501)$
B. Tests of su	bsidy types:	<i>p</i> -values					
B1. At effective 33% rebate vs. 33% rebate vs. 1:2 match vs.	ve price of § . 1:2 match . 33% discou 33% discou	\$ <i>0.33</i> unt nt	$\begin{array}{c} 0.01 \\ 0.08 \\ 0.00 \end{array}$	$0.03 \\ 0.08 \\ 0.25$	$0.27 \\ 0.08 \\ 0.80$	$0.19 \\ 0.42 \\ 0.62$	$\begin{array}{c} 0.32 \\ 0.56 \\ 0.16 \end{array}$
B2. At effective 50% rebate vs. 50% rebate vs. 1:1 match vs.	ve price of § . 1:1 match . 50% discou 50% discour	\$ <i>0.25</i> unt nt	$0.04 \\ 0.00 \\ 0.00$	$0.00 \\ 0.00 \\ 0.05$	$\begin{array}{c} 0.00 \\ 0.00 \\ 0.05 \end{array}$	$0.00 \\ 0.03 \\ 0.18$	$0.56 \\ 0.59 \\ 1.00$
C. Tests of su	bsidized prie	ces: p-values					
50% vs. 33% r 1:1 vs. 1:2 mat 50% vs. 33% d	ebate tch liscount		$0.06 \\ 0.13 \\ 0.00$	$0.05 \\ 0.13 \\ 0.24$	$\begin{array}{c} 0.06 \\ 0.00 \\ 0.00 \end{array}$	$0.79 \\ 0.02 \\ 0.13$	$\begin{array}{c} 0.10 \\ 0.01 \\ 0.06 \end{array}$
D. Tests of su	bsidized vs.	unsubsidized	prices: p-u	values			
D1. Low subsi 33% rebate vs. 1:2 match vs. 33% discount	<i>dy rate</i> no subsidy no subsidy vs. no subsi	dy	$\begin{array}{c} 0.00 \\ 0.01 \\ 0.00 \end{array}$	$0.78 \\ 0.01 \\ 0.02$	$\begin{array}{c} 0.00 \\ 0.00 \\ 0.00 \end{array}$	$0.13 \\ 0.02 \\ 0.04$	$0.00 \\ 0.02 \\ 0.00$
D2. High subs 50% rebate vs. 1:1 match vs. 50% discount	<i>idy rate</i> . no subsidy no subsidy vs. no subsi	dv	$0.00 \\ 0.00 \\ 0.00$	$0.32 \\ 0.00 \\ 0.01$	$\begin{array}{c} 0.00 \\ 0.00 \\ 0.00 \end{array}$	$0.08 \\ 0.00 \\ 0.00$	$\begin{array}{c} 0.00 \\ 0.00 \\ 0.00 \end{array}$

Table C.6: Descriptive Results, Within-Subjects Design

Notes: Panel A shows mean values of the donation variables for each treatment (standard deviations in parentheses). Column 1 reports the number of packages that subjects selected to give at the nominal price. Column 2 shows the net dollar contribution implied by subjects' choices, i.e., column 1 evaluated at the nominal price minus the rebate (if any). Column 3 reports the overall number of packages received by the charity, i.e., column 1 plus matched units (if any). Column 4 reports the same measure as column 3 but conditional on giving (intensive margin). Column 5 reports the share of subjects who donated at least one package (extensive margin). Panels B to D show pairwise tests between treatment conditions. Panel B compares subsidy types conditional on the effective price. Panel C compares the two subsidized prices, \$0.25 and \$0.33, conditional on subsidy type. Panel D compares the unsubsidized price with the subsidized price arising from the low subsidy rate for each subsidy type. In panels B to D, columns 1 to 3 report *p*-values of two-tailed paired *t*-tests, column 4 reports *p*-values of two-tailed unpaired *t*-tests with unequal variances, and column 5 reports *p*-values of McNemar's χ^2 tests for paired binary data.

Appendix to Chapter 4 D

D.1 Additional Tables

	Table D.1: Statements by Treatment Group (Translated)	
--	---	--

Statement	Group Positive	Group Negative
(1)	Politicians do not always have the re- quired expertise to make reasonable de- cisions. Companies and organizations share their know-how with politicians and thereby facilitate better decision making. This makes it possible, for ex- ample, to implement more realistic en- vironmental regulations.	Lobbyists only represent their own in- terests and do not consider the well- being of the general public. For ex- ample, many companies try to use lob- bying to prevent stricter environmental regulations.
(2)	Not only companies lobby. Nonprofit organizations also use lobbying to in- fluence political decisions and to imple- ment their own and their members' in- terests. For example, an environmen- tal organization can try to implement a stricter climate policy by using lobby- ing.	Big companies have the required re- sources available to lobby. In contrast, nonprofit organizations and citizens of- ten lack such resources. This results in a climate policy on behalf of companies.
(3)	Lobbying is getting more transparent. For example, on January 31, 2019, the EU parliament decided to imple- ment stricter lobbying rules to improve transparency. Therefore, citizens will have sufficient information to under- stand lobbying activities and the im- pact of different actors.	Lobbying is intransparent. Citizens do not have enough information to under- stand the lobbying activities and the in- fluence of different actors on political decision making.

 Table D.2:
 Contribution Decision (Translated)

Screen	Content
(1)	For answering this survey you will receive 8 Euro, instead of the usual 4 Euro. This means you get additional 4 Euro . On the next screen, you have the opportunity to donate part or all of this additional 4 Euro to the nonprofit climate protection organization atmosfair. atmosfair supports projects for climate protection – e.g., the installation of renewable energies in developing countries – to reduce CO_2 emissions and fight climate change. In 2018, atmosfair received the quality valuation "excellent" from Stiftung Warentest – Finanztest.
(2)	Please enter the amount you would like to donate to the climate protection organization atmosfair below: [answer field] Euro
	[Help text:] The part of the 4 Euro that you do not donate to atmos- fair will be transferred onto your study account and paid out according to your selected mode of payment in April 2020. The part of the 4 Euro that you donate to atmosfair will be transferred to atmosfair in April 2020. atmosfair will issue a certificate for the total amount of donations received.

	I ADJE D.O. VAHADJES	
Name	Description	Based on
A. General		
Female	Whether individual is female.	gender_19 (wave 44)
Age	Calculated based on the midpoint of the available year of birth categories.	year_of_birth_cat_19 (wave 44)
Married	Whether individual is married.	marital_status_19 (wave 44)
Employment	Employment status of individual. Categories: (1) full-time employed, (2) part-time employed, (3) retired, (4) other.	occupation_19 (wave 44)
Internet usage	Whether individual uses internet more than once a week.	Internet_usage_19 (wave 44)
Recruited 2018	Whether individual was recruited for GIP in 2018.	sample (wave 44)
Payment mode	The mode in which cumulative earnings were paid out in October 2019. Categories: (1) donation, (2) Amazon voucher, (3) bank transfer.	art201910 (on-site access)
Pro-environmental	Whether the environment and climate protection are important or very important to the individual.	CE41048 (wave 41)
Income	Personal income of individual. Categories: (1) $< \in 1000, (2) \ [\in 1000, \in 2000), (3) \ [\in 2000, \in 3000), (4) \ge \in 3000.$	AA43054 (wave 43)
$B. \ Beliefs$		
Belief 2018	Answer to the question "How does lobbying affect the level of climate protection in	BG38009 (wave 38)

Table D.3: Variables

continued on the next page ...

the European Union?" in November 2018. For more details see Belief 2019.

D Appendix to Chapter 4

Belief 2019	Answer to the question "How does lobbying affect the level of climate protection in the European Union? Lobbying leads to" in the survey experiment (November 2019). Categories: (2) much more climate protection, (1) rather more, (0) neither more nor less, (-1) rather less, (-2) much less climate protection, (-) I don't know. "I don't know" is coded as (0), except when the full distribution of responses is shown.	BG44009 (wave 44)
Belief 2020	Answer to the question "How does lobbying affect the level of climate protection in the European Union? Lobbying leads to" in the follow-up survey (July 2020). For more details see Belief 2019.	BG48009 (wave 48)
'. Pro-environmental b	chavior	
Contribution to CO ₂ reduction	The amount of money (in euro) donated to atmosfair. Possible values: 0, 1, 2, 3, 4.	BG44017 (wave 44)
Total environmental donation	The sum of money (in euro) donated to atmosfair (choice in the survey experiment) and WWF (as part of the payment of the cumulative earnings in April 2020)	BG44017 (wave 44) art201910, betr202000 betr202004.mcs, and win202004 (on-site access)
Car usage	Number of days per week individual usually uses a car (self-reported).	BG48023 (wave 48)
Taking the plane	Whether individual took a plane at least once within the last six months (self-reported).	BG48024_h (wave 48
Purchasing local, organic products	Whether the individual purchased local, organic products at least once within the last six months (self-reported).	BG48024-g (wave 48

166

Appendices
continued		
Bringing own shop- ping bag	Whether individual brought his/her own bag for shopping at least once within the last six months (self-reported).	BG48024_d (wave 48)
Considering product sustainability	Whether the individual considered a product's sustainability in a pruchasing decision at least once within the last six months (self-reported).	BG48024_a (wave 48)
Donating	Whether individual donated to an environmental organization at least once within the last six months (self-reported).	BG48024.f (wave 48)
Volunteering	Whether individual volunteered for an environmental project at least once within the last six months (self-reported).	BG48024_b (wave 48)
Protesting	Whether individual protested for more environmental or climate protection at least once within the last six months (self-reported).	BG48024_c (wave 48)
Signing petition	Whether individual signed a petition for more environmental or climate protection at least once within the last six months (self-reported).	BG48024_e (wave 48)

	Belief 2019			Belief 2020		
Variable	(1)	(2)	(3)	(4)	(5)	(6)
Group Positive ^{a}	0.44***	0.44***	0.42***	0.04	0.02	0.01
	(0.05)	(0.06)	(0.06)	(0.04)	(0.04)	(0.05)
Group Negative ^{b}	0.06	0.07^{*}	0.08^{*}	0.05	0.05	0.04
	(0.04)	(0.04)	(0.05)	(0.04)	(0.04)	(0.05)
p-value $(a = b)$	0.00	0.00	0.00	0.63	0.56	0.61
Belief 2018	Yes	Yes	Yes	Yes	Yes	Yes
Female		Yes	Yes		Yes	Yes
Age		Yes	Yes		Yes	Yes
Married		Yes	Yes		Yes	Yes
Employment status		Yes	Yes		Yes	Yes
Online status		Yes	Yes		Yes	Yes
Recruited 2018		Yes	Yes		Yes	Yes
Income			Yes			Yes
Pro-environmental			Yes			Yes
Observations	3071	3041	2514	2819	2802	2359

Table D.4: Estimation Results – Beliefs

Notes: The table reports the average marginal effects from ordered logit regressions. In columns 1 to 3, the dependent variable is the response to the belief elicitation question in November 2019. In columns 4 to 6, the dependent variable is the response to the belief elicitation question in July 2020. In both cases, the answer categories are (2) "much more climate protection," (1) "rather more," (0) "neither more nor less," (-1) "rather less," and (-2) "much less climate protection." The response "I don't know" is treated as neither expecting lobbying to increase nor to decrease the level of climate protection. The average marginal effects are calulcated based on the predicted beliefs if everyone in the sample would be assigned to the indicated treatment instead of the control group. Standard errors are in parentheses. $*p < 0.1, \ **p < 0.05, \ ***p < 0.01.$

	Treatment		
Variable	Group Positive	Group Negative	$\begin{array}{c} \text{Comparison} \\ p\text{-value} \end{array}$
A. Observed behavior			
Contribution to CO_2 reduction	$2.40 \\ (1.84)$	$2.30 \\ (1.85)$	0.22
Total environmental donation	3.05 (2.84)	3.06 (2.87)	0.94
Observations	1031	1026	
B. Stated behavior			
Car usage (days per week) ^{a}	$3.65 \\ (2.26)$	$3.78 \\ (2.26)$	0.22
Taking a plane	$0.04 \\ (0.18)$	$\begin{array}{c} 0.07 \\ (0.26) \end{array}$	0.00
Purchasing local, organic products	0.80 (0.40)	0.78 (0.42)	0.25
Bringing own shopping bag	0.95 (0.21)	0.94 (0.24)	0.16
Considering product sustainability	0.68 (0.47)	0.66 (0.47)	0.50
Donating	0.11 (0.31)	0.10 (0.30)	0.47
Volunteering	0.04 (0.21)	0.04 (0.20)	0.89
Protesting	0.03 (0.18)	0.03 (0.16)	0.34
Signing petition	0.14 (0.35)	0.13 (0.34)	0.44
Stated behavior $index^a$	0.00 (0.43)	-0.06 (0.45)	0.00
Observations	940	944	

Table D.5: Summary Statistics – Outcome Variables

Notes: The table reports the averages by treatment group with standard deviations in parentheses. The last column presents p-values from comparing each variable across the two treatment groups based on a χ^2 -test if the variable is binary and a t-test with unequal variances otherwise.

 $^a\mathrm{Based}$ on 935 and 942 observations in Group Positive and Group Negative, respectively.

D.2 Additional Figures



How does lobbying affect the level of climate protection in the European Union? Lobbying leads to ...

Figure D.1: Distribution of Beliefs in 2019

Notes: The figure shows the distribution of responses to the belief elicitation question in the survey experiment (November 2019) by experimental group. The (translated) wording of the belief elicitation question is presented at the top of the figure.





Notes: The figure shows the distribution of responses to the belief elicitation question in the follow-up survey (July 2020) by experimental group. The (translated) wording of the belief elicitation question is presented at the top of the figure.



Panel A. Much less climate protection (N = 453)

Figure D.3: Beliefs over Time by Baseline Belief

(July 2020)

Notes: Panels A to E show the average likert score for the response to the belief elicitation question of the survey experiment (November 2019) by experimental group for individuals with a given baseline belief. The likert scale ranges from -2 (lobbying leads to much less climate protection) to 2 (lobbying leads to much more climate protection). The response "I don't know" is treated as neither expecting a positive nor a negative impact. The error bars represent 95 percent confidence intervals. Panel E does not contain error bars since only 46 individuals hold such a baseline belief.

(November 2018)

(November 2019)

Date

Bibliography

- Adena, Maja, Jeyhun Alizade, Frauke Bohner, Julian Harke, and Fabio Mesters. 2019. "Quality Certification for Nonprofits, Charitable Giving, and Donor's Trust: Experimental Evidence." Journal of Economic Behavior & Organization 159: 75–100.
- Adena, Maja, Steffen Huck, and Imran Rasul. 2014. "Charitable Giving and Nonbinding Contribution-Level Suggestions – Evidence from a Field Experiment." *Review of Behavioral Economics* 1 (3): 275–293.
- Albrecht, Konstanze, Florentin Krämer, and Nora Szech. 2017. "Animal Welfare and Human Ethics: A Personality Study." CESifo Working Paper No. 6609.
- Allcott, Hunt. 2011. "Social Norms and Energy Conservation." Journal of Public Economics 95 (9–10): 1082–1095.
- Allcott, Hunt, and Todd Rogers. 2014. "The Short-Run and Long-Run Effects of Behavioral Interventions: Experimental Evidence from Energy Conservation." *American Economic Review* 104 (10): 3003–3037.
- Allcott, Hunt, and Dmitry Taubinsky. 2015. "Evaluating Behaviorally Motivated Policy: Experimental Evidence from the Lightbulb Market." American Economic Review 105 (8): 2501–2538.
- Altmann, Steffen, Armin Falk, Paul Heidhues, Rajshri Jayaraman, and Marrit Teirlinck. 2019. "Defaults and Donations: Evidence from a Field Experiment." *Review of Economics and Statistics* 101 (5): 808–826.
- Andersen, Steffen, Glenn W. Harrison, Morten Igel Lau, and E. Elisabet Rutström. 2006. "Elicitation Using Multiple Price List Formats." *Experimental Economics* 9 (4): 383–405.
- Andor, Mark A, Manuel Frondel, and Stephan Sommer. 2018. "Equity and the Willingness to Pay for Green Electricity in Germany." *Nature Energy* 3 (10): 876–881.
- Andor, Mark A., Andreas Gerster, and Jörg Peters. 2020. "Information Campaigns for Residential Energy Conservation." Ruhr Economic Papers No. 871.
- Andreoni, James. 1988. "Privately Provided Public Goods in a Large Economy: The Limits of Altruism." Journal of Public Economics 35 (1): 57–73.

- Andreoni, James. 1989. "Giving with Impure Altruism: Applications to Charity and Ricardian Equivalence." *Journal of Political Economy* 97 (6): 1447–1458.
- Arrow, Kenneth J. 1996. "The Economics of Information: An Exposition." Empirica 23 (2): 119–128.
- Banakou, Domna, Raphaela Groten, and Mel Slater. 2013. "Illusory Ownership of a Virtual Child Body Causes Overestimation of Object Sizes and Implicit Attitude Changes." *Proceedings of the National Academy of Sciences* 110 (31): 12846–12851.
- Banakou, Domna, Parasuram D. Hanumanthu, and Mel Slater. 2016. "Virtual Embodiment of White People in a Black Virtual Body Leads to a Sustained Reduction in Their Implicit Racial Bias." *Frontiers in Human Neuroscience* 10: 601.
- Bastian, Brock, and Steve Loughnan. 2017. "Resolving the Meat-Paradox: A Motivational Account of Morally Troublesome Behavior and Its Maintenance." *Personality and Social Psychology Review* 21 (3): 278–299.
- Bekkers, René. 2015. "When and Why Matches Are More Effective Subsidies than Rebates." In *Replication in Experimental Economics*, edited by Deck, Cary A., Enrique Fatas, Tanya Rosenblat, R. Mark Isaac, and Douglas A. Norton, Volume 18 of Research in Experimental Economics: 183–211. Bingley: Emerald Group Publishing.
- Bekkers, René, and Pamala Wiepking. 2011. "A Literature Review of Empirical Studies of Philanthropy: Eight Mechanisms That Drive Charitable Giving." Nonprofit and Voluntary Sector Quarterly 40 (5): 924–973.
- Bell, E., F. Bailey Norwood, and Jayson L. Lusk. 2017. "Are Consumers Wilfully Ignorant About Animal Welfare." Animal Welfare 26 (4): 399–402.
- Bergstrom, Theodore, Lawrence Blume, and Hal Varian. 1986. "On the Private Provision of Public Goods." *Journal of Public Economics* 29 (1): 25–49.
- Bertrand, Marianne, Matilde Bombardini, Raymond Fisman, and Francesco Trebbi. 2020. "Tax-Exempt Lobbying: Corporate Philanthropy as a Tool for Political Influence." *American Economic Review* 110 (7): 2065–2102.
- Binder, Martin, Ann-Kathrin Blankenberg, and Jorge Guardiola. 2020. "Does It Have To Be a Sacrifice? Different Notions of the Good Life, Pro-Environmental Behavior and Their Heterogeneous Impact on Well-Being." *Ecological Economics* 167: 106448.
- Blom, Annelies G, Christina Gathmann, and Ulrich Krieger. 2015. "Setting Up an Online Panel Representative of the General Population: The German Internet Panel." *Field Methods* 27 (4): 391–408.
- Blom, Annelies G., Marina Fikel, Sabine Friedel, Jan Karem Höhne, Ulrich Krieger, Tobias Rettig, Alexander Wenz, SFB 884 "Political Economy of Reforms", University of Mannheim. 2019. "German Internet Panel, Wave 38 (November 2018)." GESIS Data Archive, Cologne. ZA6958 Data file Version 1.0.0, https://doi.org/10.4232/1.13391.

- Blom, Annelies G., Marina Fikel, Sabine Friedel, Jan Karem Höhne, Ulrich Krieger, Tobias Rettig, Alexander Wenz, SFB 884 "Political Economy of Reforms", University of Mannheim. 2020a. "German Internet Panel, Wave 41 (May 2019)." GESIS Data Archive, Cologne. ZA7590 Data file Version 1.0.0 https://doi.org/10.4232/1.13464.
- Blom, Annelies G., Marina Fikel, Sabine Friedel, Jan Karem Höhne, Ulrich Krieger, Tobias Rettig, Alexander Wenz, SFB 884 "Political Economy of Reforms", University of Mannheim. 2020b. "German Internet Panel, Wave 44 (November 2019)." GESIS Data Archive, Cologne. ZA7593 Data file Version 1.1.0, https://doi.org/10.4232/1.13614.
- Blom, Annelies G., Marina Fikel, Sabine Friedel, Jan Karem Höhne, Ulrich Krieger, Tobias Rettig, Alexander Wenz, SFB 884 "Political Economy of Reforms", University of Mannheim. 2021a. "German Internet Panel, Wave 43 Core Study (September 2019)." GESIS Data Archive, Cologne. ZA7592 Data file Version 2.1.0, https://doi.org/10.4232/1.13673.
- Blom, Annelies G., Marina Fikel, Sabine Friedel, Ulrich Krieger, Tobias Rettig, SFB 884 "Political Economy of Reforms", University of Mannheim. 2021b. "German Internet Panel, Wave 48 (July 2020)." GESIS Data Archive, Cologne. ZA7645 Data file Version 1.0.0, https://doi.org/10.4232/1.13681.
- Bock, Olaf, Ingmar Baetge, and Andreas Nicklisch. 2014. "hroot: Hamburg Registration and Organization Online Tool." *European Economic Review* 71: 117–120.
- Cain, Daylian, and Jason Dana. 2012. "Paying People to Look at the Consequences of Their Actions." Working Paper.
- Carneiro, Pedro, James J. Heckman, and Edward J. Vytlacil. 2011. "Estimating Marginal Returns to Education." American Economic Review 101 (6): 2754–2781.
- Cartwright, Edward J, and Zarak Mirza. 2019. "Charitable Giving When Donors Are Constrained to Give a Minimum Amount." Oxford Economic Papers.
- Chao, Yu-Long, and San-Pui Lam. 2011. "Measuring Responsible Environmental Behavior: Self-Reported and Other-Reported Measures and Their Differences in Testing a Behavioral Model." *Environment and Behavior* 43 (1): 53–71.
- Charness, Gary, Uri Gneezy, and Michael A. Kuhn. 2012. "Experimental Methods: Between-Subject and Within-Subject Design." Journal of Economic Behavior & Organization 81 (1): 1–8.
- Chater, Nick, and George Loewenstein. 2016. "The Under-Appreciated Drive for Sense-Making." Journal of Economic Behavior & Organization 126: 137–154.

- Chen, Daniel L., Martin Schonger, and Chris Wickens. 2016. "oTree An Open-Source Platform for Laboratory, Online, and Field Experiments." *Journal of Behavioral and Experimental Finance* 9: 88–97.
- Chen, Haipeng, Howard Marmorstein, Michael Tsiros, and Akshay R. Rao. 2012. "When More Is Less: The Impact of Base Value Neglect on Consumer Preferences for Bonus Packs over Price Discounts." *Journal of Marketing* 76 (4): 64–77.
- Cornelissen, Thomas, Christian Dustmann, Anna Raute, and Uta Schönberg. 2018. "Who Benefits from Universal Child Care? Estimating Marginal Returns to Early Child Care Attendance." *Journal of Political Economy* 126 (6): 2356–2409.
- Croson, Rachel, and Jen Yue Shang. 2008. "The Impact of Downward Social Information on Contribution Decisions." *Experimental Economics* 11 (3): 221–233.
- Cryder, Cynthia E., George Loewenstein, and Richard Scheines. 2013. "The Donor Is in the Details." Organizational Behavior and Human Decision Processes 120 (1): 15–23.
- d'Adda, Giovanna, Yu Gao, Russell Golman, and Massimo Tavoni. 2018. "It's so Hot in Here: Information Avoidance, Moral Wiggle Room, and High Air Conditioning Usage." FEEM Working Paper No. 007.2018.
- Dana, Jason, Roberto A. Weber, and Jason Xi Kuang. 2007. "Exploiting Moral Wiggle Room: Experiments Demonstrating an Illusory Preference for Fairness." *Economic Theory* 33 (1): 67–80.
- Dave, Chetan, Catherine C. Eckel, Cathleen A. Johnson, and Christian Rojas. 2010. "Eliciting Risk Preferences: When Is Simple Better?" Journal of Risk and Uncertainty 41 (3): 219–243.
- **Davis, Douglas D.** 2006. "Rebate Subsidies, Matching Subsidies and Isolation Effects." Judgment and Decision Making 1 (1): 13–22.
- Davis, Douglas D., and Edward L. Millner. 2005. "Rebates, Matches, and Consumer Behavior." *Southern Economic Journal* 72 (2): 410–421.
- **Davis, Douglas D., Edward L. Millner, and Robert J. Reilly.** 2005. "Subsidy Schemes and Charitable Contributions: A Closer Look." *Experimental Economics* 8 (2): 85–106.
- Diederich, Johannes, Catherine C. Eckel, Raphael Epperson, Timo Goeschl, and Philip J. Grossman. 2020. "Subsidizing Unit Donations: Matches, Rebates, and Discounts Compared." AWI Discussion Paper No. 697.
- Diederich, Johannes, Raphael Epperson, and Timo Goeschl. 2021. "How to Design the Ask? Funding Units vs. Giving Money." AWI Discussion Paper No. 698.

- **Diederich, Johannes, and Timo Goeschl.** 2014. "Willingness to Pay for Voluntary Climate Action and Its Determinants: Field-Experimental Evidence." *Environmental and Resource Economics* 57 (3): 405–429.
- Diederich, Johannes, and Timo Goeschl. 2017. "To Mitigate or Not to Mitigate: The Price Elasticity of Pro-environmental Behavior." *Journal of Environmental Economics and Management* 84: 209–222.
- Diederich, Johannes, and Timo Goeschl. 2018. "Voluntary Action for Climate Change Mitigation Does Not Exhibit Locational Preferences." Journal of Environmental Economics and Management 90: 175–180.
- Duncan, Brian. 2004. "A Theory of Impact Philanthropy." Journal of Public Economics 88 (9): 2159–2180.
- Eckel, Catherine C., and Philip J. Grossman. 1996. "Altruism in Anonymous Dictator Games." *Games and Economic Behavior* 16 (2): 181–191.
- Eckel, Catherine C., and Philip J. Grossman. 2002. "Sex Differences and Statistical Stereotyping in Attitudes toward Financial Risk." *Evolution and Human Behavior* 23 (4): 281–295.
- Eckel, Catherine C., and Philip J. Grossman. 2003. "Rebate versus Matching: Does How We Subsidize Charitable Contributions Matter?" Journal of Public Economics 87 (3-4): 681–701.
- Eckel, Catherine C, and Philip J Grossman. 2006a. "Do Donors Care about Subsidy Type? An Experimental Study." In *Experiments Investigating Fundraising* and Charitable Contributors, edited by Davis, Douglas D., and R. Mark Isaac, Volume 11 of Research in Experimental Economics: 157–175. Bingley: Emerald Group Publishing Limited.
- Eckel, Catherine C., and Philip J. Grossman. 2006b. "Subsidizing Charitable Giving with Rebates or Matching: Further Laboratory Evidence." Southern Economic Journal 72 (4): 794–807.
- Eckel, Catherine C., and Philip J. Grossman. 2008a. "Subsidizing Charitable Contributions: A Natural Field Experiment Comparing Matching and Rebate Subsidies." *Experimental Economics* 11 (3): 234–252.
- Eckel, Catherine C., and Philip J. Grossman. 2008b. "Forecasting Risk Attitudes: An Experimental Study Using Actual and Forecast Gamble Choices." *Journal* of Economic Behavior & Organization 68 (1): 1–17.
- Eckel, Catherine C., and Philip J. Grossman. 2017. "Comparing Rebate and Matching Subsidies Controlling for Donors' Awareness: Evidence from the Field." *Journal of Behavioral and Experimental Economics* 66: 88–95.

- Eckel, Catherine, Philip J. Grossman, and Angela Milano. 2007. "Is More Information Always Better? An Experimental Study of Charitable Giving and Hurricane Katrina." *Southern Economic Journal* 74 (2): 388–411.
- Edwards, James T, and John A List. 2014. "Toward an Understanding of Why Suggestions Work in Charitable Fundraising: Theory and Evidence from a Natural Field Experiment." *Journal of Public Economics* 114: 1–13.
- Ehrhart, Mark G., Karen Holcombe Ehrhart, Scott C. Roesch, Beth G. Chung-Herrera, Kristy Nadler, and Kelsey Bradshaw. 2009. "Testing the Latent Factor Structure and Construct Validity of the Ten-Item Personality Inventory." Personality and Individual Differences 47 (8): 900–905.
- Epperson, Raphael, and Andreas Gerster. 2020. "Information Preferences and the Short and Long Run Effects of Information." *AEA RCT Registry*. November 13. https://doi.org/10.1257/rct.5015-1.3.
- Epperson, Raphael, Wolfgang Habla, and Ulrich Wagner. 2019. "Wie der Einfluss von Lobbyismus auf die Politik in Deutschland und der EU wahrgenommen wird." ZEW expert brief 19-04. Mannheim: ZEW – Leibniz Centre for European Economic Research.
- Epperson, Raphael, and Christiane Reif. 2019. "Matching Subsidies and Voluntary Contributions: A Review." Journal of Economic Surveys 33 (5): 1578–1601.
- European Comission. 2020. "Agreement on a Mandatory Transparency Register." European Commission, December 15, 2020. https://ec.europa.eu/commission/pres scorner/detail/en/ip_20_2425.
- Exley, Christine L., and Judd B. Kessler. 2021. "Information Avoidance and Image Concerns." NBER Working Paper No. 28376.
- Falk, Armin, Thomas Neuber, and Nora Szech. 2020. "Diffusion of Being Pivotal and Immoral Outcomes." *The Review of Economic Studies* 87 (5): 2205–2229.
- Falk, Armin, and Nora Szech. 2013. "Morals and Markets." *Science* 340 (6133): 707–711.
- Farrow, Katherine, Gilles Grolleau, and Lisette Ibanez. 2017. "Social Norms and Pro-environmental Behavior: A Review of the Evidence." *Ecological Economics* 140: 1–13.
- Feiler, Lauren. 2014. "Testing Models of Information Avoidance with Binary Choice Dictator Games." Journal of Economic Psychology 45: 253–267.
- Felgendreher, Simon. 2018. "Do Consumers Choose to Stay Ignorant? The Role of Information in the Purchase of Ethically Certified Products." Working Paper in Economics No. 717, University of Gothenburg.

- Ferraro, Paul J., Juan Jose Miranda, and Michael K. Price. 2011. "The Persistence of Treatment Effects with Norm-Based Policy Instruments: Evidence from a Randomized Environmental Policy Experiment." *American Economic Review* 101 (3): 318–322.
- Ferraro, Paul J., and Michael K. Price. 2013. "Using Nonpecuniary Strategies to Influence Behavior: Evidence from a Large-Scale Field Experiment." *The Review of Economics and Statistics* 95 (1): 64–73.
- de Figueiredo, John M., and Brian Kelleher Richter. 2014. "Advancing the Empirical Research on Lobbying." Annual Review of Political Science 17: 163–185.
- de Figueiredo, John M., and Brian S. Silverman. 2006. "Academic Earmarks and the Returns to Lobbying." *The Journal of Law and Economics* 49 (2): 597–625.
- Fischbacher, Urs, and Simon Gächter. 2010. "Social Preferences, Beliefs, and the Dynamics of Free Riding in Public Goods Experiments." *American Economic Review* 100 (1): 541–556.
- Fischbacher, Urs, Simon Gächter, and Ernst Fehr. 2001. "Are People Conditionally Cooperative? Evidence from a Public Goods Experiment." *Economics Letters* 71 (3): 397–404.
- **Freddi, Eleonora.** Forthcoming. "Do People Avoid Morally Relevant Information? Evidence from the Refugee Crisis." *The Review of Economics and Statistics*.
- Gandullia, Luca. 2019. "The Price Elasticity of Warm-Glow Giving." *Economics* Letters 182: 30–32.
- Gandullia, Luca, and Emanuela Lezzi. 2018. "The Price Elasticity of Charitable Giving: New Experimental Evidence." *Economics Letters* 173: 88–91.
- Gawande, Kishore, Pravin Krishna, and Michael J. Robbins. 2006. "Foreign Lobbies and US Trade Policy." *Review of Economics and Statistics* 88 (3): 563–571.
- Gerber, Pierre J., Henning Steinfeld, Benjamin Henderson, Anne Mottet, Carolyn Opio, Jeroen Dijkman, Allessandra Falcucci, and Giuseppe Tempio. 2013. Tackling Climate Change through Livestock: A Global Assessment of Emissions and Mitigation Opportunities. Rome: Food and Agriculture Organization of the United Nations (FAO).
- Giger, Nathalie, and Heike Klüver. 2016. "Voting Against Your Constituents? How Lobbying Affects Representation." *American Journal of Political Science* 60 (1): 190–205.
- Girod, Bastien, Detlef Peter van Vuuren, and Edgar G. Hertwich. 2014. "Climate Policy through Changing Consumption Choices: Options and Obstacles for Reducing Greenhouse Gas Emissions." *Global Environmental Change* 25: 5–15.

- **Glewwe, Paul.** 1997. "A Test of the Normality Assumption in Ordered Probit Model." *Econometric Reviews* 16 (1): 1–19.
- Gneezy, Uri, Elizabeth A. Keenan, and Ayelet Gneezy. 2014. "Avoiding Overhead Aversion in Charity." *Science* 346 (6209): 632–635.
- Goeschl, Timo, Sara Elisa Kettner, Johannes Lohse, and Christiane Schwieren. 2020. "How Much Can We Learn about Voluntary Climate Action from Behavior in Public Goods Games?" *Ecological Economics* 171: 106591.
- Goldstein, Noah J., Robert B. Cialdini, and Vladas Griskevicius. 2008. "A Room with a Viewpoint: Using Social Norms to Motivate Environmental Conservation in Hotels." *Journal of Consumer Research* 35 (3): 472–482.
- Golman, Russel, George Loewenstein, Andras Molnar, and Silvia Saccardo. 2021. "The Demand for, and Avoidance of, Information." Working Paper.
- Golman, Russell, David Hagmann, and George Loewenstein. 2017. "Information Avoidance." Journal of Economic Literature 55 (1): 96–135.
- Gosling, Samuel D., Peter J. Rentfrow, and William B. Swann. 2003. "A Very Brief Measure of the Big-Five Personality Domains." *Journal of Research in Personality* 37 (6): 504–528.
- Goswami, Indranil, and Oleg Urminsky. 2016. "When Should the Ask Be a Nudge? The Effect of Default Amounts on Charitable Donations." *Journal of Marketing Research* 53 (5): 829–846.
- **Greiner, Ben.** 2015. "Subject Pool Recruitment Procedures: Organizing Experiments with ORSEE." Journal of the Economic Science Association 1 (1): 114–125.
- Grey, Felix. 2018. "Corporate Lobbying for Environmental Protection." Journal of Environmental Economics and Management 90: 23–40.
- Grieder, Manuel, Deborah Kistler, and Jan Schmitz. 2020. "The Hidden Benefits of Corporate Social Responsibility." Working Paper.
- **Grossman, Zachary.** 2014. "Strategic Ignorance and the Robustness of Social Preferences." *Management Science* 60 (11): 2659–2665.
- **Grossman, Zachary, and Joël J. van der Weele.** 2017. "Self-image and Willful Ignorance in Social Decisions." *Journal of the European Economic Association* 15 (1): 173–217.
- **Gullberg, Anne Therese.** 2008. "Lobbying Friends and Foes in Climate Policy: The Case of Business and Environmental Interest Groups in the European Union." *Energy Policy* 36 (8): 2964–2972.
- Hamby, Anne. 2016. "One For Me, One For You: Cause-Related Marketing with Buy-One Give-One Promotions." *Psychology & Marketing* 33 (9): 692–703.

- Handel, Benjamin, and Joshua Schwartzstein. 2018. "Frictions or Mental Gaps: What's behind the Information We (Don't) Use and When Do We Care?" Journal of Economic Perspectives 32 (1): 155–178.
- Hara, Kotaro, Abigail Adams, Kristy Milland, Saiph Savage, Chris Callison-Burch, and Jeffrey P. Bigham. 2018. "A Data-Driven Analysis of Workers' Earnings on Amazon Mechanical Turk." In Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems: 1–14. New York: Association for Computing Machinery.
- Hardesty, David M., and William O. Bearden. 2003. "Consumer Evaluations of Different Promotion Types and Price Presentations: The Moderating Role of Promotional Benefit Level." *Journal of Retailing* 79 (1): 17–25.
- Heckman, James J., and Edward Vytlacil. 2001. "Policy-Relevant Treatment Effects." American Economic Review 91 (2): 107–111.
- Helms McCarty, Sara, Timothy M. Diette, and Betsy Bugg Holloway. 2018. "How Low Can You Go? An Investigation into Matching Gifts in Fundraising." *Review of Behavioral Economics* 5 (1): 23–37.
- Hertwig, Ralph, and Christoph Engel. 2016. "Homo Ignorans: Deliberately Choosing Not to Know." *Perspectives on Psychological Science* 11 (3): 359–372.
- Hestermann, Nina, Yves Le Yaouanq, and Nicolas Treich. 2020. "An Economic Model of the Meat Paradox." *European Economic Review* 129: 103569.
- Hirano, Keisuke, Guido W. Imbens, and Geert Ridder. 2003. "Efficient Estimation of Average Treatment Effects Using the Estimated Propensity Score." *Econometrica* 71 (4): 1161–1189.
- Holm, Sture. 1979. "A Simple Sequentially Rejective Multiple Test Procedure." Scandinavian Journal of Statistics 6 (2): 65–70.
- Horton, John J., David G. Rand, and Richard J. Zeckhauser. 2011. "The Online Laboratory: Conducting Experiments in a Real Labor Market." *Experimental Economics* 14 (3): 399–425.
- Horvat, Nikola, Stanko Škec, Tomislav Martinec, Fanika Lukačević, and Marija Majda Perišić. 2019. "Comparing Virtual Reality and Desktop Interface for Reviewing 3D CAD Models." In Proceedings of the Design Society: International Conference on Engineering Design, Volume 1: 1923–1932. Cambridge University Press.
- Huck, Steffen, Imran Rasul, and Andrew Shephard. 2015. "Comparing Charitable Fundraising Schemes: Evidence from a Natural Field Experiment and a Structural Model." American Economic Journal: Economic Policy 7 (2): 326–369.

- Igan, Deniz, and Prachi Mishra. 2014. "Wall Street, Capitol Hill, and K Street: Political Influence and Financial Regulation." *The Journal of Law and Economics* 57 (4): 1063–1084.
- Imbens, Guido W., and Jeffrey M. Wooldridge. 2009. "Recent Developments in the Econometrics of Program Evaluation." Journal of Economic Literature 47 (1): 5–86.
- Ito, Koichiro, Takanori Ida, and Makoto Tanaka. 2018. "Moral Suasion and Economic Incentives: Field Experimental Evidence from Energy Demand." American Economic Journal: Economic Policy 10 (1): 240–267.
- Jalava, Mika, Matti Kummu, Miina Porkka, Stefan Siebert, and Olli Varis. 2014. "Diet Change–a Solution to Reduce Water Use?" Environmental Research Letters 9 (7): 074016.
- Kajackaite, Agne. 2015. "If I Close My Eyes, Nobody Will Get Hurt: The Effect of Ignorance on Performance in a Real-Effort Experiment." Journal of Economic Behavior & Organization 116: 518–524.
- Kamhöfer, Daniel A., Hendrik Schmitz, and Matthias Westphal. 2018. "Heterogeneity in Marginal Non-Monetary Returns to Higher Education." Journal of the European Economic Association 17 (1): 205–244.
- Kang, Joseph D. Y., and Joseph L. Schafer. 2007. "Demystifying Double Robustness: A Comparison of Alternative Strategies for Estimating a Population Mean from Incomplete Data." *Statistical Science* 22 (4): 523–539.
- Kang, Karam. 2016. "Policy Influence and Private Returns from Lobbying in the Energy Sector." *The Review of Economic Studies* 83 (1): 269–305.
- Karlan, Dean, and John A. List. 2007. "Does Price Matter in Charitable Giving? Evidence from a Large-Scale Natural Field Experiment." *American Economic Review* 97 (5): 1774–1793.
- Karlan, Dean, John A. List, and Eldar Shafir. 2011. "Small Matches and Charitable Giving: Evidence from a Natural Field Experiment." *Journal of Public Economics* 95 (5–6): 344–350.
- Karlan, Dean, and Daniel H. Wood. 2017. "The Effect of Effectiveness: Donor Response to Aid Effectiveness in a Direct Mail Fundraising Experiment." *Journal* of Behavioral and Experimental Economics 66: 1–8.
- Keser, Claudia, and Frans Van Winden. 2000. "Conditional Cooperation and Voluntary Contributions to Public Goods." *Scandinavian Journal of Economics* 102 (1): 23–39.
- Kesternich, Martin, Andreas Löschel, and Daniel Römer. 2016. "The Long-Term Impact of Matching and Rebate Subsidies When Public Goods Are Impure: Field Experimental Evidence from the Carbon Offsetting Market." *Journal of Public Economics* 137: 70–78.

- Kling, Jeffrey R., Jeffrey B. Liebman, and Lawrence F. Katz. 2007. "Experimental Analysis of Neighborhood Effects." *Econometrica* 75 (1): 83–119.
- Kormos, Christine, and Robert Gifford. 2014. "The Validity of Self-Report Measures of Proenvironmental Behavior: A Meta-Analytic Review." Journal of Environmental Psychology 40: 359–371.
- Kunst, Jonas R., and Sigrid M. Hohle. 2016. "Meat Eaters by Dissociation: How We Present, Prepare and Talk about Meat Increases Willingness to Eat Meat by Reducing Empathy and Disgust." Appetite 105: 758–774.
- Landry, Craig E., Andreas Lange, John A. List, Michael K. Price, and Nicholas G. Rupp. 2010. "Is a Donor in Hand Better Than Two in the Bush? Evidence from a Natural Field Experiment." *American Economic Review* 100 (3): 958–83.
- Lange, Florian, and Siegfried Dewitte. 2019. "Measuring Pro-Environmental Behavior: Review and Recommendations." Journal of Environmental Psychology 63: 92–100.
- Larson, Tara, and C. Monica Capra. 2009. "Exploiting Moral Wiggle Room: Illusory Preference for Fairness? A Comment." Judgment and Decision Making 4 (6): 467–474.
- Latour, Stephen A., and Ajay K. Manrai. 1989. "Interactive Impact of Informational and Normative Influence on Donations." *Journal of Marketing Research* 26 (3): 327–335.
- Lewis, Joshua, and Deborah Small. 2019. "Ineffective Altruism: Giving Less When Donations Do More Good." Working Paper.
- Lin-Schilstra, Li, and Arnout R. H. Fischer. 2020. "Consumer Moral Dilemma in the Choice of Animal-Friendly Meat Products." *Sustainability* 12 (12): 4844.
- Lind, Jo Thori, Karine Nyborg, and Anna Pauls. 2019. "Save the Planet or Close Your Eyes? Testing Strategic Ignorance in a Charity Context." *Ecological Economics* 161: 9–19.
- List, John A., and David Lucking-Reiley. 2002. "The Effects of Seed Money and Refunds on Charitable Giving: Experimental Evidence from a University Capital Campaign." Journal of Political Economy 110 (1): 215–233.
- Löschel, Andreas, Bodo Sturm, and Carsten Vogt. 2013. "The Demand for Climate Protection—Empirical Evidence From Germany." *Economics Letters* 118 (3): 415–418.
- Loughnan, Steve, Nick Haslam, and Brock Bastian. 2010. "The Role of Meat Consumption in the Denial of Moral Status and Mind to Meat Animals." Appetite 55 (1): 156–159.

- Lukas, Indra, Philip J Grossman, and Catherine Eckel. 2010. "Preference or Confusion: Understanding the Differential Impact of Rebate and Matching Subsidies." Working Paper.
- Lunceford, Jared K., and Marie Davidian. 2004. "Stratification and Weighting via the Propensity Score in Estimation of Causal Treatment Effects: A Comparative Study." *Statistics in Medicine* 23 (19): 2937–2960.
- Machovina, Brian, Kenneth J Feeley, and William J Ripple. 2015. "Biodiversity Conservation: The Key is Reducing Meat Consumption." Science of the Total Environment 536: 419–431.
- Maestas, Nicole, Kathleen J. Mullen, and Alexander Strand. 2013. "Does Disability Insurance Receipt Discourage Work? Using Examiner Assignment to Estimate Causal Effects of SSDI Receipt." American Economic Review 103 (5): 1797–1829.
- Maestre-Andrés, Sara, Stefan Drews, and Jeroen van den Bergh. 2019. "Perceived Fairness and Public Acceptability of Carbon Pricing: A Review of the Literature." *Climate Policy* 19 (9): 1186–1204.
- Markussen, Peter, and Gert Tinggaard Svendsen. 2005. "Industry Lobbying and the Political Economy of GHG Trade in the European Union." *Energy Policy* 33 (2): 245–255.
- Marquis, Christopher, and Andrew Park. 2014. "Inside the Buy-One Give-One Model." *Stanford Social Innovation Review* 12 (1).
- Mason, Winter, and Siddharth Suri. 2012. "Conducting Behavioral Research on Amazon's Mechanical Turk." *Behavior Research Methods* 44 (1): 1–23.
- Matthey, Astrid, and Tobias Regner. 2011. "Do I Really Want to Know? A Cognitive Dissonance-Based Explanation of Other-Regarding Behavior." *Games* 2 (1): 114–135.
- Meer, Jonathan. 2014. "Effects of the Price of Charitable Giving: Evidence from an Online Crowdfunding Platform." *Journal of Economic Behavior & Organization* 103: 113–124.
- Meier, Stephan. 2007. "Do Subsidies Increase Charitable Giving in the Long Run? Matching Donations in a Field Experiment." Journal of the European Economic Association 5 (6): 1203–1222.
- Mekonnen, Mesfin M., and Arjen Y. Hoekstra. 2012. "A Global Assessment of the Water Footprint of Farm Animal Products." *Ecosystems* 15 (3): 401–415.
- Meng, Kyle C., and Ashwin Rode. 2019. "The Social Cost of Lobbying over Climate Policy." *Nature Climate Change* 9 (6): 472–476.

- Meyer, Andrew. 2015. "Does Education Increase Pro-environmental Behavior? Evidence from Europe." *Ecological Economics* 116: 108–121.
- Milfont, Taciano L. 2009. "The Effects of Social Desirability on Self-Reported Environmental Attitudes and Ecological Behaviour." *Environmentalist* 29 (3): 263–269.
- Mishra, Arul, and Himanshu Mishra. 2011. "The Influence of Price Discount versus Bonus Pack on the Preference for Virtue and Vice Foods." *Journal of Marketing Research* 48 (1): 196–206.
- Momsen, Katharina, and Markus Ohndorf. 2020. "When Do People Exploit Moral Wiggle Room? An Experimental Analysis of Information Avoidance in a Market Setup." *Ecological Economics* 169: 106479.
- Newell, Richard G., and Juha Siikamäki. 2014. "Nudging Energy Efficiency Behavior: The Role of Information Labels." *Journal of the Association of Environmental and Resource Economists* 1 (4): 555–598.
- **OECD.** 2012. Lobbyists, Governments and Public Trust, Volume 2: Promoting Integrity through Self-Regulation. Paris: OECD Publishing.
- **Onwezen, Marleen C., and Cor N. van der Weele.** 2016. "When Indifference Is Ambivalence: Strategic Ignorance about Meat Consumption." *Food Quality and Preference* 52: 96–105.
- Paes, Daniel, Eduardo Arantes, and Javier Irizarry. 2017. "Immersive Environment for Improving the Understanding of Architectural 3D Models: Comparing User Spatial Perception between Immersive and Traditional Virtual Reality Systems." Automation in Construction 84: 292–303.
- Paolacci, Gabriele, Jesse Chandler, and Panagiotis G. Ipeirotis. 2010. "Running Experiments on Amazon Mechanical Turk." Judgment and Decision Making 5 (5): 411–419.
- Peck, Tabitha C., Sofia Seinfeld, Salvatore M. Aglioti, and Mel Slater. 2013. "Putting Yourself in the Skin of a Black Avatar Reduces Implicit Racial Bias." *Consciousness and Cognition* 22 (3): 779–787.
- **Poore, Joseph, and Thomas Nemecek.** 2018. "Reducing Food's Environmental Impacts through Producers and Consumers." *Science* 360 (6392): 987–992.
- Rabin, Matthew. 1995. "Moral Preferences, Moral Constraints, and Self-Serving Biases." Working Paper.
- Rand, David G. 2012. "The Promise of Mechanical Turk: How Online Labor Markets Can Help Theorists Run Behavioral Experiments." *Journal of Theoretical Biology* 299: 172–179.
- Reiley, David, and Anya Samek. 2019. "Round Giving: A Field Experiment on Suggested Donation Amounts in Public-Television Fundraising." *Economic Inquiry* 57 (2): 876–889.

- Ribar, David C., and Mark O. Wilhelm. 2002. "Altruistic and Joy-of-Giving Motivations in Charitable Behavior." Journal of Political Economy 110 (2): 425– 457.
- Ross, Joel, Lilly Irani, M. Six Silberman, Andrew Zaldivar, and Bill Tomlinson., "Who Are the Crowdworkers? Shifting Demographics in Mechanical Turk." In CHI '10 Extended Abstracts on Human Factors in Computing Systems (CHI EA '10): 2863–2872. New York: Association for Computing Machinery.
- Rubin, Donald B. 1974. "Estimating Causal Effects of Treatments in Randomized and Nonrandomized Studies." *Journal of Educational Psychology* 66 (5): 688–701.
- Schultz, P. Wesley, Jessica M. Nolan, Robert B. Cialdini, Noah J. Goldstein, and Vladas Griskevicius. 2007. "The Constructive, Destructive, and Reconstructive Power of Social Norms." *Psychological Science* 18 (5): 429–434.
- Seinfeld, Sofia, J. Arroyo-Palacios, G. Iruretagoyena, R. Hortensius, L. E. Zapata, D. Borland, B. de Gelder, M. Slater, and M. V. Sanchez-Vives. 2018. "Offenders Become the Victim in Virtual Reality: Impact of Changing Perspective in Domestic Violence." Scientific Reports 8 (1): 1–11.
- Serra-Garcia, Marta, and Nora Szech. 2020. "The (In)Elasticity of Moral Ignorance." CESifo Working Paper No. 7555.
- Sinha, Indrajit, and Michael F. Smith. 2000. "Consumers' Perceptions of Promotional Framing of Price." *Psychology & Marketing* 17 (3): 257–275.
- Slater, Mel, and Maria V. Sanchez-Vives. 2016. "Enhancing Our Lives with Immersive Virtual Reality." Frontiers in Robotics and AI 3: 74.
- Smith, Alexander. 2013. "Estimating the Causal Effect of Beliefs on Contributions in Repeated Public Good Games." *Experimental Economics* 16 (3): 414–425.
- Stefanski, Leonard A., and Dennis D. Boos. 2002. "The Calculus of Mestimation." *The American Statistician* 56 (1): 29–38.
- **Tonsor, Glynn T., and Nicole J. Olynk.** 2011. "Impacts of Animal Well-Being and Welfare Media on Meat Demand." *Journal of Agricultural Economics* 62 (1): 59–72.
- **Vesterlund, Lise.** 2016. "Using Experimental Methods to Understand Why and How We Give to Charity." In *The Handbook of Experimental Economics*, edited by Kagel, John H., and Alvin E. Roth, Volume 2: 91–152. Princeton: Princeton University Press.
- Videras, Julio, Ann L. Owen, Emily Conover, and Stephen Wu. 2012. "The Influence of Social Relationships on Pro-environment Behaviors." *Journal of Environmental Economics and Management* 63 (1): 35–50.

- Warr, Peter G. 1983. "The Private Provision of a Public Good Is Independent of the Distribution of Income." *Economics Letters* 13 (2-3): 207–211.
- WeAreDynamo. 2014. "Guidelines for Academic Requesters." Version 1.1.
- Yokessa, Maïmouna, and Stephan Marette. 2019. "A Review of Eco-Labels and Their Economic Impact." International Review of Environmental and Resource Economics 13 (1–2): 119–163.
- Zimmermann, Florian. 2020. "The Dynamics of Motivated Beliefs." American Economic Review 110 (2): 337–361.
- **Zizzo, Daniel John.** 2010. "Experimenter Demand Effects in Economic Experiments." *Experimental Economics* 13 (1): 75–98.

Eidesstattliche Erklärung

Hiermit erkläre ich, dass ich die vorliegende Dissertation selbstständig angefertigt und die benutzten Hilfsmittel vollständig und deutlich angegeben habe.

 $\mathrm{Mannheim},\,28.04.2021$

Raphael Epperson

Curriculum Vitae

Ph.D. Studies in Economics, University of Mannheim
M.Sc. in Economics, University of Mannheim
Research Assistant at ZEW – Leibniz Centre for European Economic Research, Mannheim
B.Sc. in Economics, University of Mannheim
Visiting Student, National University of Singapore